

Orchestrating Network Behavior for Innovation

Busquets, Javier

Document Version Final published version

Publication date: 2010

License Unspecified

Citation for published version (APA): Busquets, J. (2010). Orchestrating Network Behavior for Innovation. Copenhagen Business School [Phd]. PhD series No. 26.2010

Link to publication in CBS Research Portal

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy If you believe that this document breaches copyright please contact us (research.lib@cbs.dk) providing details, and we will remove access to the work immediately and investigate your claim.

Download date: 04. Jul. 2025









COPENHAGEN BUSINESS SCHOOL HANDELSHØJSKOLEN SOLBJERG PLADS 3 DK-2000 FREDERIKSBERG DANMARK

www.cbs.dk

Orchestrating Network Behavior for Innovation



for Innovation

Javier Busquets

PhD Series 26.2010

LIMAC PhD School Programme in Informatics

ISSN 0906-6934 ISBN 87-593-8439-8



Orchestrating Network Behavior

PhD Series 26.2010

Orchestrating Network Behavior for Innovation

Javier Busquets Orchestrating Network Behavior for Innovation

1st edition 2010 PhD Series 26.2010

© The Author

ISBN: 978-87-593-8439-8 ISSN: 0906-6934

LIMAC PhD School is a cross disciplinary PhD School connected to research communities within the areas of Languages, Law, Informatics, Operations Management, Accounting, Communication and Cultural Studies.

All rights reserved.

No parts of this book may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording, or by any information storage or retrieval system, without permission in writing from the publisher.

LIMAC PhD School; Programme in Informatics Copenhagen Business School Department of Informatics

ORCHESTRATING NETWORK BEHAVIOR FOR INNOVATION (Thesis Dissertation)

Javier Busquets Director: Mogens Kuehn Pedersen

[1-June-2010]

Acknowledgments

I would like to thank my dissertation director, Professor Mogens K. Pedersen. His wise support and guidance helped me to stay true to myself by learning to choose a path that has culminated in this doctoral dissertation.

I also want to thank Jordi Riera, Vice-Rector of Ramon Llull University, for convincing me to start this doctoral journey. Above all, I am honoured to have been able to call him a friend for more than 30 years. I would also like to extend my sincerest thanks to José María Añón and Joan Ramón Mallart for their friendship, and for helping me to open new doors when this research project was more a dream than a reality. My thanks also go to Multiasistencia and IBM Spain because this project would have been impossible without their support and patience.

I am grateful to have had the opportunity to be Department Chair of the Department of Management of Information Systems (MIS) at ESADE. I would like to thank the whole team who have been an inspiration as I wrote and assembled this dissertation between 2003 and 2008. I also want to thank ESADE's management and Dean's office for their support throughout this project: Carlos Losada, Xavier Mendoza, Alfons Sauquet and Josep Bisbe.

Above all, I would like to thank my family for once again demonstrating their love and near limitless patience as I worked on this project. To them, I dedicate this dissertation with great affection.

Content

Chapter	1: Introduction and Overview	9
Chapter	2: Networks of Innovation, a Literature Review	19
	2.1 Introduction	19
	2.2 Main Definitions	19
	2.2.1 Innovation	19
	2.2.2 Networks	20
	2.2.2.1 Creativity	22
	2.2.2.2 Autonomy	22
	2.2.2.3 Motives.	23
	2.2.3 Power	24
	2.3 Network Structures	25
	2.3.1 Fauilibrium	
	2 4 - A Classification of Networks of Innovation	26
	2 4 1 - The Primordial Network	29
	2.4.2 - The Networks of Discrete Innovations	30
	2.4.2. The Transactional Network	31
	2.4.4 The Stratogic Alliance	JI 21
	2.4.5. The Differentiated Network	
	2.4.3 The Differentiated Network	
	2.5 Olchestiation and Powel	
	2.5.1 Focal Firms and Centrality and Roles	
	2.5.2 Roles of the Focal Firm	
	2.5.3 Control over resources	
	2.5.3.1 Relationship dynamics	
	2.5.3.2 Trust	40
	2.5.3.3 Digital Platforms	42
	2.5.4 Orchestration and Dynamic Capacities	44
	2.5.4.1 "Asset Orchestration"	44
	2.5.4.2 Value generation and appropriation	45
	2.5.4.3 Boundaries and "Complementors"	45
	2.6 Conclusion: Combining Power and Orchestration	46
Chapter	3: Orchestrating Network Behaviour for Innovation	48
•	3.1 Introduction	48
	3.2 Orchestrating Network Behaviour for Innovation	
	3.3 Networks as Optimum configurations for innovation	52
	3.4 Strategic Management and Power through Commitment	
	3.4.1 Autonomy and Creativity	
	3.4.2 Path dependencies and Deep Structures	
	3.5 - Network Structural Change	56
	3 5 1 - Dynamic Ontimums	56
	3.5.2. Centrinetal and Centrifugal forces	58
	3.6 - Digital Platforms as Innovation Infrastructures	60
	37 - The Network Leader	
	3.7.1 - Organisational Canacity	ייייייייייייי בא
	3.7.1. ² Organisational Capacity	۲0
	2.7.2 Economic conscient	04 בס
	2.9 Conclusion	/۵
		ðð

Chapter 4: Research Perspective and Design	70
4.1 Introduction	70
4.2 Paradigmatic Approach: Critical Rationality and Positivism	70
4.3 Methodology	72
4.3.1 Abduction	73
4.3.2 Method	74
4.3.3 Units and Levels of Analysis	75
4.3.4 Paths and Narratives	77
4.4 Case Selection and Presentation	77
4.4.1. Multiasistencia	78
4.4.2. IBM Barcelona Lab	79
4.5 Data Collection	80
4.5.1 Interviews	80
4.5.2 Other Sources	83
4.6 Conclusion	85
Chapter 5: Orchestrating and Network Structural Change	86
5.1. Introduction	86
5.2. Case Presentation	86
5.3. Multiasistencia	87
5.3.1 Network Creation (1984-1995)	88
5.3.2 The Network Boom (1995-2000)	
5.3.3 Network Change (2000-2004)	91
5.3.4 Network Behaviour for Innovation (2004-2008)	95
5.3.5 2008 to Present: Towards a Dominated Network	
5.4 Case Study: IBM Barcelona Lab.	
5.4.1 Network Initial Conditions: The PC in the Bank Branch.	
5 4 2 - Network Change (1994-1996)	100
5.4.3 - Network Behaviour for Innovation (1996-2004)	102
5 4 4 - Becoming a Competence Centre	108
5 5 - Propositions	100
5.5.1 - Orchestrating Networks: The Dual Nature of Power	109
5.5.1. Orenestrating retworks. The Data Nature of Fower	112
5.5.2. The Dynamics of Co Creation Strengtherence	116
5.6 - Conclusions	117
J.0. Conclusions	11/
Chapter 6: Orchestrating Generative Networks of Innovation	110
6.1 - Introduction	110
6.2 - Congrative Networks: A Framework of Commitment	110
6.2 - Dominant Designs and Business Models	172
6.4. Co croation	123
6 E Digitally enabled commitments	125
6.5. A dynamic view of commitment	120
6.0 A dynamic view of communent	124
6.7.1 Concernations Material Material	134
6.7.2. Disital Distance and Constraints Naturality	135
6.7.2 Digital Platforms and Generative Networks	OCT
6.7.3 The Role of Orchestrating in Factor / Market creation	13/
6.8 Conclusions	138
Chapter /: Network Leadership	140

7.1Ir	troduction	140
7.2 N	letwork Leadership	141
7.	2.1 Innovation and Isolation	142
7.	2.2 Empathy	144
7.	2.3 Exerting Autonomy	145
7.	2.4 Resilience	145
7.	2.5 Authenticity	146
7.3 0	onflict Management	146
7.	3.1 Role and Relational Conflicts: Re-thinking Centrality	147
7.	3.2 Political Mastery	152
7.	3.3 Managing Transitions: linking the old and the new	154
7.4 0	Prganisational Capability for Co-Creation	158
7.	4.1. Communication	158
7.	4.2. Managing Strategic Relationships	160
7.	4.3. Mutual Adaptation for Co-creation	162
7.	4.4 Multi-disciplinary Teams	165
7.	4.5 Recognition	166
7.5 E	conomic Capacity	168
7.6 P	roposition: Orchestration as Leading Leaders	169
Chapter 8: Con	clusions: Perspectives on Innovation and Leadership	171
9 References		192

Chapter 1: Introduction and Overview

This thesis is about innovation and power. Human nature has always been expressed by our capacity to innovate and adapt to almost any environment (Bowlby, 1962; Giddens, 1991). In the 20th century, the primary function of business organisations was to invent, produce and commercialise their products and services in different markets. As a matter of fact, business organisations in the last century proved to be the best way of disseminating innovation (Schön, 1971). Currently in the 21st century, there is a call to better understand how new ideas, technology and sources of knowledge are managed, based on the premise that novelty can unfold anywhere and that innovation cannot be considered a linear process consisting of a chain of activities.

Innovation is an abstraction embracing many different activities to ensure the long-term survival of firms: from design processes for new products or services to solving problems, increasing efficiency, reducing waste, developing the market and creating new business models (Day & Schoemaker, 2000; Van de Ven et al., 2008; Chesbrough, 2006;2003). On the other hand, innovation is paradoxical in nature in that it can conflict with the existing status quo, that is, incumbent structures which are referred to as the "rules of the game." This term stresses their political nature which, in business, essentially represents the rules and norms of markets and business organizations (or hierarchies) as well as their interaction.

Early innovation studies stressed that novelty (new entrants, new laws, new technologies, and other "invaders") can "annoy" markets by introducing differences that eventually change the "rules of the game" and alter the existing equilibrium. This process can create new conditions which are "destructive" for incumbent firms and existing offerings (Schumpeter, 1942). In contrast, and changing the perspective to the firm level, Alfred D. Chandler's (1962) classic text, *Strategy and Structure*, hones in on innovation to discuss how enterprises develop new strategies to cope with new demands and how, as a consequence, they design appropriate structures to control and communicate activities through the different units that make up the business firm.

Innovation at the firm level, in contrast to the processes of organic life and death, emphasizes managerial action to drive changes to fill the gap between strategies and the needed structures. It can be assumed, then, that innovation is a purposive managerial action requiring choices, investments and the mobilization of resources to avoid inefficiencies along the way. Innovation thus drives a complex organisational metamorphosis to unfold novelty, which, in turn, requires time to provide results and is expressed through a process. As such, innovation can be seen, metaphorically, as a path or a journey (Van de Ven et al., 2008), underscoring the inherent activity of "organising" as the creation of paths towards efficiency which are not pre-defined but dependent on managerial decision-making and action (Van de Ven, 1992).

This path or journey consists of steps into the unknown, managing important risks that may vary along the way. The question is how to set the path. Options include markets and hierarchies, and their interaction. Today's business context reveals increasing discontent with inefficient market fluctuations and cycles and, at the other extreme, mechanistic hierarchies. This researcher stresses here that innovation is dependent on having the appropriate structures, and that these structures, at the same time, can be seen as the result of innovation processes since novelties lead to structural change. Even though there are many *business configurations* (Mintzberg, 2002), organisations have shown a strong tendency to conform to "the machine" paradigm due to their need to develop scale and production efficiency. These kinds of structures show a strong propensity to engage in repetitive behaviour, being prone to highly stable organisational processes which can generate deterministic paths due to their dependency on the inheritance of past efficiencies.

In contrast, Networks lie between markets and hierarchies, and are becoming a preferred context for a strategy based on innovation (Penrose, 2008; Powell et al., 2005). What distinguishes a network from other organisational structures is that it embraces the autonomy to choose to establish relationships driven by creative and differentiated actors. Networks reveal relationships and dependencies among actors (Pfeffer & Nowak, 1976), stressing that innovation

does not unfold in isolation, along with the need for governance and coordination. Here we can use the term 'orchestration' as managerial action to define how a new path for innovation is created and how this path crystallizes into viable economic outcomes. Understanding this process is the main motivation behind this research.

Orchestrating Network Behaviour for Innovation

The main question in this thesis is: "How does orchestration steer network behaviour for innovation?" To respond to this query, this phenomenon is presented as a managerial function to create new structures. Although this discourse is nascent, it has already received some attention in academic literature, combined with the strategic importance of networks, system integration and digital platforms (Chesbrough, 2008; Adner, 2006), the latter emphasising the role of Information and Communication Technologies (ICTs).

Orchestrating can thus be viewed as a managerial function that supersedes how economic factors can be managed to produce efficiently in mass markets. In other words, *Orchestrating Innovation* can be seen as an activity intended to manage investments in order to build paths of efficiency to better compete in existing markets but also as an activity for factor or market creation (Ghemawat, 1997; Miller & Olleros, 2007). Paths are no more than expressions of dynamic and structural change, and networks have been used to examine these phenomena (Powell et al., 2005). These activities also require a better understanding of the duality of dynamic capacities to adapt and transform the environment (Teece, 2009; Augier & Teece, 2008; Pedersen, 1996). This is why I ground this function on a critical review of the existing literature, stressing the longitudinal empirical work carried out from 2004 to 2008 on two different cases of orchestration, spanning a period of approximately 25 years each. The aim of this thesis, then, is to contribute to existing theory and to managerial reflection, as well as to propose ideas for further research.

Innovation, Networks and Power

With this function, the three main subthemes of this work are synthesised as follows: (1) innovation as a source of future value; (2) networks as different from organisations; and (3) power as management's main activity and a means to understanding organisational structures. The intention is to make this research operational by focusing on the dual nature of innovation, that is, to create new factors and markets, and by examining the role of digital technologies in this process. Networks are indeed a good context to examine both the sources of innovation and how users learn to cooperate and use innovations.

We have alluded to innovation as a journey. This notion has an antecedent in the metaphor of Ulysses and his journey from Ithaca to Troy and back. Ithaca here is a metaphor for Ideas, while Troy represents the world (In Latin, Terra). One case for the completion of the myth of Ulysses is the invention of the concept of Science: Ulysses returns to Ithaca from the contingent and temporal world to the eternal place where ideas reside. In this metaphor we can see the need for efficiency, on the one hand, because the time for the journey is limited just as human and organisational life is. On the other hand, there is a need to manage innovation, since realistic commitments are required under different conditions and sets of constraints, given that the journey is long and sea and metrological conditions changeable. Perhaps the most celebrated episode of the Odyssey is when Ulysses binds himself to the mast to protect himself from the sirens' songs attempting to lure him off-course (see Elster, 2000). Here, Ulysses reveals his commitment to himself, to his men and the ship. Elster grounds his theory of binding rationality on this episode. By binding, we refer to the definition of an objective and remaining true to this goal throughout the journey, until the end a fact which gives the trip meaning and represents effective actions, as occurs with a path towards innovation.

The second aspect examined in this thesis is networks. The intellectual appeal of Innovation Networks resides in the variety of issues raised at the organisational level. Furthermore, these networks call upon individual attributes such as creativity, knowledge and talent which contribute new combinations of existing resources (Pedersen & Larsen, 2006) with an intensive use of information and communication technologies (ICTs). Indeed, innovation is increasingly dependent on digital platforms (Gawer & Cusumano, 2008). Moreover, in the current digital era, it is somewhat difficult to define formal structures or ascertain where boundaries lie. Creativity and talent can be found beyond the walls of incumbent organisations which, in turn, are evolving and becoming more porous. In addition, the establishment of relationships with actors and partners from other organisations leads to the emergence of business networks.

What distinguishes a network from other organisational structures, however, is that the former embraces creative and autonomous workers as part of its very foundation. This is not merely aesthetic intent since it has evident economic benefits: in knowledge-intensive environments, the average EBITDA¹ per employee is 6 times greater than in productive industrial environments (see Beardsley et al., 2006). Indeed, networks are increasingly seen as emergent, alternative ways to organise, based on autonomy and voluntary relationships.

Autonomy deals with empowerment and allows the actors to do more, conceiving networks as relationships among active members in contrast to hierarchical organisations where agents receive orders. Autonomy is about freedom. It highlights that group collaboration does not occur by accident or through coercion but by choice. There may be forces other than autonomy that link people together to co-create and co-produce. In this thesis we shall explore new phenomena that represent emergent organisational patterns which conflict with the traditional idea of hierarchical organisations and "central planners" (Hayek, 1949) as well as with "chaotic markets", which are also subject to non efficient cycles (see Penrose, 2008).

For example, the Linux and Open Source movements encompass more than 200,000 programmers outside formal organisations. If then, we accept that group collaboration is achieved based on human relations outside the "scientific",

¹ Earnings Before Interest, Taxes, Depreciation and Amortization

Taylor-based work plant model, what cohesive force is behind this organisational phenomenon? The concept of knowledge workers emphasises the idea of enhancing effectiveness through talent, interactions, diversity and heterogeneity (Powell et al, 2005). In a network context, creativity and inspiration do matter, but so do perceptiveness, communication and trust. Innovation occurs at the boundaries between mindsets, between subjectivity and objectivity, exploring the provisional territories of the unknown through communication and interactions (see Leonard-Barton, 1995) to create novelty.

Finally, this is why power is the main theme of this thesis. It is worth noting that innovation, as a systematic activity, needs to understand power rules both at the structural level and at the level of managerial action in order to cope with its paradoxical nature, i.e., being critical to ensure long-term survival but, at the same time, serving as a source of conflict and tension in both markets and organisations by challenging the forces of equilibrium (Gersik, 1991). Innovation is not necessarily "good" or beneficial to all. Existing organisations and structures may not accept changes which are dependent on deviations from a previous path. Power is a resource to cope with resistance and with complexity because innovation is not an activity that unfolds in isolation. In a general sense, power is expressed by exerting ones influence to set boundaries or limits of action, reducing uncertainty and holding potential conflict at bay.

Power and innovation, though, have compelling antecedents in both markets and hierarchies. Although hierarchies seemingly have a problem with the markets (and indeed with innovation); due to markets being unpredictable and hierarchies being bent on repetitive behaviour, leading to the production of outdated products and services for non-existing markets, jeopardising the firm's performance and its future existence. On the individual level, there are classic and compelling antecedents which argue that both no control, or too much control may create spaces in which people cannot perform (see Freud, 1921/2001; Morgan, 1997). In addition, there are also "anomic contexts," to use Durkheim's (1937; 1933) and Parson's (1937) terminology, which can repress and eventually destroy the individual's capacity for creativity and learning.

Network Leaders

In this thesis there is an explicit interest in understanding Orchestration as a network capacity, but also the way in which individual leaders are related to this concept in terms of how they exert power and organise for innovation. Thus, this thesis not only examines innovation but also "the innovators" and "the leaders of innovation," since this activity is dependent on strong psychological commitments. Managing innovation is a critical activity (Burns & Stalker, 1961); it is not easy but, rather, decidedly risky and subject to uncertainty. In fact, in contrast with mainstream myths which see the manager of innovation as "the heroic entrepreneur," evidence shows that few managers survive the process of steering tottering organisations down new paths in their quest for structural change (see Kets de Vries, 2009).

As argued, an increasing number of voices are questioning not only existing power structures in organisations and managerial practices (Benkler, 2006) but also the role of management (Hamel, 2007), thus superseding the administrative canon in which human beings are conceived as agents who are passive, reactive, opportunistic and "bound" in terms of rationality. So if group collaboration is achieved based on human relations outside the "work plant" model, what cohesive force is behind this organisational phenomenon? What source of power is behind it? Similarly, power and organisational structures can shape contributions from new voices, along with the contribution of novel ideas and interactions from this new type of worker.

With respect to the question posed at the beginning of this introduction: it is quite obvious that managing innovation is difficult and risky and thus implies facing important challenges. This thesis also examines individual capacities in a more normative way, by considering situations in which the organisations studied face the risk or the reality of discontinuing some business units or production lines. It is precisely in this context that Orchestration has been observed. The question here is, returning to the myth of Ulysses, how managers are able to make realistic commitments under different conditions or sets of constraints, and how this affects innovation, since the journey is long, and the sea and metrological conditions, may change.

Organisation of this Thesis

This thesis presents Orchestrating Business Networks for Innovation as a managerial function to create and capture the value of innovation. It is only one of many ways to innovate. The core of this function is to intermediate between the past, the present and the future, mobilising relationships and networks to unfold novel results and create a path for efficiency whose objective is to set a path for innovation. This function presupposes that networks are a natural context for creativity and autonomy, opening up new possibilities, because managing network behaviour can change the rules of the game. Managerial capacities focus on how to dynamically manage the equilibrium and boundaries of action. How is the function of Orchestrating Network Behaviour for Innovation better able to adapt as a process-like approach?

This thesis is grounded on the strategic management and dynamic capacities perspective, that is, based on understanding how potentialities unfold towards future capacities (Teece, 2007; Augier & Teece, 2008) and underscoring the notions of co-creation and collaborative organisation (Eisenhardt & Schoonhoven, 1996). Relationships are thus considered to be the principal resource of any network, and, by engaging in relationships, we must understand that they are investments. Since innovation is a journey, the different factors, dependencies and context will change along the way, as will the managerial actions affecting creativity. It also implies openness, new voices, perspectives and the ability to develop learning capacities within the network. By the same token, most products require services from several resources, and firms may need several products and services as well as a combination of knowledge to develop a new resource.

Based on the above arguments, this thesis consists of eight chapters. Following this introduction, Chapter 2 reviews the existing literature on networks of

innovation, providing definitions for the terms *innovation, networks* and *power*, and reviewing the literature according to three power dimensions: centrality, control over resources and dynamic capacities. A classification of Networks of Innovation based on these dimensions is also given. Finally, the existing gap that has been pointed out in recent literature between power and orchestration is highlighted.

Chapter 3 presents Orchestrating Network Behaviour for Innovation as a managerial function. This role is presented as a path that starts from the current situation and drives network behaviour to a possible future, seeing innovation as a process or a journey affecting: (1) network dynamics and digital platform management and (2) individual management regarding organisational, economic and conflict management capacities. Networks evolve over time; they are not static structures. A process-level approach and the managerial function in innovation are highlighted. How do processes cope better with the need to innovate in networks? What does network memory mean? How does it affect current and future actions in helping to redefine behaviour and developing guidelines for new actions?

Chapter 4 describes my paradigmatic position and research design, explaining the theoretical foundations for the methodological approach and the exploratory method used. The point of departure for this work in terms of this researcher's scientific position is explained, and the set of analytical techniques and perspectives which serve as the basis to perform the research presented in this thesis are elaborated upon.

Both Chapters 5 and 6 contain an analysis and description of the cases on which the empirical part of Orchestrating Network Behaviour for Innovation is grounded. Chapter 5 presents orchestration's capacity to dynamically manage network boundaries and explores the concepts of roles, relationships, autonomy and robustness; it also proposes a way in which to analyse the cases presented at different stages of network evolution: their formation, growth, change and stabilisation to dynamically co-create digital platforms. Chapter 6 explores the role of digital platforms in Generative Networks and their relationship with the creation of new markets and dominant designs, co-creative activities and programmed changes based on digital platforms.

Chapter 7 specifically analyses the role of the Network Leader, focusing more on a normative view and exploring subjective perspectives regarding managerial practice according to my observations and reflections in this research. Here, the capacity to manage conflict, organisational capacity and performance is examined. Moreover, the role of leadership and its implications for Orchestrating Networks are analysed.

Chapter 8 concludes with the theoretical contributions of the function presented in this thesis and poses questions and challenges for further research.

Chapter 2: Networks of Innovation, a Literature Review

2.1.- Introduction

This chapter presents the literature review of this thesis: Networks of Innovation and defines *Innovation, Networks* and *Power,* organised as the three main themes of this thesis. Next, the main themes that relate networks and innovation are given, as well as the concept of Network Structure and Equilibrium that prepares for the next chapter where the main notion of this thesis, Orchestrating Network Behaviour for Innovation, grounded in structural change and dynamics, is presented.

Finally, the literature of Networks of Innovation is reviewed around the notion of focal firms and orchestrators to describe three power dimensions: centrality, control over resources and dynamic capabilities presenting different structures and configurations. This has two intentions: first, a classification of Networks of Innovation is presented with emphasis on these power dimensions, and second, the gaps between Orchestration and Power are presented and explored.

2.2. – Main Definitions

2.2.1. – Innovation

This thesis defines *Innovation* as the relationship between current business affairs and possible future ones. To that end, it is recognised that innovation is a process that can be represented metaphorically as a journey or a path (Van de Ven et al., 2008). As a process, innovation is required to guarantee long-term organisational survival and to overcome sub-optimal regimes (Elster, 2000:1). However, it is also a source of conflict and tension, since any innovation introduces a difference that can upset the forces of equilibrium in organisations, industry structures and markets (Gresov & Drazin, 1997).

Innovation is linked to *revolutionary behaviour*, according to Elster (1989; 2000), and starts by holding some resources for short-term consumption, and investing them to build and co-create *something new*. This recognises that the potential results and benefits of these shared investments require time. As such, a process view is needed to examine and understand the set of activities involved.

Indeed, innovation dynamics may be synthesised into two major processes: (1) product/market competition, and (2) factor/market creation (Miller & Olleros, 2007; Ghemawat, 1997). The first process linking innovation with business activities is to find or create superior *positions* in good industry structures (Porter, 1980; 1996) and/or to identify, select and deploy *resources* to generate superior performance and competitive advantage (Barney, 1994; Wernerfelt, 1984; see Christensen, 1997; Christensen & Overdorf, 2000).

The second major process, factor/market creation strategies, considers innovation related to changes in structures, behaviour and styles, requiring the definition of a path to manage transitions. This approach is normally analyzed from the dynamic capacities school (Teece et al., 1997; Peteraf et al., 2008), which essentially focuses on organizational change by using capacities and the re-combination of existing resources (Augier & Teece, 2008). Processes of change are normally associated with networks, systemic innovations and digital platforms for extending the resource base at the structural level (Miller & Olleros, 2007; Dyer & Kale, 2007). The strategy to engage in network exchanges is coherent with the aim of sharing the costs and risks associated with "leap frogs" or radical innovations (Penrose, 2008).

2.2.2.- Networks

Network is defined as a set of actors who are interconnected by a series of relationships (Emerson, 1962). Among the many types and existing definitions of Networks and Business Networks (see Alstyne, 1997), *Networks of Innovation* highlight the fact that firms cannot innovate alone, requiring searching for and engaging in productive relationships (Penrose, 2008; Gnyawali & Madhavan, 2001; Jarrillo, 1993).

According to the literature reviewed, studies on Networks of Innovation focus on one of several areas: (1) the types of innovation involved, (2) network functions, (3) clusters, and (4) structural attributes such as "innovativeness" and structural change.

First, Networks of Innovation allow scholars to examine the type of innovation that takes place, that is, whether these networks give rise to: (1) radical or continuous innovations (Christiansen, 1997) or (2) unique or systemic changes (Teece, 1987). Here, innovation studies emphasize the level of connectivity between complementary components (Chesbrough, 2008; Teece, 1989). Second, studies on network functions encompass: (1) diffusion (Rogers, 1995) or – in very few analyses - on innovation generation (Van de Ven et al., 2008; Van de Ven & Rogers, 1988), (2) temporary entrepreneurial structures (Elfring & Hulsink, 2007), or (3) even *transitional organisations* placed at the meso-level of analysis, between industry structures, markets and business organisations (Miller & Olleros, 2007). Third, other studies focus on networks as "innovation clusters" or "hot spots", revealing the dependencies of contemporary firmsupon their network relationships. Clusters enable and support business by raising productivity, the speed of innovation, and the faster formation of new companies (Porter, 1998; Cowan, 2007).

Finally, structural studies examine focal firms, relational dynamics, network density, positions and the network capacity of "innovativeness" (Powell et al., 2005; Powell, 1990; Damanpour, 1991). They focus on processes of innovation and dynamics (Murray, 2002; Brown & Eisenhardt, 1996) that are normally expressed as the generation of paths or trajectories (Pavitt & Steinmuller, 2002; Pavitt, 1984).

In all cases, the emphasis of networks is on actors' autonomy, creativity and motives for engaging in exchanges and communication, aspects that are briefly reviewed in the next sections.

2.2.2.1.- Creativity

Innovation, as the propensity for problem solving and invention, is a natural (innate) human capacity (Huizinga, 1927/2002; Winnicott, 1971). Current notions of innovation embrace sophisticated processes and the management of many sources, making it an organizational activity, that although grounded in natural (innate) capacities, requires learning processes as well. One can define creativity as a function, an *autonomous activity* that gives us "the capacity to create the world" as a result of the experience of being alive and connecting what was unconnected (Kets de Vries, 1980; 2009). It is a part of instinctual human capacities and not necessarily a response or reaction to any external stimuli (Winnicott, 1971; 1964). Above all, however, it is related to the need to have fun and feel pleasure (Freud, 1966). Creativity is linked to fun, since it mixes reality and imagination, implying "as if" situations for problem-solving, theories, symbols, projects, riddles and playing (Huizinga, 1927/2002; Winnicott, 1988). This natural capacity allows for the exploration of boundaries between subjectivity and objectivity to recreate the world around us, to understand it better and to engage in productive social interactions.

2.2.2.2.- Autonomy

The key notion here is autonomy, the capacity to choose and engage in voluntary associations (Hayek, 1949/2009; Rand, 1967; Child, 1972). The notion of autonomy is a relative concept since it implies degrees. Further, in order to make sense of the concept, it needs to be differentiated within the framework of the different contexts in which we carry out our activities. Autonomy in terms of citizenship implies freedom of choice with a certain set of constraints normally based on constitutions, laws, as well as on social norms. In business activities, the degree of choice may refer to "job description" or "accountability".

Networks of innovation refer to the degree in which one can establish relationships with other firms beyond the existing legal and cultural organizational norms. Indeed, it is supposed that autonomy allows the actors *to do more*, conceiving networks as relationships among active members, with an emphasis on the links that are set beyond formalized relationships, thus emphasizing commitments and trust dynamics (Thorelli, 1986; Ring & Van de

Ven, 1994; Barney, 2007). Yet, the autonomy is not absolute but rather in degrees and is dependent on the autonomy of others, as it takes two to collaborate. Therefore, by stressing autonomy, the need for constraints is emphasized, since it is not the case that any choice goes.

This contrasts with the passive recipients of orders given by agents in bureaucracies, and also highlights the fact that teams or "group collaboration" are dependent on willingness to collaborate, individual capacities (see Bowlby, 1985), which constitute constraints that make this cooperation possible, emphasising that relationships among autonomous actors do not occur by accident or through coercion. There may be some other forces that link people together to co-create and co-produce. Actors that forge a Network for Innovation tend to create enduring social relationships based on rational choice and the calculation of future benefits and costs (Cook & Emerson, 1984; Child, 1972).

2.2.2.3.- Motives

According to the literature reviewed, actors join networks in order to: (1) share the costs and risks of innovation (Penrose, 2008); (2) take advantage of the superiority of networks in terms of transaction costs (Kogut, 1988; Coase, 1988; Williamson, 1991; 1975), thus reducing risks of opportunism and economic friction between partners along with misunderstandings and conflicts that may lead to delays, breakdowns, and other malfunctions (Williamson, 1994; Eisenhardt, 1989:2); (3) take advantage of strategic and resource-based fit (Kogut, 2000); (4) benefit from social and cultural fit among actors (Eisenhardt & Shoonhoven, 1996); and (5) benefit from trust in network dynamics (Ring & Van de Ven, 1994).

On the other hand, Sociology is based on exchange theories (Emerson, 1962; 1987; see also Molm, 2003) in which the main assumptions are that network formation is the result of: (1) an actor's behaviour motivated by the desire to increase gain and avoid loss; (2) exchange relations developing into structures of mutual dependency and power dynamics; (3) actors engaging in recurrent, mutually contingent exchanges with specific resources and partners over time; or

(4) valued outcomes obeying the economic law of diminishing marginal utility (Molm et al., 2000; Molm, 2003) which leads to the principle of equilibrium that is presented later.

On this basis, Exchange theories predict that the behaviour of individual and autonomous actors depends on exchange dynamics and its effects as outcomes in the structure itself (Emerson, 1962; 1987; see also Molm, 2003). This requires a closer look at structural dynamics and the notion of power and governance.

2.2.3.- Power

Power is the capacity to generate dependencies in others in order to steer their behaviour. It is considered to be a *resource* in organisational literature. In a general sense, it is expressed through the causation of the following effects: (1) the influence over others' behaviour (Scott, 2001); (2) the ability to determine boundaries or limits of action (Bettelheim, 1988; Bowlby, 1962; Emerson, 1987; Mintzberg, 1973); (3) reduced uncertainty (Thompson, 1967/2003; Child, 1972) and (4) holding conflict (Pfeffer, 1994; Milgram, 2005).

As such, power can be seen as linked to strategy and the contingent nature of human action (Clegg, 1989), which is subject to constraints (Barnard, 1938/1968), both in economic and social life (Elster, 1989). Thus, power is connected to strategy, since it is about making a choice to set chosen constraints on organisational behaviour, that is, about shaping a style. Second, power relates directly to structures (Giddens, 1987). Structures are defined by their boundaries (limits of behaviour), roles and relationships, variables that, on the one hand, are interrelated, and do not vary independently, and, on the other, which shape and are shaped by behaviour (Schön, 1971; Giddens, 1987).

Power is related to conflict theories (Gersick, 1991; Pfeffer & Salancik, 1978). This is in contrast to "politics" or the resistance to managerial action (Pfeffer, 1994; Pfeffer & Salancik, 1978). Indeed, networks are political organizations, the arena of tensions and conflict, that Thorelli (1986) metaphorically calls "network entropy." This consists of: (1) conflicting relationships in order to control resources (Gresov & Drazin, 1997); (2) relational tensions between autonomy

and cohesion (Nhoria & Ghoshal, 1997); (3) loose goal-setting and direction; and/or (4) other variables that may conflict with managerial purposive action (Pfeffer & Salancik, 1978; Benkler, 2006). The control of entropy is thus, in a continuous flux of balance with mutual adaptation between actors and the governance and coordination mechanisms (Thorelli, 1986).

Power studies generally start at the relational level (Clegg, 1989; Emerson, 1962) and expand into the aggregated level, addressing the critical notion of "coordination" (Galbraith, 1977; 1973) and its influence on relational dynamics, roles and network boundaries, variables that are interrelated with the notion of structure (Schön, 1971; Abbott, 1995). Power in management studies is normally associated with strategy, choices and action towards *equilibrium* and cohesion, that is, how to set and transform structures, the key notions that are explored in the next sections.

2.3.- Network Structures

A network structure is the combination of roles, relationships and boundaries (Abbott, 1995; Schön, 1971). Obviously, structures also embrace network resources such as technology, trust and core competences (Kash & Rycoft, 2000). Boundaries can be defined as limits of behaviour (which has resource-based, strategy, technological and social dimensions), being the result of cooperative decision making as well as investment, which clearly limits and shapes the roles and relationship dynamics, and therefore the structures.

Since networks emphasize autonomy, there may be a variety of actors and roles. In this regard, networks cannot be considered "unitary" or "monolithic" organisations, what should be emphasized is the role of their actors, the reciprocal interactions linked with this, the fit between the actors (at strategic, resource and social levels) and how decisions are made in order to build the structure (Einsenhardt & Schoonhoven, 1996). As such, networks have to be seen as complex systems (Garnsey & McGlade, 2006) which, by definition, *need boundaries to survive and set equilibrium*.

2.3.1.- Equilibrium

Power therefore, has the function to shape Network structures and influence the effectiveness of innovation by maintaining equilibrium within the creative tensions between actors and their conflicting interests, objectives and demands in the network (Quinn, 1988; Shenkar & Zeira, 1992).

Equilibrium is defined as the situation in which the network actors expect to receive more value back, than the risks and costs they incur by being a part of the network (Barnard 1938/1968). In fact, studies on innovation related to equilibrium and change have commanded the attention of theorists for a long time (Gersick, 1991; Weik & Quinn, 1999; Demers, 2007; Lam, 2007).

As such, networks have to be seen as complex systems (Garnsey & McGlade, 2006) where power has the function to shape network structures and explain and locate the limits. Complex structures, by definition, need "rules of the game" to set an equilibrium that leads to a pattern of behaviour which is conducive to survival (Stacey, 2003).

Sociology focuses on how these limits are set: they can be the result of coercion, force and manipulation or of legitimacy, influence, wisdom and signification (Clegg, 1989) which may lead to social consensus to build structures (Parsons, 1937). Consensus is a more democratic (bottom-up) approach which can be understood as the set of constraints or "binding obligation" a society chooses as the rules of the game; this includes sanctions which serve solely as a deterrent (Parsons, 1937).

2.4.- A Classification of Networks of Innovation

After these main definitions, we present a classification of different "Networks of Innovators" based on power This is essentially focused on purposive action and the presence of a focal firm. This classification is based on reviewed literature (Powell & Grodal, 2007; Conway & Steward, 1998) where the criteria is focused on two variables: (1) the existence of a focal firm in their configurations vs. socially emergent relations of all kinds (Scott, 1991; Tuomi, 2002; Bijker & Law,

1992), with little or no managerial action or design, emphasising "political" ties, friendship or work relations (see also White, 2008; Krackhardt, 1992); and (2) the general capacity to innovate (or innovativeness) vs. the purposive capacity to generate one specific innovation, stressing that the locus of innovation, which can be defined as the spot where innovation is "concentrated," can vary from the focal firm to the network (Powell et al., 1996).

The network map presented in Figure 2.2 is derived from traditional static sociogram analysis (or visual structural analysis). It has a quantitative dimension (inclusion, number of ties and density) and a qualitative dimension regarding network components (diversity, sources of innovation, and formal and informal structures). In the figure, and according to literature, the following structural aspects are emphasized: (1) the tendency towards actor isomorphism or differentiation; (2) formal or informal relationships; (3) the type of innovation: continuous, radical, and unique vs. systemic; (4) the variation from top-down managerial processes, associated with task definition, which leads to the creation of relationships and networks.

Focal Firm Centric	Main assumptions: bottom-up and top down; ties and tasks interaction; multi-function Radical Innovation	Incremental innovation – on the existing process Main assumptions: top down; tasks before ties; one function Keiretsu / Networks of Providers
	Strategic Alliance / Differentiated Network Nodes: Differentiated and Autonomous Relationships: Symmetric (peer-to-peer)	The Transactional Network Nodes: dissimilar (complementary). May be similar if there is competition or geographic assignments The work creates the Network
	Social Organization of Specialists Communities of Practice	Project-based Organizations
ocial-Centric	The Network creates the task (Multi-task) Social Innovation (e.g., language) Main assumptions: bottom-up; ties before tasks; multi- function	Incremental and radical innovation Organizations based on time-dependent teams Main assumptions: bottom-up and top down; ties and tasks interaction; one function
	Nodes: Similar Common Social Identity; Conformity; Isomorphism	Nodes: dissimilar (complementary). May be similar if there is competition or geographic assignments

Figure 2.2: Networks of Innovation Source: adapted from (1) Conway & Steward (1998) and (2) Powell & Grodal (2007)

28

2.4.1.- The Primordial Network

The *Primordial Network* is the expression of the emergent and natural social order (White, 2008; Giddens, 1987; Emerson, 1962; Merton, 1959; Freud, 1966). The literature normally assumes bottom-up or *emergent* processes of social formation, with the term "social," referring to little or no managerial action (see also White, 2008; Krackhardt, 1992). These include the structure itself as a result, normally through an unconscious processes of identification.

Identification does not necessarily presuppose common identity or trust among members, since this social process is a primordial enactment that "fills the gaps" with one's internal assumptions projected onto others (Wilson et al., 2008). Primordial Networks can evolve into shared identities, trust and learning (Lave, 1988; Lave & March, 1975) that can be found in *communities of practice*, although they may be based on tacit knowledge and experience when producing innovations (Wenger et al., 200; Wenger, 1998).





Structural Hole Theory

Social Capital Theory

Figure 2.3: Structural Hole and Social Capital Theories

Focal firms, as depicted in figure 2.3, are created by a collective, but they can also find good opportunities for innovation by linking to distant groups, the focal firms serving as bridges (Burt, 1992). Moreover, focal firms can create a context

in which actors exchange more contacts and social capital in order to enhance the appearance of novelty (Coleman, 1990). In these networks, social structures can last for generations, presenting what is called familiarity or embeddedness (Granovetter, 1985) as a process in which identification can be grounded on familiar roles and structures (White, 2008). High levels of embeddedness may lead to resistance to innovation (Sambamurthy, 2000).

2.4.2.- The Networks of Discrete Innovations

The Networks for discrete innovation are formed or designed to develop a specific innovation and can vary from project-based organisations (Söderlung & Bredin, 2006; Schön, 1971) to long-term relationships with strategic purposes (Gulati et al., 2000), depending on the nature of the innovation.

Network duration and dynamics can be the result of purposive action and/or socially emergent processes, combining design and purpose (Doz, 1996). Their classification can range from informal to contractual, representing the degree of purposiveness or calculation and relating decision-making processes to network formation and evolution (Child & Rodrigues, 2003; Kogut, 2000; Merton, 1959). The social centric approach recognizes that these networks are characterised by initial "informality" (Gummesson, 1999; Mintzberg & Van der Heyden, 1999). In this kind of network there is no clear role for the focal firm. At times, strong leadership may emerge, while on other occasions, leadership is a role that can "rotate" throughout the project. In the literature, references to business networks date back to the late 19th century in the form of "industrial districts" or "clusters" consisting of geographically concentrated groups of small or medium-sized organisations (Breschi & Malerba, 2007) which depended on one another for a variety of services, sometimes related to technology (Porter, 1998; Cowan, 2007).

The main risk, however, is precisely the presence of different leaders with *different* perspectives and interests who may wish to establish different rules of the game at the structural level, in terms of roles, relationships and functions. In this case, political mastery is an essential skill, as is the capacity to generate trust

in order to exchange knowledge and resources for a given project or a task (Thorelli, 1986).

2.4.3.- The Transactional Network

Focal firms steer *Transactional Networks* that have a clear and conscious principle of operation that is driven by the task design, the project and a strong focus on efficiency and operational robustness (Vervest et al., 2005). These relationships are normally considered "formal" and "purposive" in most of the reviewed literature, recognising the power relations between the focal firm and the network (Wasserman & Galaskiewicz, 1994; Powell & Grodal, 2007).

Some examples are *Keiretsus* or *Networks of Suppliers*. Normally in these networks, deep structures remain, and focal firms co-ordinate activities to create sustainable incremental innovations to resolve problems (Lasserre, 1992), as well as to create mechanisms to spread innovation (Rogers, 1995). They do this by downsizing-driven activities that are de-layered, core-competence-based and "lean and mean" (Lasserre, 1992). The reasons for their formation include: (1) a firm's focus on its core competencies (Cohen & Levinthal, 1990); (2) the reduction of internal control in favour of externalising certain activities due to their specialisation; and (3) the reduction of the transaction costs associated with dealing with network partners (Hagel III & Singer, 1999; Gurbaxari & Whang, 1991; Williamson, 1994; 1991). Specialisation and differentiation are brought about by the selection of the firm that controls the network and the various components it comprises.

2.4.4.- The Strategic Alliance

Focal Firms steer *Strategic Alliances* when the actors' roles and governance are clearly set out and when relationships tend to be *long-term* (Gulati et al., 2000). Relationships are normally considered "formal" and "purposive," recognising the power relations between the focal firm and the network (Wasserman & Galaskiewicz, 1994; Powell & Grodal, 2007). Strategic alliances for innovation

emphasise one focal firm that exhibits clear managerial intentions and rules of the game to drive core capabilities, while recognizing emergent processes. These can include system integration and open innovation (Chesbrough, 2008) or R&D activities in a particular field (Powell et al., 2005). For this, managerial action focuses on partner selection and *fit*, affecting the alignment of objectives (Jarrillo, 1993), performance levels, cycles of slack (free resources) and investments (Nhoria & Ghoshal, 1997; Huber, 1991; 2006).

It can be argued that differentiated networks can be a natural extension of Strategic Alliances, since they emphasise the resource view on competitiveness. However, in this case, the focal firm can be associated with the role of *Headquarters*, and putting an emphasis on autonomy may lead to more than one differentiated focal firm (one for each node in the network), thus emphasising power dynamics as a key variable to understanding network evolution.

These Networks of Innovation emphasize resource-based views that argue that each actor may succeed by owning specific, rare, inimitable and unique resources, including tangible and intangible ones (Wernerfelt, 1984; Barney, 1994; 2007). The resource view in networks emphasizes "resource bases" such as relationships, governance and trust (Barney, 2007); product and service specificity (Powell, 1990); knowledge exchange (Tsoukas, 1996; Kogut, 2000); network core capacities (such as software production) and complementarities on their resource base (Teece, 1987; Kash & Rycoft, 2000); and social values and culture (Salk & Simonin, 2003; Eisenhardt & Schoonhoven, 1996).

2.4.5.- The Differentiated Network

The Differentiated Network can be seen as an evolution of strategic alliances adding legal boundaries or as an epistemic project to better understand the innovation dynamics of global firms. This approach emphasises a critical dimension for managing innovation: slack, introduced by Cyert and March (1963) in the *Behavioural Theory of the Firm*.


Figure 2.4: Slack and Optimum

Slack is essentially a notion used to define "free resources" such as innovation funds or discretionary payments (see Van de Ven & Hargrave, 2004; Nhoria & Ghoshal, 1997). Slack management is about balance and equilibrium, since, according to Nohria and Ghoshal (1997), innovation and slack have both positive and negative effects: positive, since slack encourages exploration and experimentation (March, 1991) and negative, because (too much) slack may encourage complacency and discourage discipline (see figure 2.4). A connection can be seen between slack and "loosely-coupled" structures. In both cases, their dependence with innovation is an inverse U-Shaped curve, therefore suggesting the idea of optimality. Here, an optimal amount of slack fosters innovation and serves as a point of equilibrium between experimentation and discipline. This optimal amount of slack is a question that every organisation needs to solve case by case (Nohria & Ghoshal, 1997, pp. 64). Structural dynamics is governed by boundaries and deep structures, and slack is normally associated with discretionary managerial decisions (Cyert & March, 1963). However, they both share the idea of optimality and innovation, understood as a response to cope with the administrative paradox formulated by Thompson (1967) in which organizations need to be efficient at exploiting their resources and flexible enough to dedicate free resources, investing them and build something new.

Networks emphasize a dialectical approach, recognising that they are complex structures of command, expertise, commitments and influence where orchestrators exert power to ensure innovation effectiveness, by maintaining the creative tension between the conflicting demands (Quinn, 1988), and ensuring purposive action that combines design and socially emergent processes (Doz, 1996).

The focal firm can be associated with the *Headquarters* (HQ), as mentioned, serving as the guardian of the doctrine, selecting territories for expansion, choosing the methods of diffusion, training people, socialising and incubating new agents, monitoring performance and keeping the network informed (Barltlett & Ghoshal, 2002). However, every node within the network has autonomy, though we can consider that the firm sets the appropriate, nominal boundaries as a focal firm.

Autonomy raises the issue of strategies and structural fit and its linkage to mutual adaptation between actors in the network. Here, two types of adaptations can be distinguished: (1) technical adaptations in product features or the production process and (2) mutual adaptations that bind companies together, often in a direct physical sense, reflecting a mutual commitment which, at the same time, is limited in time and empowers the companies (Hakansson & Snehota, 1989; 1995)

Orchestration here relates to the management of a complex and plural political system. Business units, depending on their autonomy may have the capacity to adapt to every market and social reality, thus leading to differences and local innovations. Indeed, innovation is related to domestic units and their capacity to distribute innovations between actors, as well as how other Headquarter-level innovations can be spread throughout the network, that is, how they engage in an inter-organisational exchange process.

2.5.- Orchestration and Power

We now present the results of the literature review regarding a specific set of Networks of Innovation, that is, those that have a focal firm in their structures. In the next sections, we review power and orchestrating theories, normally crystallised in the dimensions of (1) Centrality and Roles, (2) the control over resources, and (3) dynamic capacities.

2.5.1.- Focal Firms and Centrality and Roles

A role, in abstract, is defined by a position, a function and the relationships with other members of the structure (Alvarez & Svejenova, 2005). In other words, a role belongs to the structure, not to the individual, since roles are ways of enacting a position in the existing game. This is why roles can include status and symbolic components. Power Theory considers "roles" or functions as sources of power (Mintzberg, 1973).

The literature on Orchestration identifies the classic notion of power as *centrality* with the existence of *focal firms* in networks (Ahuja, 2000; Krackhardt, 1990). The assumption is that the more central the node, the greater the influence that node has on overall network behaviour and the more dependent other actors become on the focal firm's influence (Thorelli, 1986; Ferriani et al., 2009). Centrality is based on position or pre-eminence (Wasserman & Galaskiewicz, 1994), legitimacy and signals of status (Molm, 2003), as well as on access to critical resources or knowledge sources (Kogut, 2000; 1988). Centrality is also explored for how it affects network-based teams and performance (Ferriani et al., 2009) or how it is correlated to absorptive capacity (Tsai, 2001) and knowledge recombination (Pedersen & Larsen, 2006).

For example, the literature recognises a focal firm's influence on creating and managing a network of suppliers (Jarillo, 1992); it is also seen as a strategic apex or co-ordinator of inter-firm exchanges (Lorenzi & Baden-Fuller, 1995) or, more recently, as an Orchestrator for innovation (Dhanaraj & Parkhe, 2006).

Focal firms can shape the possibility of creating alliances with such nonredundant partners (Powell et al., 2005). This is based on the notion of "multiconnectedness," that is, the degree to which actors are aligned with multiple role-relationships (Conway & Steward, 1998), becoming more and better informed, and thus increasing their bargaining power and gaining efficiency compared to less central firms (Gnyawali & Madhavan, 2001; Burt, 2004). Moreover, centrality is also related to the need to invest resources and costs for intense co-ordination and administrative activities. It therefore requires a balance between centrality, innovation and performance (Tsai, 2001), as well as the capacity to manage free resources for innovation.

2.5.2.- Roles of the Focal Firm

The literature on Orchestration has metaphorically associated focal firms with *managerial roles* such as leaders, architects, builders, and nurturers (Hinterhuber, 2002) in line with Thorelli (1986), who argues that position and roles are intertwined. However, the most effective innovation leaders do not exhibit a single role; rather, they are characterised by demonstrating a variety of styles and playing other roles in the network, such as those proposed by Kelley (2005): anthropologists, experimenters, cross fertilisers, herders, collaborators, architects, set designers, care givers and storytellers.

An interesting gap in the literature exists between Orchestration roles and leading innovation roles. Van de Ven et al. (2008) propose different innovation roles, in general: (1) the sponsor or entrepreneur who breaks the rules of the game and proposes new ones; (2) the critic, in opposition to the first role, tending to assume a conservative position, challenging innovation; (3) the institutional role which settles disputes and structures; (4) the mentor role, supporting the process of innovation; and (5) the sponsor who procures, advocates and champions the initiative (see Figure 2.1).



Figure 2.1 Different Innovation Roles

2.5.3.- Control over resources

According to reviewed literature, Orchestrators can exert control over these resources by: (1) *centralising* decision-making and becoming a strategic apex in the network (Lorenzi & Baden-Fuller, 1995); (2) controlling some strategic assets that may provide a competitive advantage (Wernerfelt, 1984); (3) bridging structural holes (Burt, 1992); (4) applying sanctions where needed (Jarillo, 1993; Powell et al., 1995); and (5) managing *network processes* such as resource selection and mobilisation, thereby facilitating co-operative strategies (Häcki & Lighton, 2001; Brown et al., 2002) and managing global operations (Fung et al., 2007).

Orchestrating therefore, can be explored by the effective management of control of network resources such as (1) position, centrality and managerial roles (already described); (2) the relational dynamics between actors in the network structure; (3) trust management and (4) digital platforms, since innovation is (increasingly) dependent on digital assets, communication and interactions (Von Hippel, 2005), aspects that are covered in the next sections.

2.5.3.1.- Relationship dynamics

In Managerial Science, Teece et al. (1997) define relationships as a *dynamic resource* based on interactions, communication and exchanges. Relationships start at the individual level and can be defined by their purpose, that is, they are the result of choices made and the intentions to engage in exchanges, as well as the latter's level of formalisation, intensity and frequency (White, 2008; Granovetter, 1985; Aldrich, 1979). Much of the literature focuses on exploring the relationship between formality and innovation (see Powell & Grodal, 2007; Mintzberg & Van der Heyden, 1999; Gummesson, 1999). While formalization depends on a set of norms and criteria to define what is formal and what is not, the notions of intensity, strength, and frequency can only be explained observing time evolution and dynamics.

Although relationships are normally represented as "lines" due to the profusion of graph theory (see DeBresson & Amesse, 1991; see also, Conway & Steward, 1998), the real meaning of relationships is expressed through their dynamics which shape actors' behaviour and the relationships between them. Here, the preferred unit of analysis is power (Emerson, 1962; 1987; Abbott, 1995), focusing on how actors exchange resources and develop dynamics of dependency, in other words, symmetries and asymmetries. A relationship between actors A and B who exchange the respective resources x and y can be represented by a dyad such as Ax-By. The availability and value of the resources determine the power and dependency between A and B. The more dependent B is on x, the more power A has over B and vice versa (Emerson, 1962). Social exchanges not only include tangible assets, but also capacities such as affection, approval and status (Molm, 2003). Because power is dependent on its value to others, it is an attribute of the relationship, not only a question of the actors or the resources since what is valuable for one actor may not be so for another.

Power and innovation are thus related to: (1) the nature and context of the relationship and its function, taking into account information and exchange of goods; (2) its formalisation, that is, the relationships that are managerially designed or emergent; (3) being competitive or cooperative (Child et al., 2006);

and (4) the intensity, strength and frequency of interactions which are normally shaped by emotional attributes. Emotions are taken into account when scholars differentiate between "strong" and "weak" ties (Granovetter, 1973; 1985) which are the translation of Bowlby's Theory of Relational Dynamics to Sociology.

This is based on: (1) attachment or the need to be close to someone on whose life we are dependent, a need which has developed through nurturing; (2) caregiving, that is, the pleasure of taking care of others; and (3) sex as attraction or desire, which leads to the need for fusion and completion (Bowlby, 1962; 1980; 1985; see Aron & Westbay, 1996).

Theory of Attachment	Theory of Strong/Weak Ties
Attachment	Trust
Care-giving	Commitment
Sex	Intimacy

Table 2.1: Comparison between Attachment and Strong/Weak Ties

Moral approaches include "positive" vs. "negative" dynamics normally associated with trust and/or friendship (Krackhardt, 1990). Negative relationships are based on lack of trust, opportunism and visceral human passions such as envy and fear (Elster, 1989). A negative relational dynamic reduces the frequency of interactions, while a positive one increases the exchange frequency (Cook & Emerson, 1984).

Maybe one of the most critical indicators of relational dynamics (and power) is *reciprocity,* focusing on the *underlying structure* of the relationship, which can be termed *asymmetric* or *unilateral* when the flow is one-way, and symmetric or *bilateral* when the flow is two-way. Asymmetric relationships imply inequality in power relationships, and symmetry implies peer to peer relationships (Pedersen & Larsen, 2006; Krackhardt, 1992).

Symmetric and asymmetric dynamics shape knowledge exchanges, routines and relationship-specific assets requiring a dynamic approach to purposefully understand relational dynamics with innovation (Dyer & Kale, 2007). Therefore, dynamic capacities stress how competitive advantages can be obtained when

firms move away from "arms-length relationships" to create "purposeful idiosyncratic relationships" that can facilitate new combinations of resources and capabilities (Teece, 2009; Dyer & Singh, 1998; Almeida & Kogut, 1999).

The above discussion emphasises that a central problem in developing a scientific theory about relationships is the intertwined and underlying economic, social, cognitive and emotional processes involved. Similarly, social dynamics and innovation dynamics differ, and people can use an almost infinite variety of units of analysis, causes and reasons to explain relational behaviour.

2.5.3.2.- Trust

The Neo-Classical perspective does not consider fairness or equity in decisionmaking processes, therefore omitting the critical role that trust plays in network dynamics (Ring & Van de Ven, 1994; Teece, 2009). This is simply because it considers innovation to be dependent on investments, learning capabilities and knowledge transfers between firms (Gilsing et al., 2008), which require some degree of exposure (Zollo & Winter, 2002; Contu & Willmott, 2003). Resourcebased views recognise that trust can be a source of competitive advantage depending on how actors invest in governance schemes and how they set relational dynamics and interactions (Barney, 2007).

Any innovation requires long, strategic relationships in order to create new knowledge. The latter, in turn, requires interactions and exchanges whose nature is essentially tacit (DeBresson & Amesse, 1991). Transaction Cost Economics – although important – fails to grasp the whole phenomenon, as it is biased towards controlling opportunism better. For example, Pedersen and Larsen (2006) suggest that the exchange of "innocuous knowledge" is one process to engender trust, and which allows actors to exchange non-competitive or Intellectual-Property-based knowledge and to later share, combine and co-create new knowledge to change perspectives and innovation structures.

Clemmensen et al. (2008) provide a set of constructs or definitions from different perspectives. These are reproduced in Table 2.2 with some comments (see Clemmensen et al., 2008 for the complete set of references).

Basis of Trust	Construct
Rational Choice	A rational actor only bestows trust if his/her calculations suggest that the benefits of reciprocated trust are higher than the losses implied by a betrayal.
Cognition-based and Diversity	Cognition-based trust is grounded on the factual knowledge that the party doing the trusting has about the trustee.
	Acceptance of diversity – the good and bad side of any relationship.
Value/Norm-based	Trust cannot exist unless individuals share the same values.
Affect-based trust	Emotional trust consists of the emotional bonds between people such as those provided by the relational dynamics in care, attachment and passion.
Expectation-based	Identification of the other's expectations.
Impersonal trust	Impersonal or passive trust is given to a social institution.

Table 2.2: Trust Constructs. Source: Clemmensen et al. (2008)

Networks are inter-organisational forms, which, dependent on trust, are superior to exchange transactions and the internalisation policies predicted by classic Transaction Cost Economics. In consequence, networks can serve as a search and evaluation procedure to evaluate the other parties' technologies while ensuring non-cash agreements on technological know-how (DeBresson & Amesse, 1991).

One actor's exposure may entail the risk of exploitation by another actor or the non-alignment between competitive and collaborative behaviour. Child et al. (2006) refer to competitive learning processes as opposed to co-operative ones in which actors can take advantage of other actors' knowledge and not give back anything in return. Elster (1989), by contrast, sees opportunism as a major force for cohesion in social behaviour regarding access to scarce resources. Organisation Theory has devoted a lot of attention to controlling opportunism through *formalised* arrangements or contracts inside and between firms (Eisenhardt & Schoonhoven, 1996). The classical approaches of Agency Theory and Transaction Cost Economics are aimed at formalised organisations and networks, respectively (Eisenhardt, 1989: 2). In contrast, with the risk of

opportunism found in markets, the actors engaged in a network, in any type of relationship, *expect goodwill* from the others, since relationships require investments whose returns or benefits are postponed in time (Thorelli, 1986).

2.5.3.3.- Digital Platforms

A digital platform is a set of modular components that, on the one hand, offers a standard framework and, on the other, facilitates plug-and-play options (Van Heck & Vervest, 2007). The contribution of any actor to the digital platform makes the network dependent on: (1) the platform functioning; (2) how the network builds and develops it (Von Hippel, 2005; Delporte-Vernieren et al., 2004); and (3) the way inter-organisational processes are digitalised and the correct definition of interfaces among firms (Vervest et al., 2008). In addition, a digital platform offers shared processes, interoperability and compatibility (Alstyne, 1997). The first obvious effect is noise reduction and lower transaction costs among firms (Gubaxari & Whang, 1991).

The author suggests considering digital platforms as a proxy in a Network's *shared investments for innovation.* Indeed, a digital platform may consist of a platform based on digital assets (such as web services) which can be co-created and co-produced, thus underscoring the *role of technology as a space for creativity and co-creation.* This idea highlights the digital platform's role as a resource to contextualize and make sense of the ongoing network activity (Orlikowski, 1996; Orlikowski & Iacono, 2001), but it is also a means to propose new functionalities and compose them with the existing set of digital assets (Gawer & Cusumano, 2008).

Since innovation is a process of stepping into the unknown, dependencies arise in terms of information exchanges and knowledge generation, thus highlighting the importance of digital platforms in network dynamics for innovation. Digital platforms can be seen as infrastructures for innovation (Lyytinen & Newman, 2008; Van de Ven et al., 2008), particularly in networks, since they relate to power, structural resource and working systems (Mason et al., 1997; Day & Shoemaker, 2000). Moreover, digital platforms are the arena in which to carry on with research on alternative ways for software development. This is evident in Open Source communities (Von Hippel, 2005; Raymond, 2001) or agile developments which can be found between too much and too little structure, focusing on collaboration and quick (or agile) adaptive responses to requirements (Kautz & Zumpe, 2008). Indeed, structures and working systems shape and are shaped by information systems, since communications are based on informationrelated work (Orlikowski, 1996).

In this context, actors' choices to co-create and deploy digital systems are not the result of agency, teleology or blind adaptation (Lyytinen & Newman, 2008). We thus consider digital platform as a network structural communications system (Orlikowski & Iacono, 2001) that can foster togetherness and trust, allowing information to become knowledge (Kogut, 2000; Kogut et al., 1992) enabling actors to absorb and recombine this knowledge base (Cohen & Levinthal, 1990; Teece, 1987) and, thus, permitting the re-combination of the network's digital resource base, and becoming the means through which the actors' interactions can create new paths.

On the other hand, the social constructivist views stress the capacity to develop innovation throughout social interactions (Elkjaer, 2003; 1999; Tuomi, 2002; Bijker & Law, 1992). In any case, digital platforms in this thesis are considered to be an outcome of the interaction process among actors in a network and a matter of strategic importance, since network and digital platform interactions shape the opportunities to develop options and strategic change and structural design (Easterby-Smith, 1997; Orlikowski, 1991).

In this specific view, strategy links with innovation for: (1) the creation of new platforms and attracting others to develop new add-ons, and (2) expanding markets towards the existing company platform (Gawer & Cusumano, 2008). Creation, co-creation, and the management of add-ons (complementors) are the essence of dynamic capabilities and networks for market / factor creation, where Orchestrating Networks for innovation is an important governance mechanism.

2.5.4.- Orchestration and Dynamic Capacities

Strategic management and dynamic managerial capacities highlight the importance of managerial action and power to purposefully create, extend and modify the firm's resource base as processes of becoming (Teece, 2009; Eisenhardt & Martin, 2000). For that matter, Orchestration is presented in the existing literature as a *network capacity* to dynamically organize innovation (Dhanaraj & Parkhe, 2006). Ritala et al. (2009) propose an *Innovation Orchestration Capability* for "future-oriented value creation" (p. 572) since value creation is dependent on the ability to connect different experts and specialists and enable them to work together (Dyer & Singh, 1998). All these authors analyse Orchestration at the organisational and individual levels, emphasising the fact that individual skills are to individuals what organisational capabilities are to firms.

This means that individuals and organisations can be separated at the analytical level although they are interrelated, since individuals can shape organizational dynamics and relational capabilities which are increasingly important in an interconnected business world, and are seen as sources of differentiation (Dyer & Kale, 2007).

In particular, the literature of dynamic capabilities emphasizes as network capacities (1) asset orchestration; and (2) value generation and appropriation which are highly interrelated. There are additional related notions according to this view, which are described briefly in the next paragraphs.

2.5.4.1.- "Asset Orchestration"

The concept of "Asset Orchestration" proposed by Teece (2007; 2009) emphasises: (1) the co-ordination of disparate actors without "central planners" (Hayek, 1949); and (2) the combination of emergent collective processes and managerial design (Doz et al., 2000). This notion of governance highlights autonomy (see also Benkler, 2006), and is intended to combine complementary and co-specialised assets among firms (Teece et al., 1997). Pedersen and Larsen (2006) extend this framework to network models of innovation, precisely

underscoring the actors' autonomy and self-control, and the network's capacity to re-combine its resource base, grounded on knowledge dynamics between actors (Cohen & Levinthal, 1990).

2.5.4.2.- Value generation and appropriation

Dhanaraj & Parkhe (2006) hone in on the focal firm and specifically define *Orchestration for Innovation* as the capability to build and purposefully manage inter-firm innovation networks, suggesting a function of value creation and absorption. Orchestration is about generating network externalities and *increasing returns*; this is how network dynamics "expand the pie", by focusing on the orchestrator's capacity to extract *more value* from the network.

Implicitly this approach is related with above discussion on power and the notions of centrality, roles and the capacity to control resources. Dhanaraj and Parkhe discuss how the Orchestrator may have the reputation of attracting members to the network and thus obtain a bigger piece of the pie by managing knowledge mobility, trust, socialisation and shared identity to ensure the stability of network dynamics.

Trust is needed to allow an actor to be exposed to sharing knowledge for a common goal, and to cope with different "enactments," understandings, perceptions and interests in order to be able to propose and combine them with others. The role of the Orchestrator is thus to create an open atmosphere of communication by helping to discuss and establish the basic rules of the game. Trust diminishes the threats of the actors' opportunism, and fosters an equitable distribution of value (Dhanaraj & Parkhe, 2006).

2.5.4.3.- Boundaries and "Complementors"

Chesbrough (2008) argues that the sources of differentiation and even the *firm's boundaries* are dependent on the ownership of knowledge, the management of complementary assets and Intellectual Property (IP). Protection. Value appropriation is dependent on: (1) the nature of knowledge and how replicable it is, with an emphasis on Intellectual Property (IP) protection, and (2) the degree of complementarities needed for a particular innovation, its nature and the

capacity to manage them internally or/and externally through open innovation processes (Chesbrough, 2003; 2006). To this end, complementary assets can be generic - and easy to replicate and specialised and/or co-specialised (see also Williamson, 1995). The economic value of these complementary assets is the sum of their generic use, that is, their specific use plus their residual value in their next best alternative use.

As a consequence, when firms own relevant complementary assets, the outsourcing or licensing of some technologies may be "virtuous" because the firms may exert power over the others and influence how the pie expands, through licensing, the creation of secondary markets and by being able to commercialise an externally accessed innovation (Chesbrough & Teece, 1996).

2.6.- Conclusion: Combining Power and Orchestration

In this chapter, I have reviewed and presented literature about Networks of Innovation and Orchestration as one way to govern networks. Networks are structures grounded on exchanges where power and trust shape the roles, relationship dynamics and boundaries that lead to specific structures. In this chapter, I have also presented existing network structures with focal firms and orchestrators, observing how literature shows that networks with focal firms are associated with *diffusion for specific innovations*, stressing centrality, control over resources and "node-anchoring" dynamics. In contrast, literature that presents the network as the locus of innovation assumes no specific focal firm and an implicit correlation with "innovativeness" as the network capacity to generate innovations. It is also suggested that networks designed for general "innovativeness" assume the capacity to generate core competences as some degree of stability, trust and embeddedness.

The reviewed literature on Networks of Innovation has a bias towards static structural approaches. In contrast, the literature on strategic management and dynamic capabilities emphasises the importance of Orchestration in developing *future paths for profitable innovation* but it is almost silent about (1) how Focal Firms and Orchestrating affect structural changes, that is, boundaries, roles and relationships and (2) although in innovation studies there is a tradition of setting *trajectories and paths* (Pavitt & Steinmuller, 2002; Pavitt, 1984; Dosi, 1982), this line of thought has a bias towards path dependencies, determinism and teleology where research focuses on how Orchestrating, as a network capacity, can set a *new* path.

In order to fill these gaps, this thesis relies on the proposal of the *managerial* function of Orchestrating that focuses on strategic management to examine network structural dynamics and a process view to better explore how managerial action can change the "rules of the game" governing how actors interact and change their relational dynamics, roles and boundaries. To do so, this thesis is grounded on *managerial power* as the resource to drive change and manoeuvre between flexibility and loose alignment and, at the same time, provide efficiency, certainty and stability (Thompson, 1967/2003; Chandler, 1962). With this approach, Orchestrating focuses on the managerial capacity and power to set strategies and, as a consequence, consider innovation as the capacity to set paths for efficiency between the proposed strategies, while stressing decision making for the appropriate actions and resource mobilisation in the design of new structures for future organisations. The managerial function of Orchestrating also considers the importance of digital platforms as structural resources that play a strategic role in innovation since they emphasise the importance of having *reliable information and knowledge* with which to make choices when designing organisations (Galbraith, 2005; 1977).

In accordance with previous streams of innovation, structural dynamics and dynamic capabilities, the next chapter of this thesis proposes the function of Orchestrating that is discussed and researched in Chapters 5 and 6 (Chapter 4 contains the research design). It will present empirical work and adopt a propositional style by comparing existing literature with empirical findings to fill the gap presented.

Chapter 3: Orchestrating Network Behaviour for Innovation

3.1.- Introduction

In this chapter the managerial function of Orchestrating Network Behaviour for Innovation is defined as establishing a path towards innovation. The role of the focal firm or Orchestrator and its capacity to manage structural dynamics and digital platforms is specifically proposed. This researcher takes a dynamic capacities perspective to analyze structural change and examine the change to boundaries, relationships and roles, as well the role of information and communication technologies (ICTs) in structural changes (Lyytinen & Newman, 2008; Kautz & Zumpe, 2008).

This chapter is organised as follows. First, the concept of Orchestrating Network behaviour for innovation is defined. Second, we incorporate the concepts introduced in chapter 2 and developing them in a dynamic context to extract the meaning of optimality for "loose coupled structures," defining what a path of innovation is in this context. Thereafter, the main components of the function are described. The themes of structural dynamics and innovation with digital platforms are highlighted and developed in further chapters of this thesis. Finally, Orchestrating, at an individual level of analysis introduces managerial capabilities such as organization, conflict management and economics.

3.2.- Orchestrating Network Behaviour for Innovation

Orchestrating Network Behaviour for Innovation is defined as a function to build a *path* towards innovation (figure 3.1). It is assumed that for innovation it is up to the managerial function to dynamically manage equilibrium, requiring choices and purpose. These paths are considered to be expressions of network dynamics, thus implying that technological and social dimensions evolve by influencing one or the other. The success of changes is dependent on how this path is set and through which decisions and actions aimed at managing degrees of freedom in the network, that is, which boundaries or constraints.

In this context, strategic management is considered to be a critical network managerial capacity to fix a purposive set of actions to build a path for all the actors in the network. Innovation is enabled by shared investments and shaped by a set of external constraints which need to be managed or orchestrated in order to set a course of action based on choices and activities in a context of radical uncertainty (Cheng & Van de Ven, 1996). It is in this context where orchestrating is placed as a managerial capacity to exert power to deal with structural dialectics. Orchestrating Innovation considers that paths are not predefined trajectories that need to be set by fixing and regulating constraints, since continuity may be challenged by tensions among autonomous actors with different interests. More specifically, the role of the focal firm and its capacity to control relational dynamics and other resources such as technology and dynamic capacities at the structural level are explored, suggesting that different conditions can result in different patterns of network evolution.



Figure 3.1: Network Behaviour for Innovation

To that end, orchestrating is a function of (1) changing the rules of the game to keep the game alive and (2) avoiding disruptions, thus ensuring continuity through the connection between the old and the new, or between past behaviour patterns and the present and future ones. The hypothesis is that this function has a positive influence on the network's structural innovative capacity or innovativeness, and takes advantage of digitally-enabled processes (see Figure 3.2).

Orchestration shows power through its influence on relational dynamics, roles and network boundaries. With this argument, it is stressed that decision-making and action-setting are always processes of balancing different forces, as all the actors in the network may exert a certain degree of power and influence since innovation does not necessarily end with performance improvements.

Power in this function underscores the role of strategic leadership through "how" things are made through choices and managerial actions of design and adaptation (Augier & Teece, 2008), recognizing the flexibility of setting direction and incorporating other views, other perspectives, and/or other emergent actions. It represents setting a path by managing optimality and equilibrium (Eisenhardt & Schoonhoven, 1996; Eisenhardt & Martin, 2000), thus the governance of different optimal situations can be achieved along the way, followed by sub-optimal ones and the transitions in between (Brown & Eisenhardt, 1997), giving shape to cycles and fluctuations in an ongoing equilibrium.



Figure 3.2: Orchestrating Network Behaviour for Innovation as a Managerial Function

3.3.- Networks as Optimum configurations for innovation

To ground the managerial function presented in this thesis, it is assumed that Networks are *optimal configurations or structures for innovation* (Ferriani et al., 2009; DeBresson & Amese, 1991; Conway & Steward, 1998). The notion of optimality comes from the common consideration that too little structure makes it difficult to manage change, while too much makes it hard to move (Burns & Stalker, 1961; Lawrence & Losch, 1967). To explain this, it is necessary to present the concept of loosely-coupled structures (Orton & Weik, 1990; Brown & Eisenhardt, 1996) as lying in between unstructured chaotic markets, and too much power leading to highly controlled bureaucracies (Thorelli, 1986; Powell, 1990).

The author here suggests that too much and too little structure leads to chaos, or negate the capacity to organize. Chaos is normally associated with little or no structure. However, according to mathematics, chaos has another meaning, implying a state of "bounded order." Order and chaos cannot be explained as graphs of dots and lines, but according to structural dynamics, too many or too few constraints reduce the levels of freedom and thus lead to deterministic patterns of behaviour (Van de Ven et al., 2008), which become chaotic.

Brown and Eisenhardt (1997) emphasise the dialectic of power between cohesion and autonomy when they describe networks as "semi-structural, lying midway between highly structured or mechanistic organisations and unstructured organic organisations with few responsibilities and processes" (p. 28). I interpret that the extremes, no structure or tight structures, tend to be chaotic (non-predictable and deterministic respectively), where loose coupled structures incorporate the capacity to change and set paths, thus being the optimal structure to fix order as a continuous process of organising (Bohm & Peat, 2002), as depicted in Figure 3.3



Figure 3.3: Order and Chaos

The latter has an interesting connection to administrative theory and psychology. Although there are many configurations and structures, business organizations reveal an important bias towards mechanistic processes (Hamel, 2007), often associated with greater operational efficiency than is normally evident in largescale operations. This way of organising human endeavours drew the attention of early sociologists who studied how labour conditions affected employees' emotional moods. Their conclusions – drawn on studies from the 19th century – were that creativity is dependent on power intensity: too much control and too few rules produce "anomic forms of division of labour" (Durkheim, 1937) or, according to psychodynamics, the repression of creativity (Winnicott, 1965; 1971; 1988), and the consequent destruction of innovation. Anomie can be defined as a subjective sense of despair and uselessness due to the interaction with social context, also having clinical implications for mental health (Kets de Vries, 2006; Winnicott, 1988), which are out of the range of this thesis. While early on, Durkheim focused on "too much structure" to explore anomie, Parsons (1957) used this term to study "too little structure" and innovation in society, which, according to him, could be considered as deviant behaviour since it broke the rules of the game, making society unstable (Parsons, 1937).

With the power perspective chosen in this thesis, it is recognized that networks change through a dialectical process, reflecting the natural tensions among actors to control resources and to set direction. These dialectical dynamics also recognise that networks are complex structures of command, expertise, commitments and influence, and this author posits that power studies show their influence on relational dynamics, roles and network boundaries. These aspects are reviewed in the following sections in which the components of the presented function are elaborated upon: (1) strategic management and power through commitments (section 3.3); (2) network structural change and equilibrium (section 3.4); (3) digital platforms as infrastructures for innovation (section 3.5) and the (4) Network Leader describing organizational capabilities, conflict management and economic capabilities (section 3.6).

3.4.- Power and Commitments

This author proposes that the orchestrator specifically exerts power through *commitments*, that is, a combination of power and trust (Elster, 2000; Ghemawat, 1997; Bowlby, 1962), the latter serving as the *raison d'être* to look into the future: to try to project this future before leaping into it. Commitments emphasize their relevance in uncertain environments (Cook & Emerson, 1984) and the need to establish "psychological contracts," stressing mutual understanding among actors in network activity (Child et al., 2006). In turn, network activity requires setting prerogatives and obligations, thus stressing the intense psychological effort and patience needed to achieve the required trust to set commitments. It is also assumed that trust is dynamic or time-dependent, as are the essential relationships. It consists of different phases: (1) deterrent and calculative at first, (2) based on knowledge exchange and representations, and (3) identity-based in more evolved relationships (Shapiro, 1987; Clemmensen et al., 2008). We turn now to explore some pre-conditions to commitment.

3.4.1.- Autonomy and Creativity

The main pre-condition for commitment in this function is autonomy (or *freedom*) and creativity as presented and discussed in chapter 2. It is a classic argument that individuals and even societies have constrained their freedom with explicit rules of the game (Huizinga, 1927/2007) to protect the very freedom and the individuals that constitute the society itself (Popper, 1957; Merton, 1949; Freud, 1966/2007:1; 1966/2007:2). This is particularly to protect humankind from visceral passions through shared consensus and limits on behaviour. On the individual level, we have argued that creativity needs constraints in order to be productive.

3.4.2.- Path Dependencies and Deep Structures

The second pre-condition is memory. As in any other organization, in a company, the past is stored in organisational memory which expresses itself in many ways, but this thesis is interested in the dynamic: in existing behaviours, processes and patterns. Some scholars explain change with the notion of "deep structures" or the set of fundamental choices a structure has made (Gersik, 1991) that provides a certain behaviour. Past choices affect roles, relationships and boundaries, as well as other structural resources. In contrast with popular wisdom, memory not only resides in "images," "myths" or "culture." Memory resides in deep structures, in behaviour, which is highly stable due to: (1) the tenacity of initial choices related to the beginning of the existence of structure and self-survival mechanisms (Eisenhardt & Schoonhoven, 1990); (2) the fact that structural behaviour is reinforced through feedback loops (sometimes called organizational learning) to keep the structure intact or subject to short-term adaptations (March & Simon, 1958; Argyris & Schön, 1978; 1996); and (3) symmetries and asymmetries in relational dynamics affecting deep structures (an aspect I develop in following paragraphs).

We argue that deep structures can be thought of as the design for the rules of the game where equilibrium periods can be compared to a game in play (Gersik, 1991; Huizinga, 1927/2007; Winnicott, 1971) in which actors play some roles and establish some relationship dynamics within certain boundaries of behaviour, and where the benefits of playing are greater than the costs and risks associated with doing so. As argued, roles are linked with relational dynamics where power is analyzed through *reciprocity*, termed as *asymmetric* or *unilateral* when the flow is one-way and symmetric or *bilateral* when the flow is two-way. As argued, asymmetric relationships imply inequality among actors and, as a consequence, a latent conflict.

Based on this approach, the notion of "organizing at the edge of chaos" obtains full meaning as an optimum proposition. Too many constraints reduce the degree of freedom, and the dependency with the past is so strong that structures can only take pre-determined steps towards teleology. With no constraints, network behavior is unforeseeable. In both cases, deep structures are chaotic, and the required value equality cannot be fulfilled in the long run.

In social systems the issue is "how" power set boundaries, from consensus to repression. The resulting behaviour generates what is called path dependencies, or in other words, the future of a structure is dependent on the existing one (Pavitt, 1984; Pavitt & Steinmuller, 2002).

3.5.- Network Structural Change

Orchestrating through commitments is about: (1) having few choices; (2) maintaining consistency with respect to these choices to direct the course of action; (3) holding conflict inherent to any change; and (4) a dynamic capacity to set constraints to broad patterns of network behaviour (Ghemawat, 1991).

3.5.1.- Dynamic Optimums

The objective is to search for an ongoing equilibrium. Optimums are not a teleological "final objective" as critics of rational choice argue (Elster, 2000; 1994). Rather, orchestrating is a process in order to set the path, along which different optimal situations can be achieved, followed by sub-optimal situations

and the transitions in between (Brown & Eisenhardt, 1997). This process thus shapes the different cycles and fluctuations as depicted in Figure 3.4. Commitments require interacting with others and exploring the unknown to manage investments and create new meanings. This can be achieved by creating experiments and designing new steps to be taken.



Figure 3.4: Network Dynamics as Local Optimums and Transitions

Innovation is a process of experimentation: outcomes are never the best (of all) options. They are more a modest set of local maximums such as "A" and "C" (in Figure 3.4) along a continuous path of improvement. Innovation may also go through troughs "B" and "D," recognising that change can include a reduction in performance due to these transitions.

In fact, studies on innovation related to equilibrium and change have commanded the attention of theorists for a long time (Gersick, 1991; Weik & Quinn, 1999; Demers, 2007; Lam, 2007). According to these studies, organisational changes can be explained through teleology, life cycles, dialectics (power dynamics) and evolution (Poole & Van de Ven, 2004) and how equilibrium and change occurs: (1) gradual change, where there is a process of on-going adaptation, and (2) punctuated equilibrium where new "species" arise abruptly, through sudden "punctuations" or revolutions (Gersick, 1991; Weik & Quinn, 1999). As Gersik notes, human structures may look turbulent though the

underlying structures are stable, while hypothesising that others may look stable for a period of time, but they can mask deep structural changes that suddenly burst in punctuations. The assertion is that a change from one game to another is through transitions - at times through upheavals - leaving the structure temporarily disorganised and necessitating the making of choices around which a new deep structure forms.

Dooley and Van de Ven (1997) suggest that innovation dynamics can be expressed as a set of combined phases of divergent and convergent behaviour, which explains the dynamics described above, of the peaks and troughs, respectively. Convergence means a synthesis towards equilibrium, narrowing the possibilities or constraints for purposive action. In contrast, divergence is about searching, "branching behaviour," experimentation and the exploration of new possibilities (Dooley & Van de Ven, 1997; see also March, 1991).

3.5.2.-Centripetal and Centrifugal forces

In this thesis network boundaries is defined as a *set of fundamental commitments* between network actors, commitments which shape the network's structural traits: roles, relationship dynamics, complementarities, the technology deployed (e.g., a BNOS), and co-ordination mechanisms (Thompson, 1967/2003; Malone et al., 1987; Montgomery & Oliver, 2007). This research takes into account the fact that boundaries, relationship dynamics and roles are interrelated variables, not independent ones (Abbot, 1995).

We argue that orchestrating is intended to create competitive advantage through the careful selection of commitments with specific partners, choosing the type of relationship and the deliberate investment in these features. Indeed, the notion of relation-specific assets in networks refers to the assets of a partner who customises them in order to adapt to another partner and facilitate a systematic frame of innovation (Dyer & Kale, 2007). In order to study this phenomenon, we use the tensions between centrifugal and centripetal forces as the units of analysis since they shape the network structure (Montgomery & Oliver, 2007). Centripetal forces dominate the network when actors feel motivated to be a part of it. Relationship dynamics will tend to be cohesive, but they are the result of a combination of factors, including economic aspects and social opportunities (Eisenhardt & Schoonhoven, 1996).

Relationships are not necessarily intended for investment, as argued in this thesis. Thus, we take into account the fact that cohesive forces in social action may hide a conflictive nature, such as many forms of opportunism, which can be defined as the asymmetry between intentions, and what is given and what is received (see Elster, 1994; Child et al., 2006). As argued, innovation is a different game, one where relationships require commitments in order to invest and engage in co-creative activities. Powell et al. (2005) focus on power to explore network creation processes in innovative ventures. They propose several attachment dynamics: (1) accumulative advantage; (2) homophily or isomorphism; and (3) "follow the trend." First, accumulative advantage occurs when a node receives a disproportionate number of new connections. Second, with isomorphism, new partners are chosen on the basis of their similarity. Third, network expansion follows a herd-like process in which choices are made, depending on others' choices, either as a response to external pressures or as an imitative behaviour.

Centrifugal forces, by contrast, can encourage actors to weaken the relationship and their commitments, or leave the network altogether if any of the strategic or resource complementarities offer a poor fit and lead to competitive relations more than do co-operative ones.

Centripetal and centrifugal forces may vary along the way, depending on the capacity to reach commitments. This can affect: (1) the actors' strategic position; (2) their access to new or complementary resources or new markets; (3) new sources of knowledge (Kogut, 2000; Jarrillo, 1993); (4) co-ordination costs and risks (Williamson, 1991; (5) their capacity to search for complementarities (Teece, 2009); and (6) their social values and trust. From a dynamic view, we can also see that centripetal forces lead to cohesion when there are positive economic returns (Teece et al., 1997), stability, network externalities and lock-in (Shapiro & Varian, 1999), while sources of change can unbalance the equilibrium

as a centripetal force. Moreover, where a network can go (its future structure) is a function of its current structure (Teece et al., 1997). This, in turn, is a function of its capacity to change or establish new paths, and overcome the forces of resistance to change due to structural rigidity or over-embeddedness (Sambamurthy, 2000; Granovetter, 1985).

3.6.- Digital Platforms as Innovation Infrastructures

The cases analysed as part of this thesis are based on information-related work, stressing that actors' choices are not the result of agency, teleology or blind adaptation. We refer here to the fact that digital platforms are a means through which the actors' interactions can create new paths. It is understood that digital platforms are the result of many accretions which influence each other over extended periods, thus permitting the re-combination of the network's digital resource base with a focus on agile developments (Kautz & Zumpe, 2008) and becoming the means through which the actors' interactions can create new paths.

It is important to note that we have been differentiating between "systems" and "structures". Up to now, information systems have emphasized clearly set interfaces to establish limit and integrability mechanisms and tools between the network system and the different actors (Delporte-Vernieren, 2004); and providing information transparency and tangibility for network automated operations, as well for generating more information to innovate (Zuboff, 1985).

Very recently, Miller and Olleros (2009) have proposed that the *degree of tightness* among software components may lead to different strategies around digital platforms from (1) high system integration (e.g., software commercial platforms) to (2) modular system design. The second emphasises the co-creation of digital platforms that should: (1) solve a critical business problem, developing a function or a set of functions that can be described as the "system of use"; (2) be easy to connect and disconnect or easily present plug-and-play options (Van Heck & Vervest, 2007); (3) include an embedded network logic supporting inter-and intra-organisational processes and control variables; (4) come with a

standard and modular architecture, a "Lego-like" structure such as web services that can be combined in specific ways.

3.7.- The Network Leader

Leadership is a critical activity in innovation (Mintzberg, 1973; Kets de Vries, 1980; Van de Ven et al., 2008; Álvarez & Svejenova, 2005). The (few) choices available to develop a consistent continuity in terms of strategy (or pattern of behaviour) lead to steps or experiments that can be consistently small or punctuated by a few large ones (Ghemawat, 1991). Triggers may come from exogenous forces, such as changes in technology, regulations, institutions or endogenous forces, such as a customer demanding better performance or new solutions (Miller & Olleros, 2007; Doe & Schoemaker, 2000).

Commitments stemming from choices are the basic argument of *binding rationality*, that is, the freely chosen decision to bind oneself to proceed. This idea is proposed by Elster (2000:1; 2000:2). The latter uses the metaphor of Ulysses "binding himself to the ship's mast" in the famous episode with the Sirens, recognising his own weakness upon hearing the sound of the sirens' songs but also showing his courage by carrying on with his journey to Ithaca.

Taking this perspective, we consider strategic leadership from an individual level of analysis (Ritala et al., 2009) and define three dimensions of Orchestration grounded on the managerial functions proposed by Pedersen (1996):

- Organisational Capacity: managing choices and actions regarding how to select partners; how to manage relationships; how to achieve structural and strategic fit, and managing co-operation for co-creation while keeping the network united;
- (2) Economic Capacity: how to make investments together (Gilsing et al., 2008; Murray, 2002) and generate a value regime for value creation and absorption; and
- (3) Conflict management.

3.7.1.- Organisational Capacity

Orchestrating is a function that builds a *path of efficiency* since it provides value by not representing a traumatic break with past activities, something which always implies a waste of resources. It is a way of re-combining them in different ways and adding new components to existing ones. It is assumed that orchestrating smoothly paves the way. We consider that successful innovations depend on how Orchestrators manage teams and relationships, thus inviting individuals to consider divergent behaviour, along with a wider range of alternatives in diverse and multi-connected networks (Powell et al., 2005) that drive convergent behaviour, focusing on specific innovations. Leaders listen, scan and monitor the network, and use their status to obtain relevant information, as well as organize people to achieve goals. But they also decide to establish relationships with talented individuals and specialists who may be connected (Dyer & Kale, 2007). We therefore focus on the capacity to manage boundaries: how to select partners, how to achieve structural and strategic fit, how to manage relational dynamics and manage co-operation for co-creation activities while keeping the network united.

This capacity is directly related to commitment by regulating network boundaries through centripetal and centrifugal forces that may lead to lock-ins and lock-outs, respectively (Ghemawat, 1991). Lock-in refers to organisational persistence and core-capabilities. Commitment is not over-embeddedness, resistance to change, and other rigidities (Leonard-Barton, 1995) since commitment always presupposes autonomy. By contrast, lock-out, is the motive behind commitment, when firms decide to disinvest - since, once this decision is made, it may be uneconomical to restart without jeopardising the levels of efficiency (Ghemawat, 1991). This discussion is synthesized in Figure 3.5.



Figure 3.5: Commitments as dynamic capacity to set constraints and boundaries of action

With choices and actions we mean acting while thinking or thinking in action (Schön, 1983). Thinking is based on experience and risk management strategy, highlighting the need to make sense of "turning points" and then mobilising resources to overcome inconsistencies which arise between means and objectives (Thompson, 1967/2003). Thinking underscores these risks in terms of the ability to cope with the complexity inherent in conceptualising "discontinuities" or "transitions" that occur when the *gap* between the requirements associated with contextual conditions and the range of existing knowledge and technology grows (Gilsing et al., 2008; Miller & Floricel, 2007).

Thinking, at individual level, is very much related to resilience, the latter defined as the magnitude of disturbance or sub-optimality that the network can absorb before the network reaches a new equilibrium and structure (Holling, 1973).

By action, we mean experiments and projects that can be defined as how one establishes a new relationship with the future. Experiments take less time than other projects, and allow for some rehearsal and the repetition of some activities that are needed to ensure that all the actors involved in the innovation can feel powerful, and thus avoid the risk of failure (Pisano, 1990; 1991). This is why

experience is of great importance in this approach due to the combination of novelty with some pre-rehearsed components. Experiments combine decisionmaking and action, exploring the real gap between the existing repertoire of strategies, technologies and knowledge and the needed one, but also to how organisations make sense of it through multiple enactments and perspectives (Weik, 1979).

Orchestrating set steps, "big" or "small," reversible or irreversible, may be different for every organisation. It is also very difficult to establish common ground since the steps to build a path depend on: (1) how the different actors engage in projects; (2) how actors adapt to each other by maintaining coherence in terms of objectives, functions, processes, resources and expectations; and (3) how actors engage and in what type of relationships for co-creation.

3.7.2.- Conflict Management

Innovation is a source of conflict since it unfolds despite current structures (Weik & Quinn, 1999), challenging the status quo and the *forces of equilibrium* (Gresov & Drazin, 1997). Gaps and contradictions normally feed conflicts. Conflict can be defined as a situation in which the different parties are faced with perceived incompatible goals and seek to undermine each other's goal-seeking capability (Thomas, 1976). The main sources of conflict in innovation are related to early triggers of change, such as market shifts, changes in technology, the dissolution of some activities or products (due to lifecycles), new waves or perhaps better ideas that start by creating a difference (Day & Shoemaker, 2000).

As already argued, power is the resource to cope with conflicts in organisations. We consider that conflict is a living thing; in other words, living organizations present conflicts. This is, of course, the opposite view to Taylor's approach to organisational theory which strives for "peace and harmony" while repressing conflict and hiding anomic psychic prisons (Kets de Vries & Miller, 1984; Morgan, 1997).

Conflict starts when minorities detect early signals of change. In networks, executives do not necessarily have a unified view of the firm and express (1)

opposite views, (2) that role ambiguity arises when network management is unclear about partners' expectations and (3) that role conflict appears when priorities among actors clash with one another, thus, leading to a confrontation between conflicting demands (Shenkar & Zeira, 1992). Another important source of conflict is that innovation activities unleash different human emotions and passions - from euphoria (at the beginning), to anger and despair (when confronted with reality and complexity in the middle periods), and sadness (at "the end of the tunnel" phase).

Another way of expressing this is with the idea of how to link the new (what has resulted from the innovation) with the old. Integrative mechanisms are important in an entwined process of adopting the new and letting go of the old. Special managerial ability is required to help people to address the end and "let go" (Bowlby, 1985) due to the heavy psychological investments people normally make in innovations (Sutton, 1987). For example, Sutton proposes organisational funerals and even mourning periods. If people feel that everything represents a loss, with no gain at all, resistance may develop and jeopardise a change program, feeding resentment and anger (Bridges, 2003).

These conflicts add a critical source of divergence for leaders of innovation at the subjective level: this is isolation (Burns & Stalker, 1961). An nnovation leaders will tend to compensate for this isolation by *overplaying or underplaying* his or her role (Burns & Stalker, 1961), that is, following the herd of network behaviour (ups and downs) and trying to establish unilateral control while repressing conflicting situations, stopping any attempt to resolve conflicts through the use of dialogue, and preferring instead to keep potential conflicts hidden (Argyris 1986; 1990; 1994; 2004; Argyris & Schön, 1996).

The opposite of isolation is communication, of course. According to Bohm (1965), dialogue is a critical capacity to hold conflict, "Dialogue is a special kind of collaborative conversation, quite distinct from discussion, which is primarily competitive." Dialogue is a free flowing conversation among people, allowing them to discover insights that they could not attain individually. This communicative capacity is very much aligned with a position of power since it refers to the collection, or absorption and dissemination of information

(Mintzberg, 1998; 2002) by the manager or the appointed people or spokespersons picking the right message, the appropriate channel and the right moment to communicate in a formalised context. This role can also be played by self-elected disseminators willing to reveal their special relationship with management to others, that is, using formal and informal channels.

In line with these arguments, we explore how conflict management is related to boundary management and commitments among leaders in order to examine this variable in the success or failure of innovation networks. Orchestrators can be pluralist leaders and promote co-creative and co-operative resolution. For this, they need to be assertive and authentic, as shown in Figure 3.6. Communication mechanisms can be characterized as: (1) assertive - facing and managing the conflict; (2) buffering - assuming an "as if" attitude towards the organization while maintaining their autonomy and continuing with the project; and (3) adapting to – accommodating – conflict (Thomas, 1976).

I believe that power starts with authenticity, a term which means having "full power over" in Greek etymology. This attitude towards collaborative resolution contrasts with some non-transparent styles and hidden agendas which project an image of manipulation, and another source of power: coercion and arbitrariness, away from the needed commitment to innovation.



Figure 3.6: Assertiveness and Co-operation (Adapted from: Thomas, 1976)

Commitments may be faced with several paradoxes, since any change process can require a search for new partners or for some relationships to be dropped along the way as a result of disinvestments (Kash & Rycoft, 2000). As Child et al. (2006) argue, "This may of course offend current partners and threaten current trust, if it is not pursued wisely" (p. 410). Relationships may have their own cycles: they may simply end because partners have reached their objectives or because the project has ended.

3.7.3.- Economic capacity

By "Economic Capacity" we refer to (1) the creation of value regimes to generate and absorb value and levels of performance (see figure 3.4), as well as (2) innovation outcomes. The main assumption here is that governance and coordination generate super additive sum games in which Orchestration is about expanding the pie (generating lock-in) and taking a bigger piece than the rest by controlling critical resources (Danaraj & Pharke, 2006).

By the same token, Orchestration is also about profitable innovation and commercialising innovations effectively (Chesbrough, 2008). While we pay attention to transaction costs since they are a pre-requisite for network formation

and evolution, they do not explain, for example, how *new investment* policies should be managed since the term "cost" is largely problematic when analysing risks and future events (Augier & Teece, 2008). It thus requires a better examination through the use of empirical cases.

Innovation outcome is the extension of Network behaviour as a dynamic capacity. During the change process choices can be made regarding which resources to develop, what should be recombined and which competences should be discarded (Doz, 1996). We propose to examine *resilience* (not breaking the rules of the game) as an outcome of a process of change. Resilience requires examining how much change the network has absorbed to reach a new equilibrium. In addition, resilience may require (1) re-evaluating the initial conditions; and (2) how much the network members (among whom the network started) have adapted to the new situation. Since research is limited in time, one can set nominal boundaries in the process and analyse resilience as the different balance of centripetal and centrifugal forces, and the new sets of symmetries and asymmetries in deep structures such as (1) members of the network co-creating with complementary equals; (2) a network of weak members trying to gain power in order to confront a stronger competitor; (3) bootstrap alliances in which one or more weak partners tries to reach agreements with a powerful partner; (4) the end of a specific network with the aim of exploring other areas in the future; (5) amicable (or otherwise) separations (or lock-outs) (see Bleeke & Ernst, 1995).

3.8.- Conclusion

In this chapter, I have presented Orchestrating Network Behavior for Innovation as a managerial function. This function is intended to set new paths, through a process-based approach that dynamically manages equilibrium, and is presented as a proposition based on commitments which we assume shape boundaries, roles and relational dynamics. The "outcomes" of innovation are precise changes at the structural level, which lead to new behaviours, and which also feed new structures, and so on. I have also discussed the opportunity to consider resilience
as an outcome in order to assess the degree of change a structure can absorb as its capacity for opening new paths.

Orchestration may then be increasingly important in innovation when uncertainty and complexity increase, making trust, equity and conflict-resolution procedures of capital importance in terms of being able to understand how governance structures emerge, evolve and dissolve over time. In the next chapter, I present the research design to test this function in real settings and continue presenting empirical evidence.

Chapter 4: Research Perspective and Design

4.1.- Introduction

This chapter describes how this thesis is organised, its objectives and design, including the paradigmatic approach, scientific methodology, and operative method. I, as the author, choose dynamic capabilities and strategic management that incorporate ideas from Strategic Management, Economics, Sociology, Psychology and Power (Peteraf & Barney, 2003), including explaining, describing and predicting organisational behaviour. The latter requires continuously updating and refining theoretical frames and expanding knowledge (Peteraf et al., 2008) and the multidisciplinary approach (Pettigrew, 1990).

Orchestrating Network Dynamics for Innovation stresses the importance of understanding organisational processes better, and focusing on the "how" (see Peteraf et al., 2008). In order to address these issues, I first discuss the methodological foundations of this research design. I describe *Abduction*, which was proposed and developed by Peirce in greater detail. This idea is coherent with several nuances within Positivism and Realism (Habermas, 2002) and with the pragmatic approach found in Dynamic Capabilities (Teece et al., 1997; Eisenhardt & Martin, 2000). I then describe the operative method that I have applied in order to obtain the data, as well as the data sources. Lastly, I present the narratives resulting from the empirical research, leading to the presentation of the theoretical construct, an object of this thesis.

4.2.- Paradigmatic Approach: Critical Rationality and Positivism

A paradigmatic approach represents the basic assumptions of the researcher's purpose of activity (research interest), the nature of the examined object (ontology), the relationships with the object (epistemology) and the suitable methodology to examine the object of research. Kuhn (1978) defined two paths for paradigmatic formulation: (1) the stable or normalised path and (2) the exceptional path for upheavals of knowledge. The current status quo – or

normalised view – of Organisational Theory recognises the following paradigms: Positivism, chosen in this thesis, as well as other models based on interpretative approaches such as Critical Theory and Constructivism (Tsoukas & Knudsen, 2003; Guba & Lincoln, 1994).

I choose Positivism as the paradigm since it is based on the *principle of demarcation,* focusing on what is real and what is not (Donaldson, 2003; Popper, 1969). Positivism assumes a realistic approach to the exploration of network structures and boundaries (Conway & Steward, 1990). This approach inherently incorporates nominalist qualities such as the reality *presented* by the researcher in the empirical work, which is held and framed by the intentions and objectives of the research question, without which the data would become meaningless.

Second, the researcher's relationship with reality (epistemology) can obtain knowledge about said reality. One of the major flaws in organisational research is the profusion of a-historical and a-contextual studies, with the risk of incorporating collective trends and researchers' sociological biases (Pettigrew, 1990). I assume that the researcher is a contextualised individual *inside* reality (Hayek, 1949/2009; Rand, 1967), capable of gaining the right perspective and seeing research as a learning process and *as an expression of autonomy* (Vygotsky, 1987; Winch, 1958). This idea emphasises the individual's active differentiation from social co-operative systems (Rand, 1967) and real facts, providing an interpretation in order to construct theory. This can be obtained by a set of empirical and *systemic observations,* and it is the subject of further empirical tests to justify theory building.

This approach stresses the importance of dedicating sufficient time to building a perspective, carrying out research and examining long periods of organisational history and its context (Pettigrew, 1990). Context is critical since it may offer different explanations regarding triggers for innovation and change (Mason et al., 1997). This requires: 1) vertical levels of analysis (from macro to micro levels), and 2) horizontal levels of analysis, referring to sequential inter-connectedness, locating change in the past, present and future times (Pettigrew, 1990; Mason et al., 1997). Change is not predefined or expressed by a set of pre-established

trajectories. It requires the careful examination of context, triggers, organisational structural changes, decisions, and the development of technological solutions (Mason et al., 1997; Weik, 1989), the main objectives of this empirical research.

4.3.- Methodology

In terms of methodology, research on strategy is divided between the "what" and the "how" (Poole & Van de Ven, 2000; Peteraf & Barney, 2003). Processes are intrinsically connected with the creation, extension and modification of the network resource base (Pedersen & Larsen, 2006). In the process domain, research is normally based on the question of "how" and leads to qualitative data and the use of longitudinal cross-case analyses (Vidich & Lyman, 1994; Eisenhardt & Martin, 2000; Eisenhardt & Tabrizi, 1996). Power is very much related to capacities as the ability to perform a particular action in terms of the power needed to carry it out. However, until the action is executed, this power remains latent. As a matter of fact, managerial power, structural attributes and dynamic capacities are sources of power to drive change (Teece, 2009). Here, technology and digital platforms can be seen as a reservoir of potential power, a metaphorical fountainhead from which change springs (see Mason et al., 1997) since these technologies have been proven to be instruments of structural change from within, affecting roles, relationships and organisational (and network) boundaries (Lyytinen & Newman, 2008; Pedersen & Larsen, 2006).

In this thesis, I focus on process-like theories at the organisational level, including variables such as dynamism, time, development and outcomes (Petigrew, 1990), thus analysing innovation throughout its entire lifecycle, starting with the antecedents of major changes and examining innovation processes throughout their development (Huber & Van de Ven, 1995; Van de Ven & Hargrave, 2004). This is done to better understand the relationships between power and innovation where digital platforms, as discussed, are conceptualised as instruments or means for change, taking into account their different roles (see Orlikowski & Iacono, 2001).

Taking a process view provides a deeper exploration of innovation processes since: (1) it allows us to examine how processes and outcomes shape each other (Pettigrew, 1990), and (2) it focuses on the "how" and dynamic interrelationships between decisions, the actions taken, and how these influence structural dynamics that address complexities and unsolved previous questions (Eisenhardt & Martin, 2000). This approach is summarised in Table 4.1 below.

Research Design		
Ontological approach	Realism	
Epistemology	The relationship with reality is based on autonomy and distance with the object of study, implying systematic observation, contextual analysis and time.	
Methodology	Methodological Individualism (avoiding fashionable "social constructions of reality" which may bias reality) Pragmatism. Abduction. A learning process, creative, contextualised and reflexive.	
Method	Case Method; objective approaches such as interviews and ethnographic studies	
Theories and	Process-oriented theory	
representations	Narratives	
	Managerial Function	

Table 4.1: Paradigmatic and Methodological Approach

In the following sections, the theoretical approach developed by Peirce is described in greater detail.

4.3.1.- Abduction

Methodology refers to the method theory applied to research, which, in this case, intends to construct a valid hypothesis regarding the innovation function. I use the methodology defined by Peirce as *Abduction*, "a process of argument with whose aid is to obtain [a] true statement of reality" (Peirce, 1992; Habermas, 2002, page 113). Abduction has been used before in research on Orchestration (see Ritala et al., 2009), since this methodology is used when the existing theoretical development is inconsistent and even contradictory (Ritala et al., 2009). Second, according to the literature reviewed, there is no theoretical work specifically dedicated to Orchestration and network structural change. Finally,

this thesis contains a historical empirical focus, recognising that time is needed to interpret organisational change and requires a careful examination of causes and effects (Zollo & Winter, 2002). In this context, the methodology needs to adapt to reality to better cope with organisational dynamics in order to provide a richer theoretical development (Orton & Weik, 1990; Peteraf et al., 2008).

Abduction combines rational deduction from some hypotheses and tests them with empirical data. At the same time, however, observation leads to an inductive process of theory-building. This, in turn, is based on the critical review of existing knowledge (or theories), the observation of reality and on *expanding knowledge boundaries* grounded in an interactive learning process (Peirce, 1992). Reality exists independently of our observation. However, we make sense of it and we can provide meaning by building new knowledge. Therefore, learning processes should be based on controlled feedback between the accumulation of knowledge, theory-building (a synthesis in the managerial function object of this thesis) and empirical research (Eden & Spender, 1998).

4.3.2.- Method

As already mentioned, the key issue regarding structural change and innovation is the question of "how" they occur. Answering this question in terms of dynamic capacities and innovation may provide a richer representation of organisational life due to its process-based approach (Peteraf & Barney, 2003; Huber & Van de Ven, 1995). Given the exploratory nature of this research, I have opted for a research method that combines ethnography and case study research as a tool for demarcation between the context and the phenomenon (Yin, 2003; Have, 2004). I give special emphasis to qualitative data and empirical analysis, a coherent approach with dynamic capacities and innovation process studies (Huber & Van de Ven, 1995; Dyer & Singh, 1995). Case studies are based on systemic observations and grounded on a positivist approach since they constitute empirical research of this thesis focuses on two case studies that are described from the protagonists' perspectives, with their own visions of reality which I have endeavoured to respect (Garfinkel, 1967). Likewise, the case study method is appropriate for this research since I wished to study a contemporary phenomenon within its real-life context, where the boundaries between the phenomenon and the context were not so clearly evident. I also used multiple sources of evidence (e.g., interviews, documents, and observations) in order to eliminate biases (Yin, 2003).

4.3.3.- Units and Levels of Analysis

In order to examine innovation, it is important to define the nominal limits of this research or clarify the researcher's perspective. Assuming autonomy, we carefully need to set "nominal" boundaries to analyse this reality as a complex one (Garnsey & McGlade, 2006). For example, Neo-Classic Economics considers reactive players with zero commitment to economic activities that only build capital, derive its marginal product for an arbitrarily short time period and then resell it. Under such conditions, the choice requires predicting only one thing: price (Garnsey & McGlade, 2006; Augier & Teece, 2009), but not commitments and innovation (figure 4.1). Assuming autonomy in order to better cope with processes and structural dynamics, power dynamics in the preferred choice, more research is needed (Powell & Grodal, 2007; Dhanaraj & Parkhe, 2006; Poole & Van de Ven, 2004). In this regard, I analyse commitments as the main theme and their expressions as centripetal and centrifugal forces to examine structural dynamics, roles and relational dynamics.

I also consider other levels of analysis: inter-organisational and inter-personal relationships (the latter among leaders), examining organisational and individual levels of Orchestration (Ritala *et al.*, 2009).



Figure 4.1: Autonomy, "Coupled Levels" and Units of Analysis (Adapted from: Garnsey & McGlade, 2007)

4.3.4.- Paths and Narratives

I explain and analyse innovation and change through narratives. Innovation and technology-based business stories show how entrepreneurs and executives seek to use technology to improve organisational fit and economic gains and how they bring about something new (Mason et al., 1997). Information and Communication Technology-based narratives provide an account of how major decisions and ICT investments are made, revealing how the operations, structures, functions and power distribution of organisations have changed as a result (Mason et al., 1997). This recognises that narratives also help us to understand meta-phenomena, in other words, that organisational change may be a process and an outcome of the innovation process at the same time (Huber & Van de Ven, 1995).

4.4.- Case Selection and Presentation

I have selected cases that cover extreme situations to highlight differences and key variables (Yin, 2003; Pettigrew, 1990; Eisenhardt, 1989:1). I wanted to compare a successful entrepreneurial and innovative focal firm, Multiasistencia, and its evolution from 1983 to 2007 with the same orchestrating phenomenon in a multinational. In the first of these cases, the IBM Barcelona Lab successfully developed innovative software solutions from 1984 to 2004. The cases present important similarities. Their innovations focused on the development of technological solutions such as digital platforms along with network change. The two cases are critical and unusual, and they both include disruptive innovations and major changes in contextual demands and market shifts. In addition, the management of efficiencies such as results, boundaries and domains of action are critical aspects to examine in order to understand their evolution.

In the IBM case, the Lab successfully orchestrated its network from 1994 to 2004. In the cases I focus on digital platforms (Gawer & Cusumano, 2008) as outcomes of innovative efforts, although the main unit of analysis of this work is structural change.

4.4.1. Multiasistencia

During the study period, Grupo Multiasistencia was a leading company dedicated to providing complex services to banks and insurers through its original home insurance claims and repairs services. During the study, the company maintained operations in Spain, France, the UK and Portugal, serving approximately 9,000,000 end customers. It offered its services by outsourcing contracts to 100 banks and it managed a network of approximately 11,000 trade professionals. The firm's evolution spans from its foundation in 1983, to 2009. This Orchestrated Network is European-wide presenting end customers and corporate clients through its Control Centre located on the outskirts of Madrid. From this Control Centre it managed the main process it called its "Comprehensive Claim Management Service" (CCMS). The Centre employed 375 Customer Service Representatives (CSR) who received requests for home repairs by phone, e-mail, and the Internet.

They deployed and supervised jobs to a network of trade professionals who carried out the repairs. Since 2004 on, Multiasistenica has managed a ubiquitous Business Network Operating System (BNOS) based on different technologies such as the Call Centre, the Internet, Web Services, and Mobile Systems. The BNOS incorporates corporate customer demands as turnkey services. All services are audited through Service Level Agreements (SLA) with all the agents in the network. Multiasistencia has boosted its value and innovation by collaborating with its corporate clients in developing new services. Figure 4.2 represents the relationship model and the business network actors with respect to the firm at the end of the 90s when Multiasistencia appeared as the focal firm within this business network. In the 90s, Multiasistencia was very successful in managing groups and attracting corporate clients such as banks and insurers.



Figure 4.2: Grupo Multiasistencia

4.4.2. IBM Barcelona Lab

The IBM Barcelona Lab was created in 1984 when a group of engineers developed a turn-key application to integrate the PC for one of the most important savings banks in Europe. This project was the basis for the LAN-DP (LAN-Distributed Platform) software product. In 2005, there were more than 1,500,000 licenses for this product used by 1,500 banks around the world. The second phase spans from 1994 to 2004. In 1994, IBM faced one the worst crises in its history. Given this situation, the company needed the LAN-DP application to be transferred to another of its labs in order to develop economies of scale. The Barcelona Lab had to re-invent itself, lowering its centre of gravity and defining itself as open to the customer, thus Orchestrating a Network of banks, other Labs and Independent software vendors (ISV) worldwide pioneering *hybrid programming,* which allowed it to design a new software product in collaboration with its customers: the Websphere Business Component Composer (WBCC). In Figure 4.3 below, I outline the project-based relationship model, the business network actors, and the Lab's central position as the focal Firm.



Figure 4.3: The IBM Barcelona Lab as a Network Organisation

4.5.- Data Collection

The main objective of this thesis' empirical work is centred around: (1) structuring action over time; (2) a contextual examination of the diverse relations and dependencies in each context and at different levels of analysis; (3) a comparison between the case studies, with data collection taking place from 2004 to 2007; and (4) a pluralist approach, emphasising the fact that the main sources consisted of interviews, phone conversations, focus groups, observation and secondary sources (Pettigrew, 1990). The data were mainly gathered through interviews with executives and employees at workshops conducted in June 2004, February 2005, and March 2006. The average duration of each interview was 90 - 180 minutes. Managers related their stories of how the change process evolved in their respective firms.

4.5.1.- Interviews

I divided the interviews into three parts: (1) the first was about the company, the type of business, the context and the motivations for change (Pettigrew, 1990); (2) the second was about the interviewee's subjective view regarding the change

process (their stories) and "how" they did it, emphasising the decision-making process with a focus on the activities, the value for the actors, the innovation process and how relationships were established, changed or brought to an end; and (3) the third part was related to strategic issues such as results, organisational changes and resource mobilisation. The interviews were semiconstructed and concentrated on managerial decision-making. Interviews included opposing questions and listening to different perspectives at an individual level, as well as the capacity to lead focus groups to contrast conflicting information.

In addition to the interviews, there were countless telephone calls, conversations, and comments throughout the process, which also allowed me to collect informal data. For example, I was invited to a customer event with two of the companies where I had the chance to share meals, receptions and coffee breaks and spend considerable time with the companies in order to gain their confidence and gather good ethnographic material. I took written notes during these meetings and recorded them by date and subject.

Strategic Profile	Multiasistencia	IBM Barcelona Lab
Chairman	3	N/A
CEO	3	IBM Spain 1
CIO	6	6
Lab Manager	N/A	6
Business Development Manager	2	6
Country Managers	3	2
Production Manager (COO)	2	3
Network Suppliers	2	2
Network Corporate Clients	6	

 Table 4.2: Description of Case Data (Number of Interviews)

As Table 4.2 shows, interviewees were selected because of their leading positions in the respective organisations and in the innovation process under analysis. They included initiators of change, those affected by changes, and others external to the focal firm. Another source was focus groups with the different managerial levels. Interviews were prepared and conducted over three years, from 2004 to 2006, allowing them to validate the findings from the case data. Interviews were recorded on paper notes, audio tapes, and 6 hours of videotapes (resulting in two edited DVDs), along with teaching cases for Multiasistencia and the IBM Barcelona Lab.

4.5.2.- Other Sources

The teaching cases also allowed me to invite executives to my classes. This method provided good additional insights from the interaction between these managers and participants. I asked for permission to record these classes (sometimes via an assistant taking notes and others by video recording) in order to be able to analyse and compare these sources of information, and triangulate them with others, both in time and with other perspectives.

One of these cases, the "IBM Barcelona Lab: Orchestrating Banking Innovation" (<u>www.ecch.com</u>) was named *the best case outside the USA and Canada* by the North American Case Research Association (NACRA) in October 2006. The presentation of this case and "Multiasistencia on the Internet (A), (B) and (C)" in 2005 to the same conference allowed me to discuss and check the structure of the cases and their validity with the academic community.

The research process took four years for each case on average. Data collection and analysis occurred iteratively, thus guiding subsequent data gathering (see Table 4.3). In presenting data, I emphasise people's voices. As already stated, other sources included direct observation and secondary sources such as company reports or press releases. With the understanding that every research project has its limitations, I am confident that I have an accurate enough impression of what happened in these firms to understand the process' object of study.

Multiasistencia	IBM Barcelona Lab
 Field notes taken during visits to the company Press notes and clippings from 1998 to 2004 Multiasistencia Magazine for their network of professionals and network of clients (2004-2008) Internal company documents: internal communications, reports, process descriptions, and documentation for software developed Documentation regarding the company's financial situation Personal interviews: (1) with company executives: the President, Group CEO, General Managers in Spain, France, and UK, the International Expansion Director, the CIO, the Network Manager, and the COO; (2) With executives in two of Multiasistencia's most important corporate clients: SegurCaixa, belonging to the 'la Caixa' Group, and BBVA Seguros; (3) With managers of the franchised repair companies Focus groups with the Board of Directors in 2005, 2006 and 2007 Attending client meetings in 2006. 	 Field notes taken during various visits to the company Documentation regarding innovations introduced by the Lab from 1984 to 2004 Internal company documents: internal communications, reports, process descriptions, and documentation for software developed Documentation regarding the company's financial situation (confidential and cannot be released) Personal interviews: (1) with Lab executives: Lab General Manager, the Director of Business Development, the Director of Sales for Banking and (2) With an executive from the "la Caixa" Group Focus groups: one per year from 2003 to 2007

 Table 4.3: Data Sources for Empirical Research

4.6.- Conclusion

In this chapter, I have described the boundaries of the empirical research. The chapter includes objectives and design, including the paradigmatic approach, scientific methodology and operative method. I chose dynamic capabilities and strategic management to focus on the "hows". In order to address these issues, I have described *Abduction*, coherent with Dynamic Capabilities (Teece et al., 1997; Eisenhardt & Martin, 2000) and have endeavored to expand knowledge boundaries to examine the autonomy dynamics or degrees of freedom needed to drive changes affecting network structures. I have also described the operative method that I have applied emphasizing narratives and longitudinal studies. In the next chapter, I analyze the cases presented to examine changes in boundaries, roles and relationships in order to contrast with the propositions described in Chapter 3, where I defined Orchestrating Network Behavior for Innovation as a managerial function.

Chapter 5: Orchestrating and Network Structural Change

5.1. Introduction

This chapter applies the research method to Network Behaviour with two empirical cases, using power as a unit of analysis and centripetal and centrifugal forces to examine roles, relationship dynamics and network boundaries. In order to explore these questions at the empirical level, the strategy followed in this chapter is as follows: first, the present situation is defined and the Multiasistencia and IBM Barcelona Lab cases are analysed, with a focus on their evolving structures (roles, network co-ordination models, integration processes and type of relationships) and their stories as processes. However, to make this simpler, several stages are defined that act as references in the transformational processes which were used for over 20 years in Multiasistencia and the IBM Barcelona Lab.

5.2. Case Presentation

This section presents Multiasistencia's and IBM Barcelona Lab's evolution as a succession of local maximums in an ongoing process of experimentation regarding their respective business approaches, market development, network structures and digital platforms delivered. By comparing these two cases with their dynamic structural change as a path of efficiency for the proposed Orchestrating Network Behavior for Innovation function, innovation process is empirically analyzed, observing Orchestration from the focal firm's central

position. If we observe the empirical phenomena from another perspective and try to imagine the role of Multiasistencia when orchestrating, for example, versus that of IBM, we see how it develops its full potential.

Thus, this chapter presents the analysis of two narratives: in the first, we see the generation and evolution of a network from its origins, the Multiasistencia case, representing a natural way to create firms; in the second, we see a process of network generation and evolution within a global firm that can be framed as a Differentiated Network.

5.3. Multiasistencia

This section describes and analyses Multiasistencia, a focal firm within a network that provides different services for banks and insurers. This case was selected to study different stages within the firm in which to examine the equilibrium, innovation and change throughout its 25-year history (from 1984 to 2009). This period even encompasses the disruptive change from 2000-2003 when Multiasistencia successfully implemented a digital platform. Its evolution reveals optimal and sub-optimal stages affecting the various network relationships and the roles of different. For analytical purposes, we propose a dynamic, five-stage process spanning: (1) the network's initial situation (1984-1995); (2) the boom and expansion of the network (1995-2000); (3) the change in the network (2000-2004); (4) network behaviour promoting innovation as an equilibrium to innovate with information through a digital platform (2004-2008); and (5) the focal firm's introduction of some changes in network governance (from 2008 onwards).

5.3.1-Network Creation (1984-1995)

This phase is defined by an entrepreneur who started Multiasistencia after identifying a structural gap between small customers (hotels and restaurants) and repair services. According to the founder, "What we proposed was to establish criteria for these repairs, guaranteeing a set price and response, consistent service and quality standards" This idea was expressed in the company's slogan and promise in the 90s, "Yes, we can". It was a promise of trust that filled the structural gap and generated a new type of business and a new "repairs market."

Multiasistencia provided its repair service through its International Control and Co-ordination Centre. The Customer Service Representatives (CSRs) working there received the initial call from end users looking for the final solution, in many cases home repairs. Once the claim was accepted, the CSR deployed jobs to a network of self-employed trade professionals, co-ordinated by the focal firm.

5.3.2.- The Network Boom (1995-2000)

Multiasistencia's boom generated increasing returns that acted as a centripetal force, attracting new actors to the network and expanding boundaries. These included banks which wanted: (1) fast access to the new insurance market where time-to-market was crucial and working with Multiasistencia offered this possibility; (2) lowered transaction costs to take advantage of increasing economies of scale; (3) access gained to a very specialized service; and (4) a focus on their core activities. This allowed banks to generate more value by

cross-selling insurance to their home mortgage clients or credit card holders. Trade professionals, by contrast, had access to larger markets, ensuring a constant flow of jobs. Meanwhile, end customers appreciated the 24/7 availability, time guarantees (3 hours for emergencies and 48 hours for the rest) and warranties (3 months). By the end of 2000, Multiasistencia reached 9 million end customers in Europe through 100 major corporations such as banks and insurers that outsourced this service. From 1995 to 2000 Multiasistencia grew from €40 m in turnover to €90 m. Its growth generated important economies of scale, increasing returns and network externalities that increased lock-in.

However, there was a latent and increasingly important asymmetry with trade professionals. In 2000, the network encompassed approximately 11,000 trade professionals. This collective was only 20%-dependent upon delivering the service. There was a "culture of freedom" which hid latent opportunistic relationships during the first phase, serving as a centripetal force so long as the business grew. Nonetheless, as the case shows, relationships based on opportunism can be very cohesive. The attractiveness of the business and the focal firm assumed full responsibility for co-ordination (pushing the actors in the network). There were no clear rules of engagement or commitment and, therefore, no clear boundary setting.

In 2000, the Control Centre needed to make 5.5 calls on average to find an available trade professional. As such, the firm's economic boom also masked an important escalation of increasing co-ordination and transaction costs which drained productivity and finally led to a sub-optimal situation, making it more risky and costly to be a part of the network. For example, productivity dropped

from 1,900 claims/customer service representatives a year in 1995 to 1,200 in 2000. While the Control Centre grew from 100 CSRs in 1995, to 450 in 2000, their activity focused increasingly on searching for available trade professionals and responding to complaints from end customers and banks, meaning that they dedicated less time to developing more business.

In parallel, Multiasistencia focused on innovation and new service development. For example, it succeeded in: (1) reducing the average claim cycle completion from 30 to 5 days; (2) reducing the process for a bank customer to change accounts from 2 weeks to 2 hours, bridging banks with utilities and invoicing providers through an automatic branch office billing service; and (3) linking banks/insurers with trade professionals to carry out repairs through a comprehensive claim management service. One manager explained,

"Multiasistencia was very creative, but faced what can be described as an invisible glass ceiling: we did not really understand how to control a whole lot of fragmented operations in a manner that was more professional, more automatic."

We can even see Multiasistencia as a focal firm in the *centre* of a network, but one which lacked power since the relational dynamics made Multiasistencia *much more dependent on the surrounding relationships* based on opportunism, thus leading to an increasingly costly structure which jeopardised the business. This was "the invisible glass ceiling."

5.3.3.- Network Change (2000-2004)

New management joined the focal firm in 2000 and decided to change the rules of the game and define a new set of commitments and boundaries with the trade professionals and banks. The management team decided to commit to trade professionals by asking them to do more by themselves while coaching them in the process of: (1) changing their role from that of self-employed workers to managers of small firms of 10-15 professionals; (2) managing a much larger number of repairs; (3) co-investing in software modules (web services) to manage these new businesses (financials and job assignments) which were developed initially by the focal firm; and (4) accepting a new co-ordination schema to a) pull jobs from the digital platforms, as well as digital billing and Service Level Agreements (SLA), and b) leaving the sales process exclusively to the focal firm.

On the one hand, this new schema meant that these newly created repair firms had to implicitly accept sharing co-ordination costs; and, on the other, they had to accept greater dependency on Multiasistencia. However, this new power equilibrium was based on mutual commitment and trust. As one manager added, "The communication policy of sharing our objectives with them was, therefore, an absolutely key factor. There has been a relationship of sharing the objectives and of growing together." However, not all trade professionals accepted these new rules of the game, and from 2000 to 2003, some of the trade professionals left the professional network, thus ending their relationship with the focal firm. The focal firm's management team also changed the relational dynamics with banks by balancing the centrifugal force of conflict due to dissatisfaction. Multiasistencia's management wagered on "openness" to define a new set commitments. It established agreements with its client banks to co-create the internet solution, the Business Network Operating System (BNOS), among all the network actors. As one manager related,

"We had a model of relations with these trade professionals which was not very well integrated with them. We offered work and they took it, if they wanted to, and if they didn't want to, they didn't take it. It was much better to reach a level of greater commitment, that is, we undertook to do a series of things, to give them a volume of work, and they undertook to accept it and to carry it out within certain dynamics."

The BNOS was designed to manage claims and process repairs in a fully automated manner, linking banks and trade professionals in real time, and it was governed by 100 variables. Multiasistencia's management asked for a new commitment from banks through sharing the scalability problem, presenting the solution of leveraging digital platforms and asking them to co-create the new digital platform to run network processes. This led to a peer-to-peer type of relationship, creating autonomy on both sides and allowing the co-creation of software modules which combined existing co-specific software applications to build the BNOS. According to one manager, "80% of processes developed were through co-creative processes." However, in a few cases, Multiasistencia had to create lock-in policies by increasing greater dependency and raising exit costs to keep some corporate customers from leaving the network and consequently losing business revenues. As one manager indicated,

"We deployed the repair work to the trade professional; previously all these activities were by phone, but by replacing all the elements of the relations, except for the incoming call from the customer requesting the service, with information flows, allowed us to be much more agile, to reduce times, to be much more flexible and to devote the human time to managing the exceptions and the complicated processes,, instead of performing manual tasks which can be carried out more efficiently by a machine".

Their action can be interpreted as exerting autonomy by fostering a *centrifugal force* to change relationship dynamics: the focal firm asked for greater selfcontrol and commitment among network actors, fought against opportunism, and created peer-to-peer relationships. In addition, Multiasistencia enhanced the CSRs' empowerment and autonomy by transforming their role from that of "information chasers" to "process owners." They began by assessing the repair according to the firm's schema programmed on the BNOS. The BNOS was intentionally designed as a boundary management system, establishing a framework or pattern of expected behaviour acting as a deterrent (boundary guardian). According to one executive, "We have substituted all human communicative elements in the repair service, except incoming calls, with machines. We have developed a sophisticated process supported by ICTs, allowing people to manage exceptions." Productivity at the Control Centre increased by 25% a year from 2000 to 2003, contrasting with the decline of previous years(see Figure 5.1).



Figure 5.1: Productivity Evolution at the Control Centre Source: Author based on estimations (1996 to 1999) and data provided by Multiasistencia (2000-2004)

Multiasistencia changed its role to that of *reference* in the Network. One manager commented, "The BNOS had been conceived as a *framework of reference* for network behaviour, freeing up resources to focus on developing new business and innovation." According to another executive, "After the change process, the business was autonomous: we did not need to push it anymore." The change process led to a new equilibrium, giving the network a totally different structure.

5.3.4.- Network Behaviour for Innovation (2004-2008)

At this stage the focal firm managed network boundaries by setting commitments which balanced the centrifugal force of autonomy with centripetal forces of increasing returns and lock-in. Multiasistencia's role was to serve as the reference in a network of innovation, opening space for growth by leveraging the digital platform as a space for co-creation. It also developed trust as a deterrent, controlling network boundaries through SLAs and through the BNOS which automated and kept track of all the processes but also acted as a space for communication and co-creation.

Multiasistencia changed its role from that of "the innovator" which saw the network as an instrument to diffuse innovation, to that of "the reference" (a trustee to identify and share knowledge). It began to look at the network as "the locus of innovation" and became a facilitator to co-create innovation by orchestrating the interaction between the different sources and voices, understanding proposals, composing them and combining them as new services. One trade professional commented,

"In this process, we learned to adapt to what they [Multiasistencia] asked of us and, in a way, well, we also learned to trust them, to be more transparent and share information to be more agile and do better business."

One bank executive added, "Multiasistencia is very innovative and always proposed new ways to develop new business".

First, CSR interaction with all the network actors served as a new source for solutions. New ideas appeared regarding conflict resolution due to gaps in service provision and the problem-solving process. Moreover, the digital platform provided a history of contacts and data analysis which fed predictive models to mobilize the right resources at the right time. Examples of this were better workload distribution in the Control Centre, process improvements, and increased availability by managing a network of three other Contact Centres belonging to insurance companies in Spain. These insurance companies acted as back-up centres to ensure access during peak periods in telephone traffic or during telecommunication network failures.

Second, corporate customers (banks and insurers) and trade professionals were also important sources of innovation. New software components (web services) were developed by co-creating new services. The digital platform was a space for service innovation and shared investments and network resources, although Multiasistencia continued to be the formal owner of the technology. Examples of this co-creation were, first, digital signatures and electronic invoicing, "desk-top audits" over the internet, allowing loss adjusters to assess claims over the BNOS, and premium services to inform customers through SMS messaging. Finally, Multiasistencia also continued innovating by replicating this model in other markets such as healthcare and public services (see Table 5.1).

Source of Innovation	BNOS Role	Benefits		
Continuous service improvement				
	Deterrent Role	Increase the Efficiency of Existing Operations		
Network Interactions	Process automation	CSR managing exceptions: Proactive quality		
	Control variables and Service Level Agreements (SLAs)	Increase in productivity by 25% a year (2000-2003)		
Customer Service	Automatic Boundary Management and Process Automation	Digital invoices: 40% cost reductions through the reduction of		
Representatives (CSRs) with		paper work		
end customers, corporate	Predictive Models	Predict New Paths for Efficiency		
customers and trade	Data Analysis	Peak Management and Service availability		
professionals at the Control		Service Enhancements		
Center with the BNOS				
	Co-Create New Services			
	Space for Co-creation	New Sources of Income		
Peer Interaction	Generation of Economies of Scope			
Software Co-creation				
	Including digital photos	Desk-top audit: New area of revenue and cost reduction (for		
	Inter-connectivity with insurance systems and trade	network actors), representing 60% of the process		
	professional systems			
	Premium services and generating SMS messaging	SMS messaging, improving communications with end customers		
		30% increase in customer satisfaction		
New Markets				
	Fostering Positive Returns	Replication of the Model in Other Areas		
Focal Firm Management	Generation of Economies of Scale	Increased business scale throughout (1) the front office; (2)		
		inter-organisational processes and (3) network structuring		

Table 5.1: Service Innovation in Multiasistencia's Network

Orchestrating the Network at this stage implied managing a new equilibrium. We can identify the following centripetal forces: trust, higher business value, and the generation of economies of scale and scope. Centrifugal forces at this point might have been misfits at the strategic or resource levels, representing a new cost base for firms which re-defined the logic of transactional costs. In some cases autonomy has also worked as a centrifugal force in terms of the search for new opportunities.

The result was the increasing dependency on Multiasistencia due to growing specialisation and different roles: banks focused on their core business and trade professionals only on repairs, leaving business management to the focal firm. The digital platform was also an important instrument to generate system lock-in and served to generate trust as a deterrent, keeping track of SLAs.

5.3.5.- 2008 to the Present: Towards a Dominated Network

As of 2008, Multiasistencia started to face competition in the very market it had created. Some new entrants, including insurance companies, began to focus more on quality assurance rather than on innovation. Multiasistencia decided to incorporate some trade professional franchises and their staff as special units to ensure top quality for some key corporate customers, developing a hierarchical relationship with them. In fact, Transaction Cost Theory suggests that firms try to control specific scarce resources and that they tend to insource them (see Williamson, 1994). This approach seems insufficient, according to the data, to fully interpret managerial decisions in fields such as this where more research is needed.

5.4 Case Study: IBM Barcelona Lab

This section describes and analyses the IBM Barcelona Lab as a focal firm leading software development for middleware integration between the front office and the mainframe. This case was selected to study different stages within the firm to examine the equilibrium, innovation and change throughout its 25-year history (from 1984 to 2009). This period even encompasses a disruptive change from 2000-2003 when IBM Barcelona Lab opened up software development to its customers as "an open lab." Its evolution reveals optimal and sub-optimal stages affecting the various network relationships and the roles of different actors. For analytical purposes, a dynamic, four-stage process is proposed that spans: (1) the network's initial situation (1984-1994); (2) change in the network (1994-1996); (3) network behaviour promoting innovation for the development and deployment of a digital platform (1996-2004); and (4) when the IBM Barcelona Lab became a Competence Centre (from 2004 onwards).

5.4.1. Network Initial Conditions: The PC in the Bank Branch

IBM Barcelona Lab started its days with one relationship with a bank for a specific project. In 1981, IBM launched its personal computer (PC). By 1984, the new product, named PC-AT, was the central computer used for application development in the special projects that the banking team was working on in Barcelona. However, the "la Caixa" savings bank in Barcelona, and other institutions in Spain, insisted on IBM in Spain leading the change, that is, introducing PCs in their branch offices to reduce IT costs.

IBM executives proposed partnering and developing an *investment project* between IBM in Barcelona and "la Caixa", informally planting the seed for open innovation through this relationship which changed from a provider/customer relationship to one in which the actors co-created something new together. The project was a good example of autonomous action, since it was launched without following the formal procedures at IBM regarding software development. The plan involved a pilot project that would allow a prototype to be designed locally, based entirely on "la Caixa" and IBM resources in Barcelona. This project was very successful and later became a standard solution for many financial institutions. It was called *Financial Branch System Services (FBSS)*.

As a result, IBM decided to formalize this innovative venture, this change of role, by converting it into an R&D laboratory for banking applications in Spain, known to its staff as the Barcelona Lab or "the Lab." In 1993, seven years after it had come into being, the Lab had consolidated its position and had more than seventy professionals on staff. In 1994, LAN-DP had 500,000 licenses world-wide. Thus, the product was entering its mature phase.

5.4.2. Network Change (1994-1996)

In 1993, IBM announced the most negative results in its entire history, 8 billion Dollars, and its market capitalization plummeted from a high of \$105 billion to \$32 billion. Given this situation, the company's corporate management, presided by J. Akers, proposed that IBM be split into seven separate companies to save it from bankruptcy. However, in 1993 when Lou Gerstner came in as CEO, his primary objective was to keep the company together as a single unit. Only twelve months later, in 1994, and in an unprecedented turnaround, the company announced profits of 3 billion Dollars.² The main decision was to shift its orientation from products to service. A result of this new orientation was IBM's decision to create IBM Global Services (IGS) as the firm's new Consultancy Division. After the turmoil in 1993, IBM began to offer its customers integral solutions.

However, there was a latent and important asymmetry with the rest of IBM, as in any other hierarchy. The attractiveness of the business around LAN-DP determined that IBM Barcelona Lab assumed full responsibility for software production where there were no clear rules of commitment with the rest of Software application groups. However, due to market shifts and new technological waves, it was decided to re-structure the whole organization. In June 1994, IBM underwent heavy cost optimisation that also had repercussions for its laboratories policy, particularly the smaller European units in Vienna, Rome, and Barcelona. This new policy led to the decision to discontinue Barcelona Lab's operations. As a consequence, the Leader who had run the Lab since its foundation resigned.

In the uncertain times of 1993 and beyond, IGS in Spain offered many of the Barcelona Lab staff a natural outlet for their professional careers. There was a constant movement of people switching from the Lab to IGS. It was more appealing to work on particular projects integrating this new technology than to wait for the unknown. By the end of 1996, the Lab had less than twenty staff members, and many had opted to join the new IGS consultancy unit.

² Gerstner, L. (2002). *Who says elephants can't dance?*. New York: Harper Collins.

5.4.3. Network Behaviour for Innovation (1996-2004)

A key aspect was boosting collaboration with the Sales Banking Organisation in Spain. When the risk that this small team would be definitively incorporated into the Consultancy Organisation was at its height, the Commercial Banking Organisation and IGS identified several projects for them in various financial institutions where the disruptive technology which became known as the internet was appearing. The new management team proposed a centripetal force: (1) developing a Lab that was open to its customers to carry out innovations based on internet technologies and (2) exerting autonomy by rethinking software development through engaging in direct relationships with customers, that is, banks aligned with new lab strategy. According to one executive,

"We focused on finding banks that worked globally and had to face global challenges in integrating their business processes and their technological assets. The Lab's origin lay in being close to customers; therefore, I decided to take advantage of this capability by opening the Lab up to customers."

IBM Barcelona Lab also changed its role(as Multiasistencia did), from being the "the innovator" and perceiving the network (in this case IBM and their customers) as an instrument to diffuse innovation to *understanding the network as the locus of innovation,* and becoming "a facilitator" to co-create innovation by orchestrating the interaction of the different sources and voices. As one Lab member related, the Lab was no longer like an Ivory tower, since

they had moved its gravity centre towards the customer. On the other hand, as argued, a new trigger appeared in the context: Internet technologies were rapidly evolving, and thus it was easy to find key customers willing to innovate with this type of technology, that is, who wanted to take advantage of the Lab's know-how. The merit was based on its early detection, and very rapidly a new set of commitments and a new method of organisation. According to one executive, it was a,

"A matter of capturing the existing innovation needs of each of the clients, making them reality and co-creating a solution with the client. As a result, we had key people in key projects."

The IBM Barcelona Lab exerted autonomy, staffing "key people in key projects" from IBM and from banks, and dealing with the leaders like technical architects or sales people. In the Lab, it was quite the norm for a professional who was working in development to then move to the project group at the end of the development phase, later going to the client to implement or start up a project based on the product that he or she might have developed personally. The Lab not only exerted autonomy, it generated a new capacity to attract even IT banking staff. As one manager commented,

"And at some point I *took a gamble*, because we needed project directors from the banks in the laboratory. And without a doubt, that has been one of the wagers that has given the best results."

The Lab also implemented a new way of developing software that shaped its very structure. In other words, the related tasks shaped ties and structures

while these relationships also forged trust and commitment which, in turn, shaped new tasks. The Lab was shaped like a daisy, where project teams rotated with software development teams in different projects, analysing, diagnosing and building software solutions. One of the Lab members stated,

"This results in plans being changed. People have to be dynamic, flexible, and know how to adapt to these changes. I would say that this is a little difficult at first; basically, professionals prefer to know very clearly what is expected of them, exactly when and how. And as soon as they get used to that dynamism, to that flexibility, they take it as a routine part of their work."

The capacity to rotate from software development to project management even to business development was considered to be a part of the Lab's tasks, thus "avoiding repetitive or routine activities," according to one manager. Indeed, experimenting entered a flow of ongoing ideas and invention, a combination of accents, changing harmonic patterns that would inter-weave the structure of relationships and networks. As a manager indicated,

"So, we put *key people in key projects*. In graphic terms, the organisation was like a flower. Its centre was formed by the core competences: Product functional development and innovation. The petals of the flower were formed by specific customer projects."

The purpose of the first line (the project teams) was to define solutions to meet customers' needs. In this context, it was the intermediary's job to define which
part of the project could be recycled. The Lab developed new roles in the network.

As discussed previously, in the IBM Barcelona Lab case, the Lab's boundaries, roles and relationships also changed. Until 1993, the Lab's knowledge base was focused on software development. By engaging in project management, it developed its business development capacities – as executives in the Lab called it. Therefore, the Lab became less dependent on the willingness of sales departments to sell their solutions. This is an important issue in multinationals such as IBM, where product and sales catalogues are vast and product divisions dedicate part of their resources to sell products to sales forces. In this case, the IBM Barcelona Lab's Business Development team did not compete, but was much closer in order to control the sales process in IBM.

The IBM Barcelona Lab deployed Internet applications to integrate new channels and, as a result, developed its Websphere Business Component Composer (WBCC), a modular software architecture to integrate multiple contact channels based on re-using existing components and creating new ones. In addition, these technologies became richer and increasingly complex, adding more features and easier access to transactional systems. This allowed for some of their other services to be increasingly available, including requests for accounts, transfers, and managing payments.

According to one executive, "WBCC was a "component composer" that gave *more autonomy to customers*, letting them build new applications, re-use and integrate them with existing applications" affecting multi-channel strategies. As one manager declared, "In essence, software projects were thought of as the development of re-usable components"- Additionally, project teams

synthesised knowledge into interconnecting software modules which were the foundation for the modular architecture that became the WBCC launched in 1998.

IBM Barcelona Lab orchestrated different sources of innovation (Table 5.2). Banks were important sources of that innovation. New software components (web services) were developed, co-created and combined with existing IBM technology. WBCC was a space for innovation and shared investments and network resources although IBM continued to be the technology's formal owner. Finally, IBM, as a corporation, took advantage of the WBCC technology to continue innovating by escalating this model to other industries, as well as re-framing its relationships with customers.



Source of Innovation	WBCC Role	Benefits			
Continuous service improvement					
	Deterrent Role	Increase the Efficiency			
Network Interactions					
	Service Level Agreements (SLAs) for software delivery	Time to Market: from 2 years to 6 months			
IBM Lab staff interaction	Standards	Customer Satisfaction			
with Banks, other IBM Labs,					
sales force and Independent					
Software Vendors (ISV)					
	Co-Create New Services				
	Space for Co-creation	New Sources of Income			
Peer Interaction	Generation of Economies of Scope				
Software Co-creation					
	Inter-connectivity between multi-channel front office with	New integration projects			
	Mainframe	Replication of cumulated knowledge in new customers			
	Multi-disciplinary teams (banks and IBM Lab staff)	Transferring knowledge to other IBM Labs			
New Markets					
	Fostering Positive Returns Replication of the Model in Other Areas				
Focal Firm Management	Generation of Economies of Scale	Value absorptive capacity by IBM as Corporation			
of the differentiated		IBM as diffusion network (e.g., 1.500 Banks use LAN-DP)			
network					

Table 5.2: Innovation in IBM Barcelona Lab's Network

5.4.4. Becoming a Competence Centre

At some point in 2004, IBM decided that, to increase productivity, the basic programming could be done from its laboratory in China. This would result in greater programming productivity and cost reduction due to off-shoring policies. The Vice-President of the Banking Industry in EMEA summed up the situation in 2004 as follows,

"IBM Barcelona Lab was essential for our competitive strategy in the Retail Banking sector not only in Spain: It was a key competitive advantage in our European and global operations. The Barcelona Lab demonstrated high organisational flexibility, and its hybrid programming techniques were very useful for the incorporation of new applications, reducing time-to-market and making it easier to share knowledge and enhancing collaboration between the Lab, the sales organisation, and the clients."

IBM decided to capitalise on the extensive know-how of its workers, although not as much in terms of technology as in terms of solutions for the financial system, transforming the software laboratory into a skills or competence centre. As one manager argued,

"The laboratory and skills centre motto has always been that we should "reinvent ourselves." But I'm also convinced that within four or five years we're going to have to dream up the next step. We are always going to be there next to our clients, because the other part of the motto is "this reinvention will only be possible if we are close to our clients' projects", and not like the shut-away or isolated laboratory concept, which is very out-dated. That would never work, we will carry on reinventing ourselves"

5.5.- Propositions

This chapter has contributed to the discussion on dynamic capabilities and power by defining and testing *Orchestrating Network Structural Dynamics* empirically. Grounded on the foregoing analysis, this chapter presents the propositions which are described in the following sections.

5.5.1.- Orchestrating Networks: The Dual Nature of Power

Proposition 1: Orchestrating Network dynamics for innovation balances autonomy and cohesion through dynamic commitments as a power expression of managerial action.

With this proposition we argue that Orchestration, by nature, exerts power in two ways. First, it is about setting direction and opening new spaces. In the cases studied, this occurs by exerting autonomy to network actors. More autonomous actors may make new choices to search for complementarities which the orchestrator can then mobilize and make the network the "locus" of innovation. In the Multiasistencia case, it works as a change process. In the IBM Barcelona Lab case, it is the result of "open[ing] the Lab to customers"- Both cases show networks as multi-connected and multi-disciplinary spaces of new relational dynamics which can combine and re-combine new sources for innovation such as: (1) autonomous staff focused on *problem-solving*, stemming from the interaction with other network actors and with information systems such as the digital platforms; (2) the *co-creation* and development of a digital platform based on peer-to-peer dynamics; and (3) the re-design of network structures through the capacity to engage in *new communication* schemas.

The second source of power is setting commitments. Innovation requires trust since actors need to expose themselves and unfold new sources, propose ideas and combine them into co-creative activities. In addition, this process emphasises that orchestrating is about *holding* conflict at bay, since autonomy stresses doing more by oneself, having self-control and less dependency on the power of others but also assuming the risk of having different objectives and perspectives. To that end, commitments among actors are about taking care of each other and maintaining an ongoing equilibrium to not break up network structures (even during deep structural changes), and, therefore, create an efficient path for innovation.

Orchestrating through commitments is about changing the rules of the game and keeping the game alive. This approach presents networks as a more natural way of organizing, thus challenging the Neo-Classical perspective of relationships as transactions- which presents many limitations for innovation studies (Teece, 2009)- and dismissing other human attributes to deal with innovation and cope with radical uncertainty. Compelling antecedents can be found in Psychology or even in classic narratives to support commitment as the force to link networks of innovation. First, among mature and autonomous individuals, commitment is the capacity to care for each other (Bowlby, 1962), assuming consciousness and self-control to choose and set relationships, thus recognising the dependencies that actors' have on each other in a network. The second antecedent is found in Elster's Theory of Binding Rationality (see Elster, 2000) as argued above. In Multiasistencia, orchestrating through commitment contrasts with the opportunism which was previously the source of latent conflict and, paradoxically, served as a centripetal force. This confirms that some social dynamics can be very cohesive but, since they may hide asymmetries, they are incompatible with co-creative processes of innovation. Partnering, creativity and autonomy lay behind IBM Barcelona Lab's idea of "key people in key projects" which sees areas of innovation in particular settings, developing the right staffing policies and generalising what is specific and a market trend.

Orchestrating also implies taking actions in those cases which led to some relationships being dropped while new ones were built, recognising that not all actors in a network can be interested in or have the capacity to establish commitments. Moreover, in the cases presented, commitments were determined precisely in sub-optimal regimes, emphasising how trust is a necessary resource that orchestrators have to foster in order to manage investments in situations where, for some actors, the costs and risks may be greater than the short-term benefits.

5.5.2.- The Dynamics of Co-Creation

Proposition 2: Orchestrating innovation in networks has a positive influence on maximising outcomes when commitments lead to peer-to-peer relationships among autonomous players.

This proposition emphasises that orchestrating is a different way of exerting power, leading to different network structures based on peer-to-peer relations. Being peers seems to be the prerequisite for co-creative activities, according to empirical results. This requires the existence of complementary assets to exchange and create something new. This dynamic affects both actors' roles and the network boundaries.

Actors become more specialised, gaining *greater differentiation* and establishing a new division of labour. In Multiasistencia, the new network structure stressed complementarities but also relational co-dependencies, since trade professionals needed to focus on their repair work, being more dependent on the sales process and market development carried out by the focal firm. The result was that tradesmen and the focal company balanced their relationship. By the same token, banks needed Multiasistencia's service quality to expand their customer business value. This is also consistent with the proposition that, in order to establish a new equilibrium, firms may create greater specialisation and differentiated resources (Wernerfelt, 1984). In the case of the IBM Barcelona Lab, this implied: (1) the capacity to co-create new software solutions; (2) the ability to leverage "external" and "internal" resources to do so (in other words, the "slack" was provided by the first

customers to carry on with the project); and (3) the capacity to integrate new technologies in the existing bases already installed.

In the Multiasistencia case, power, as expressed by exerting autonomy and managing commitments, led to a new set of social values such as the policy of "growing together," implicitly recognising economic value generation and absorption processes. The focal firm also asked its Customer Service Representatives (CSRs) to become "process owners" and to co-create, co-produce and exploit new services based on the co-operative software developed with banks. Multiasistencia gained autonomy as the focal firm, inviting those who did not accept the new schema to leave the network. The new game was now a question of not only sharing benefits, but also costs.

Tables 5.3 and 5.4 below summarise the empirical study of both cases and their respective structural dynamics affecting the roles, relationships and boundaries by balancing centripetal and centrifugal forces. Also stressed is the role of ICTs due to their importance in the dynamics of both Networks.

	Initial Network Conditions (1984-1994)	Network Boom (1995-2000)	Network Change (2000- 2003)	Network Behaviour for Innovation (2004-2008)	Towards a Dominated Network (2009-
Focal Firm Role	Bridging a structural hole	Attempt to be the strategic apex	Management of new commitments	Reference for innovation	Strategic apex
Relationships	Asymmetric (1) Contracts with customers (2) Latent opportunism with providers	Asymmetric (1) Explosion of opportunism among providers (2) Lack of trust and quality in service with banks	Balancing relational dynamics through boundary definition and commitments. Some new relationships created, others are dropped Trust as a deterrent and calculative	Commitments (peer-to-peer) towards co-creation and complementarities for co- creation Trust builds knowledge and identification	Hierarchical control over providers
Boundaries as centripetal vs. centrifugal forces	Promise to customers "Yes, We Can" Not defined with providers	Centripetal forces prevail: Boundaries expanding due to business growth, externalities, increasing returns, "follow the others" and "latent" opportunism	Management exerts autonomy as a centrifugal force vs. the centripetal force of opportunism (and its over-embeddedness)	Commitments balance centripetal forces (increasing returns) and centrifugal forces (autonomy)	New balance under study
ICT role	Operational support	Source of new income and innovation	Co-creation of a BNOS	Automatic boundary control (deterrent) and space for co- creation (knowledge based)	To be analysed

 Table 5.3: Orchestrating Network Structural Dynamics in Multiasistencia

	Initial Network Conditions (1984-1994)	Network Change (1994- 1996)	Network Behaviour for Innovation (1996-2004)	Competence Centre (2004-)
Focal Firm Role	Bridging a structural hole	Management of new commitments	Reference for innovation	Reference for innovation
Relationships	Asymmetric (1) Commitments with Customers (Banks) – some of them latent after project ending (2) Agency-based with IBM as a Corporation – latent asymmetry and opportunism?	Balancing relational dynamics through boundary definition and commitments. Trust as a deterrent and calculative	Commitments (peer-to-peer) towards co-creation and complementarities for co- creation Trust adds knowledge and identification	Commitments (peer-to-peer) towards co-creation and complementarities for co-creation Trust adds knowledge and identification
Boundaries as centripetal vs. centrifugal forces	Centripetal forces prevail: Boundaries expanding due to business growth, externalities, increasing returns and "latent" opportunism	Management exerts autonomy as a centrifugal force (Key People in Key Projects) vs. the centripetal force of opportunism (and its over-embeddedness)	Commitments balance centripetal forces (increasing returns) and centrifugal forces (autonomy)	New balance under study
ICT role	Source of new income, new types of relationships with customers and innovation Co-creation of LAN-DP	Project-based, ad-hoc solutions	Co-creation of WBCC	To be analysed

Table 5.4: Orchestrating Network Structural Dynamics in the IBM Barcelona Lab

5.5.3.- The Orchestrator as the Network Reference

Proposition 3: A focal firm can act as the network reference, influencing dynamic boundary control and trust in the network.

In this proposition, it is important to point out the structural attributes of the focal firm which act as a yardstick for innovation, capable of exerting power and strategic control on a peer-to-peer basis while maintaining the rules of the game. Although the role of peer-to-peer interactions in innovation is underscored, in the Multiasistencia case, the focal firm strengthened its central role and maintained ownership of the digital platform, thus controlling a strategic resource to keep an asymmetrical power balance with respect to the other actors, although it also maintained its capacity to generate commitments. It is important to note that the focal firm changed its role as well as the network structure and its function in the different periods analysed. In the first phases (1984-2000) the focal firm's power was based on concentrating innovation and trying to control the network as a tool to spread its innovations. During the sub-optimal regime (2000-2004) the focal firm's way of exerting power was to exert autonomy, opening new spaces and establishing new commitments. In this transition, the focal firm proposed a new game and changed the rules of that game, balancing relational dynamics. From 2004 to 2008 the emphasis was on co-creative activities to design and implement new services. Here the focal firm exerted power by increasing its centrality as the reference for innovation, controlling resources and opening new spaces, thus rolling out dynamic capacities.

I also propose that the role of reference to manifest commitments and deal with changes can be assessed by resilience as proposed in Chapter 3. In both cases (1) members of the network adopted a co-creative activity with complementary equals (2) focal firms gained power in order to confront asymmetries (3) partners reached agreements with a focal firm (4) amicable (or otherwise) separations (or lock-outs) were reached and (5) a specific structure was ended with the aim of exploring other areas in the future.

116

5.6.- Conclusions

This chapter has analysed Orchestrating as a managerial function to set a new path of efficiency, managing a process of ongoing equilibrium between the actor's benefits and costs which can vary along the way, stressing that setting a path is "changing the rules" while keeping the game alive. Orchestrating Network Behaviour for Innovation is about power: on the one hand, opening new sources of innovation and, on the other, setting commitments which shape network boundaries. This issue has been stressed, particularly in sub-optimal regimes, therefore resilience is proposed. In this respect, management is about dynamically regulating boundaries through centripetal and centrifugal forces in accordance with objectives and circumstances which can also shape roles and relationships.

Therefore, Orchestrating Innovation is an ongoing process to change network structures to manage the players' autonomy and peer-to-peer relationships for co-creation, sharing investments and engendering trust. This leads to network behaviour for innovation and the development of new opportunities. In turn, the latter leads to a "multi-connected" network, emphasising value regimes based on equity, holding conflict, and organizing forward. This serves to highlight: (1) the autonomy and the predominant relational model among the actors in the network (whether hierarchical or peer-to-peer); (2) the locus of innovation, that is, if innovation is centralised and the network's main function is diffusion or if the focal firm is the reference to generate novelty by orchestrating new sources that come about in the network; and (3) how the focal firm fosters communication within the network. Information systems and digital platforms have been highlighted, in particular, as a proxy for commitments and also as a communications system and a creative space. Creativity in social settings may be part inspirational, but it also requires perception and communication in order to become a structural process.

In Chapter 6, I analyze the "optimum state" found in both cases in more detail, where the focal firm orchestrated innovation through peer-to-peer relationships, co-creation and the capacity for opening new spaces and sources of innovation,

117

which allowed it to have the capacity to set new paths due to multi-connectivity, find new experts, end and establish relationships, and develop new digital platforms; a critical infrastructure for innovation as I shall explore in the next chapter.



Chapter 6: Orchestrating Generative Networks of Innovation

6.1. Introduction

This chapter elaborates further on the optimum condition for innovation when Orchestrating Network Behaviour for Innovation engaged in co-creative activities. There is evidence in the Multiasistencia case from 2003-2008 and in the IBM Barcelona Lab case from 1996 to 2004. This formula offered maximum results and outcomes by opening up a space for new voices and new sources of ideas (see Tables 5.1 and 5.2 in chapter 5) which complemented and re-combined with each other, to create novel results with an important economic impact on the cocreation and exploitation of digital platforms as innovation infrastructures.

This chapter presents a framework that describes commitments assumed by the focal firm in the network and compares them with exploitation and exploration factors by comparing the two case studies. In the following subsections, these issues are discussed and the cases are analysed with an emphasis on the network as an ecosystem for co-creation.

6.2. Generative Networks: A Framework of Commitment

Generative networks are expressions of optimality, and rely on Orchestrating through commitments, trust, and a symmetric (peer to peer) model of relationships for co-creation. Generative networks also represent an optimal configuration as network behaviour for innovation. The orchestrator exerts autonomy, opens space for new sources of innovation, and boosts communication between actors. Digital Platforms provide an infrastructure for innovation, giving *more autonomy to actors*, and letting them build new applications to re-use and integrate them with existing applications and

components in order to build more integrative solutions (in the IBM case) and new services (in Multiasistencia case). Autonomy is expressed with the proposal of new projects, to come up with software projects thought of as the development of re-usable components, or in other words, a re-combination of existing resources to expand the resource base, as dynamic capacities suggest. Additionally, we have shown that project teams synthesised existing and new knowledge into interconnecting software modules which were the foundation for the modular architecture of digital platforms.

With this approach, it is pointed out that, on the one hand, the network is the locus of innovation, stressing that Orchestrating is a network capability (dependent on sources of innovation), not only a question of the focal firm. On the other hand, Orchestrating is about opening and managing new sources with, (1) network interactions among actors; (2) spaces for co-creation and generation of new modules/applications/services and generation of economies of scope; and (3) the replication of existing models in new markets, industries and areas (see both studied, the focal firms chapter 5). In cases detected significant inefficiencies within the banking sector's business processes due to their profound knowledge of business and administrative processes in the industry. Given the common problems and broad applicability of these processes, both focal firms quickly found (1) economies of scale and externalities that helped apply their digital platforms to other sectors of the economy with some modifications as well as (2) economies of scope, becoming "innovation factories" for complex, process-intensive information services, balancing greater scope or breadth versus greater focus and specialisation. It is suggested that sources of innovation are dependent on the way commitments are set and their impact on innovations (from exploitation to exploration). To that end, we define the three key modes:

(1) A new business dominant logic, that is, new business models and market creation, as a result of irreversible commitments led by the focal firm.

- (2) Co-creative activities and mutual commitments and adaptation among actors at the technological and organisational levels to ensure resource and strategic fit. Mutual adaptations are basic constituents of looselycoupled structures.
- (3) Digitally enabled commitments, which include programmed information systems with the capacity to assimilate programmed limits of action and provide new functionalities based on data analyses and predictive modelling. With programmed changes, we consider that some automatic responses can be the result of dynamic models of data analysis such as new business intelligence technologies.

These three modes are at the heart of the governing function and also imply mobilising resources, for which we present empirical findings. However, two additional processes are included: one related to high levels of commitment and exploitation, and the other to low levels of commitment and ICTs used for exploration. The fourth mode might be: Dominant (existing) business logic through which networks can show persistence or attachment to previous commitments. The reasons for "persistence" must be carefully analysed as they can be the result of previous commitments or be due to chaotic behaviour, that is, the absence of clear previous constraints or boundaries that can lead complex systems to assume deterministic paths.

Finally, a fifth mode may be possible due to the exploitative role of information systems. Although theoretically possible through "artificial intelligence," this mode would require much more research (and time!) since this question is on the boundary of science. Maybe one day computers and information systems will make commitments, choices and judgements. While this may become possible in the future, it is far from being a reality today. These two modes 4) and 5) are presented to argue that commitment and the level of exploration are independent variables. Exploitation and exploration refer to the scale of the innovation and also the resource mobilisation required, that is, the money and investment, the new relationships andgetting out of old ones, etc. Orchestrating recognises that the greater the organisational capacity, the greater the scale of change (Orlikowski, 1996). This approach to adaptability can be synthesised in the following figure.

121



Figure 6.1: Modes of Commitment for Innovation

6.3. Dominant Designs and Business Models

The first mode is related to *irreversible commitments*, that is, situations where there are few choices and they are leapt into, providing for consistent behaviour which maintains organisational consistency with respect to these choices. This thesis has presented innovation dynamics towards *dominant designs* which can be defined as knowledge that crystallises from: (1) an innovation path, which implies (2) different choices and actions that set this path (3) marching in line with the articulation of preferences. Both the IBM Barcelona Lab and Multiasistencia epitomise these dominant designs with the co-creation of the two digital platforms. In turn, these led to the creation of new markets for "middleware integration" at IBM and "repairs" in the case of Multiasistencia.

In both cases, findings show how digital platforms are associated with new dominant business logic, giving these firms a "referential power" and the capacity to control digital platform developments or their ownership. This is consistent with previous research, since new dominant designs are related to new hierarchies (Van de Ven & Garud, 1993; Utterback & Suarez, 1993; Suarez, 2004)

This discussion is also related to recent proposals in the literature such as linking Orchestration and dominant designs with market and factor creation. In the early stages of any industry or market development, product designs and organisations tended to be organic, with no clear definition of rules, roles and boundaries (Burns & Stalker, 1961), thus permitting networks to be seen as natural ways to describe the stages where investments and co-creation seem to be critical aspects as well as Orchestrating managerial capability.

Centripetal forces lead to super-additive games that serve to: (1) attract new developers, new improvements, and new modular subsystems; (2) accelerate the diffusion among new adopters; and (3) thus create increasing returns, thereby confirming the existing theory which sustains that orchestrators generate a new type of asymmetry (expanding the pie but keeping a bigger piece for their own hierarchical position). Finally, we confirm that dominant designs accelerate the

diffusion across heterogeneous adopters, as cases show, since innovation leads to opening up the possibility of co-operative relationships and generating externalities which can reinforce cumulative returns.

Multiasistencia, for example, created new businesses and new markets beyond the repair service industry, becoming a service factory and managing complex informational processes such as mortgages or transfers of customer information from one bank branch to another. As one manager described, "We have created a specific call centre for this with an application that handles more than 17,000 companies offering direct debit services."

6.4.- Co-creation

Commitments in these cases were initially established through strategic or idiosyncratic relationships searching for complementarities that could generate interest in co-investing in new software solutions. This situation then led to greater interaction and increased autonomy and the different networks developing new ways to carry out their business activities. Through co-creative activities, actors can perceive that they co-share ownership of the network as well as mutual adaptability, resulting in much better time to market.

Co-creation networks can be seen as ecosystems which engage in systemic innovation, while still providing the developmental tools needed to continue improving their set of functionalities. They can however be co-created through a set of derivative developments with greater agility, to meet market demands and better time-to-market, quality and integration targets, while avoiding disruptions and reducing risks.

The new business model developed by the IBM Barcelona Lab, for example, allowed for the creation of a new concept of software production known as "Hybrid Programming." As one manager stated, "The key issue for the Lab was managing a fast time-to-market". Unlike the strategic planning model in place up to then, this concept was based on the development teams who continuously co-operated with customers on "product drops" every few months. When a customer or the sales force asked for a new requirement, it was incorporated

into the software development program. These new requests were placed on a list governed by a cost-based priorities structure (resources, time and complexity) and by profit criteria (applicability and extendibility). One of the managers explained,

"The versions depend on customer needs. Software development runs in parallel, coming out of the work of several teams responsible for each module. The new versions are launched at set intervals – every six months. We have also defined a new delivery process to ensure the quality of every product drop."

During this period, and due to the extensive experience and knowledge accumulated in integration projects, in 1998 the Barcelona Lab launched a new product: the Websphere Business Component Composer (WBCC) which is a line of pre-configured software tools using internet software. This greater flexibility permitted a new modular architecture to be defined based on "components" or small, functionality modules which provided specific solutions to the client's needs.

"It allows software application designers to build quick and flexible banking solutions, re-use existing components and applications, and integrate them into their transactional systems. From the bank side, this is a way to protect their strategic investments in the mainframe."

This emphasises that: (1) autonomy, not coercion, is a critical variable to conducting innovation activities, and (2) adaptability at the organisational and technological levels serves to build core-competences at the network level such as the capacity to co-operate and learn.

6.5.- Digitally enabled commitments

Digital platforms set the rules of the game and network boundaries, as can be clearly seen in the Multiassistencia case. Multiasistencia built a Business Network Operating System (BNOS) to automate the process of managing its contacts with

125

corporate clients and the CCMS process. This system was deployed during the change process enacted from 2000 to 2003. The objectives of the platform were to: (1) determine a pre-defined pattern of behaviour; (2) ensure process control and continuity; (3) ensure scalability and provide the tools needed to manage peaks in demand and exceptions; (4) ensure information and communication flows; and (5) act as a deterrent to monitor the defined pattern of behaviour and Service Level Agreements.

According to one Multiasistencia manager,

"We have substituted all human communication elements in the repair service, except for incoming calls, with machines. We have developed a Business Network Operating System that supports our sophisticated process, allowing people to manage exceptions to the main process"

It follows from Proposition 1 above that the use of ICTs has allowed the network to automate many processes and thus release limited resources. As one executive indicated,

"Our model is that of making the whole process transparent. The integration of processes on the internet has allowed us to define a "virtual machine of finite states" with very precise expiry dates between states. The application has about 100 controlled steps linked to our client and professional applications. When an exception occurs, we must offer all the information available to the person directly responsible for its management to make the decision-making more effective."

This exploitative use of ICTs implies the use of structured information and problem resolution that can lead business intelligence systems to develop predictive models. As one manager of Multiasistenica recalled, "data is essential" because it provides more experience to people running the business and facilitates a learning process of day-to-day activities, all of which result in internalising and mobilising knowledge. Similarly, as a result of having more information, Multiasistencia has developed experimental uses for its ICTs, allowing the firm to develop new solutions and services ("Factory of Processes") based on informational exploration.

Digital platforms contain programmed behaviour, including, for example, IT standards at different levels of connectivity and integrability: from data to processes and integrative behaviour. This allows a network information processes to be supported throughout the network and permits easy plug-and-play. Informational and technological limits can create entry barriers, but they also have lock-in effects or imply exit costs (see Shapiro & Varian, 1999). In the Multiasistencia case, the BNOS supported the new model to co-ordinate the network of trade professionals as the *framework for network behaviour and boundary management*. The BNOS is the basis of the partnership since the Internet portal is the *source of jobs*: (1) trade professionals are responsible for "pulling" jobs from the system; (2) the BNOS acts as a *deterrent* as it keeps track of Service Level Agreements (SLA); and (3) it monitors and supports the new information system. As one manager added,

"For us the Internet does not just contribute value; it has also been the facilitator of the entire new operating model that we have developed. Everything is automated and we, the people, manage the exceptions; we manage the proactive quality. Simplifying the operating idea for Multiasistencia, the new process came about when we said: 'We're going to put it on the internet'."

Moreover, the BNOS reduced transactional costs due to the reduction in administrative tasks and paperwork arising from the calculation, invoicing and collection of payments from corporate clients and end customers (Multiasistencia executives estimated that, for some trade professionals (TPs), paperwork accounted for 50% of their activity, even though Multiasistencia invoiced on their behalf). As one trade professional added, "Multiasistencia has a self-invoicing

127

system which means that once you finish the job, you don't need to issue the invoice anymore."

The orchestrator exerts power through control of the digital platforms as a resource. In Table 6.1 a synthesis of findings at "focal firm" level and "individual level" (elaborated upon in chapter 7) is presented. It is important not underestimate the question of which actor owns the platform, and other issues such as intellectual property are also critical. In addition, the platform's capacity to increase the lock-in effect among actors in the network is worth bearing in mind. But above all else, the platform distributes the power to co-create, that is, the platform can be used as a tool for creativity, increasing collegiality and a sense of ownership while, at the same time, continuously evolving to provide new offerings and service innovation.



Governance Modes	Multiasistencia		IBM Barcelona Lab		
	Functions	Managerial Enablers	Functions	Managerial Enablers	
Digitally enabled commitments	 Assignment of jobs to the appropriate trade professional Automatic invoices Online quotations for repairs Claim management 	 IT: (1) Digital Platform internet- based pervasive system connecting 100 corporate clients and 11,000 trade professionals; (2) Web Services to easily connect to the network and (3) dynamic boundary management 	 Multi-channel Mid-Layer Software Data conversion from mainframe to contact channels such as Contact Centre and the internet 	IT: Websphere Business Component Composer (WBCC) as a component-based platform for multi-channel integration	
Co-Creation	 Proactive quality; lifecycle management Ad hoc services and exception management (service improvements) Customer-centric innovation for fast time-to-market: Participation architecture to enrich the DIGITAL PLATFORM as a network resource 	 Active dialogue and conflict management Key People in Key Projects IT: Internet/Web Service Integration and Call Centre Integration with other partners 	 Customer-centric innovation: incorporating customer requirements Key People in Key Projects: rotating people and changing roles LAN-DP and WBCC as component-based solutions founded on knowledge capitalisation 	 Active dialogue to make sense of reality: "Application of technology to real problems" Conflict management Creativity – dialogue 	
Dominant Design and Market Creation	 Opening new geographical markets Creation of SMEs for repairs "around the network" – spin-out-type program Firm-driven innovation for new services Defining new markets such as Healthcare and the Public Sector Increasing scale and cost reduction for network externalities 	 IT Flexibility: Factory of processes and system renewal Knowledge generation to create new ideas Exert autonomy 	 Strategy: "Choose a series of key projects aligned with strategy" Creation of the Lab in 1984 Creation of new markets since 1994 Creation of an SME to develop software solutions, spin-out-type program 	 Information search and innovation screening Active information exchange through formal and informal means Exert autonomy 	

 Table 6.1: The Three Modes of Commitment and Innovation

6.6.- A dynamic view of commitment

Figure 6.1 shows the three levels of commitment ranging from irreversible, cocreative, to "automatic" and their relationship with exploration and exploitation. Figure 6.2 below illustrates the commitment framework graphically, depicting the dynamic effects of commitments. Irreversible commitments (from point "B" to "C") lead to dominant designs and dominant business logic, as "node anchoring" and convergent behaviour, although the path is based on a set of experiments, a set of steps until the new business logic is right.



Figure 6.2: A Dynamic View of the Commitment Framework

This process is based on mutual commitments between network actors in order to develop co-creative activities. Similarly, this process may serve to attract actors who can incorporate new complementary factors into the network and into the digital platform due to increasing returns and externalities or super-positive sum games. Attraction can be due to opportunism, as discussed in Chapter 5, or even to natural euphoria at the beginning of any innovation process (Van de Ven et al., 2008). Here we argue for the importance of commitments as criteria for demarcation and boundary setting in orchestrated networks of innovation. Running digital platforms can develop two roles. The first involves automatic boundary setting for network behaviour (as the Multiasistencia case shows).

In digitally enabled networks, some structural patterns or behaviour are driven by an information system that sets a specific set of constraints to limit degrees of freedom, precisely to establish a style (from the many which are theoretically possible) or, in business terms, a Service Level Agreement. The second role that information systems can play is that of a standard and standardisation platform for the development of new software modules for the specific function of frontoffice and back-office integration in banking systems. Although business functions are different, the social or structural function is similar, that is, setting boundaries for human activities and setting standards for behaviour. This is represented as a horizontal line that constrains the "levels of freedom" in our diagram of cycles, with the hypothesis that "ups" and "downs" can be held at this level, that is, setting boundaries of action.

Multiasistencia was entering new markets. This had two implications: geographical with entry into the UK and France, leveraging the scale and flexibility of its digital platform, and sectorial, the company's branching out into the healthcare and automotive repair industries. Furthermore, the firm also began orchestrating information from different local government agencies, providing a unique point of contact for citizens in different cities. Meanwhile, the IBM Barcelona Lab developed different innovative products to integrate a multichannel offering. At the moment of writing this thesis, approximately 1,500 banks world-wide were using this company's solutions. Therefore, these two innovation projects which were the result of taking a different look at reality, in turn, led to the creation of a new market and a new business dominant logic.

After analysing upstream dynamics, we will now discuss downstream ones and sub-optimal regimes. Once a new dominant design emerges, the competition (both external and internal) that wants the dominant position will focus on price to take advantage of the new design's weaknesses. Competition will tend to offer the same product or service at better prices, with an emphasis on large-scale production (see Suarez, 2004). The main risks are technical community

disintegration and an increase in competition. Orchestrators focus on the design and deployment of specialised assets, that is, the generation of new modules through co-creative activities, focusing on more specialised assets which are more difficult to copy. This is clearly seen in the more mature phases within the IBM Barcelona Lab and Multiasistencia cases.



Figure 6.3: A Dynamic View of Orchestrating

Figure 6.3 is a continuation of the previous figure (6.2) but in which the risks and conflict associated with structural evolution during peaks (dominant designs) and troughs (sub-optimal regimes) are added. These are explored more in-depth in the next chapter.

In 2000, companies had the choice of subcontracting Multiasistencia or building their own infrastructures. From 2005, due to the appearance of four new firms providing similar services, banks could split their claim management and repair services between two or three firms. Perhaps this is why Multiasistencia's management decided to include some key trade professionals as internal employees along the way evolving towards a bureaucracy-like firm.

Multiasistencia's pioneering effort exerted a centripetal force not only on trade professionals and banks, but on other firms that saw this as an attempt to structure repair services which were previously unstructured and thus inadvertently create barriers for new entrants to the market, competing in terms of price and scale. In 2009 the repair market was worth \in 2,500 M in which 20% was "free", that is, services were not controlled by major insurance companies. Multiasistencia controlled approximately 50% of this free market with a turnover of \in 250 M. The rest was distributed among 4 other major players. Since this service was driven by market rules and, due to the financial crisis that started in 2008, many small banking institutions started to sell this business unit to insurance companies, thereby challenging the market, since major insurers tended to in-source the service. Transaction Economics could have predicted this situation which, according to this research, occurred due to the absence of commitments serving as boundary criteria. If the criteria to mobilise resources are driven by price, then Transaction Cost Economics can predict that valuable resources will tend to be in-sourced by firms. The process of in-sourcing literally dries up the market and makes it a hostile place for newcomers.

In the case of the IBM Barcelona Lab, once the dominant design was released, the major risks it faced included the technological community's dissolution due to internal competition (a hypothesis) and internal policies promoting off-shoring. When a company depends on only one product, this can be a question of time since any software application, such as the first LAN-DP product or WBCC, will quickly enter a mature phase (4-5 years) where: "80% of the value perceived by customers was based on customisations," as one IBM executive related regarding their product.

The Websphere platform controlled approximately 20% of the system integration market in 2009, according to Gartner. One can suppose it controlled a much higher figure in the banking sector. This same source (see Norton, 2009) claimed that IBM could offer quick time-to-market and increased customer satisfaction for IBM's *agile Websphere implementation methodology*, based on 3-day workshops (see Figure 6.4). As this research shows, this is due to more than 25 years of proven experience, at least for the IBM Barcelona Lab, which is now a part of the multinational's global service operations and is, therefore, considered a local element in global innovation (see Nohria & Ghoshal, 1997).



Figure 6.4: Agile Implementation of IBM's Websphere (Source: Norton, 2009)

This thesis has emphasised that centrifugal forces may have affected both cases: once the market is created, it follows its own rules, and firms tend to adapt to these rules following well-known principles. It is, of course, possible to start a new orchestrated path, as divergent behaviour, finding new opportunities, as is proposed in Figure 6.3 above, to avoid the tendency to create sub-optimal regimes and further cycles. This is, naturally, a hypothesis that will require further research.

6.7.- Propositions

From the theoretical framework presented in this chapter and the findings of the case studies, we formulate a set of propositions that emphasise the role of networks and information systems which link and qualify the different modes of commitment to guide future research in this area. This cross-case analysis serves to highlight two focal firms and their establishment of a strategic direction and determining paths to efficiency, revealing a capacity to co-ordinate actions and build digital platforms.

The IBM Barcelona Lab specifically co-ordinated a business network of software innovation for the banking and finance industries. Digital platform development was the main activity and task for the innovation network. Customisation was enabled through the interactive relationships established by the Lab and its customers to integrate new demands and enrich existing solutions and systems. This enabled the dynamic analysis of commodity-specialist input factors and the identification of optimal sourcing options. While for Multiasistencia, the main ethos underlying its business objectives was to satisfy customers' needs and requirements within the limits governed by SLAs and information systems, for the IBM Barcelona Lab, in contrast, bi-directional adaptation was performed through negotiations on project scope, terms, conditions, and deliveries.

6.7.1. Generative Networks

Proposition 4: Generative Networks are optimum configuration for self-renewal and dependent on the predominant symmetric relationship dynamics for co-creation.

By 'generative' we refer to new sources of innovation and communication to engage in a multi-disciplinary and multi-connected network. Orchestration can new commitments as driven by the focal firm. This leads to new business development and new market development, co-creative activities and digitally enabled changes. Hence, it follows that increased levels of strategic choices and commitment lead to more radical levels of innovation through co-creation activities and a new dominant business.

Moreover, we can deduce that innovation is positively related to the firms' selfrenewal abilities. In Multiasistencia, change took place by the company redefining its commitments from a previously chaotic set of opportunistic relationships; in the IBM Lab, the latter simply had to open up to create relationships with its customers in order to co-create and leverage power inside IBM with this important resource. Mutual agreements, trust, and a business-tobusiness symmetric model were instituted. In some cases, the lasting existing relationships were more prone to be dropped. At other times, they evolved to a new schema.

6.7.2.- Digital Platforms and Generative Networks

Proposition 5: Digital Platforms serve as proxies for commitments among network actors.

Digital platforms are digital infrastructures to generate innovation. Trust may be impersonal, for example, with respect to the technology and the network itself (Shapiro, 1987). In the research presented in this paper, digital platforms played a critical role, serving as proxies for trust in the networks by providing transparency and being built as a result of shared investments and co-creative activities. Indeed, a digital platform is a space for information-sharing that allows for the transformation of the network's knowledge base. The digital platform plays a key role in network domains by: (1) exerting boundary control and surveillance, allowing the creation of network externalities and increased returns, network scale and lock-in and (2) expanding network domains of knowledge as a result of co-creative activities, continuous improvements and the generation of new markets. This underscores the role played by technology in fostering the dynamic interplay of capabilities. This interplay leads to proposals for the creation of new services through the assimilation of new knowledge and reworking the existing knowledge base in order to expand value options and increase returns. Multiasistencia focused on creating a new digital platform, its Business Network Operating System, allowing the focal firm to manage adaptation among participants, structuring and integrating information between nodes and links, and allowing firms to connect and disconnect. Furthermore, the new BNOS spread out along the network and helped to establish structured patterns of behaviour, institutionalising them and allowing free resources such as information to flow with higher levels of adaptability.

In addition, Multiasistencia wanted to increase the availability of information in real time, along with reliability, transparency, automatic tracking, control,

scheduling, and routing optimisation. The result of having more information was an increase in the level of control over the processes involved (not over employees through direct supervision). Another similarity is the establishment of more or less stable electronic links between network participants. The IBM Barcelona Lab, in contrast, managed a large network of personal contacts and relationships which gave rise to more of a process-like activity since they developed projects. Platform owners can benefit a great deal from innovation in complementary products and services added to the platform. They also benefit from the overall system: development costs drop as the scale of developers or contributors increases. Learning costs also decrease in co-productive innovation processes. Another issue is how to set criteria to provide the *right incentives* to the right participants. It is also important to understand which talented contributions are considered valuable and how they should be brought into the community, and how incentives change as the knowledge base changes (Pedersen & Larsen, 2006). This can be considered an important line of future research.

6.7.3.- The Role of Orchestrating in Factor / Market creation

Proposition 6a: Orchestrating Network Behaviour lead ICT-enabled networks to optimum states for innovation which engender the dynamics for market creation

Proposition 6b: The Orchestration of (1) irreversible commitments, (2) co-creation and (3) digitally enabled changes has a positive impact on network business growth, self-renewal and market creation.

Proposition 6c: Orchestrating enables explorative processes such as co-creative actions towards dominant designs.

This thesis has proposed that Orchestrating can generate new markets, although this is a first step to which more research must be devoted. Markets can be seen as a new game, although not exactly driven by trust. In the first steps though, when activities do not have enough scale, it seems that trust and commitments are essential to create communities. Once the super-additive games create a critical mass, the logic seems to be price driven and only competitive, thus drastically reducing the degrees of autonomy needed and dismissing the role in management in market/factor creation activities. Market logic tends to dissolve existing communities and increase competition and opportunism, thus reducing autonomy or, in other words, adding more constraints that can diminish the ability to innovate. Perhaps this is why there is no innovations market.

However, Orchestration can generate new commitments driven by the focal firm which can lead to new lines of business and new markets. Hence, it follows that increased levels of strategic choices and commitment lead to more radical levels of innovation through co-creative activities and the new dominant business. Moreover, it can be deduced that innovation is positively related to the ability to self-renew. This implies that there is choice regarding a new path. Management becomes an exercise in value generation by defining the limits of appropriate actions, commensurate with suitable expenditures of organisational resources. In other words, a criterion of successful management is knowing when to allocate limited resources to the more demanding long-term choices and commitments.

In both cases studied, significant inefficiencies were detected in business processes within the banking sector due to their profound knowledge on the industry's business and administrative processes. In this sense, they became specialised in complex, process-intensive information services, balancing greater scope and breadth versus greater focus and specialisation.

6.8.- Conclusions

Orchestrators exert power through commitments among autonomous parties, leading to the development and use of new digital platforms. In the two cases analysed, the focal firms managed co-creative activities and adaptation among participants, structured and integrated information among nodes and links, and allowed firms to establish compatible goals. The use of ICTs permitted the networks to: (1) automate many processes and/or (2) standardise activities, and

thus release free resources (or slack), as a result, balancing exploitative modes with increased interaction and information sharing.

As such, the new information system, spread out among the network, helped to establish structured and institutionalised patterns of behaviour that led to market creation through a dominant business logic and dominant digital platform design which attracted other actors, thus creating a critical mass which, in turn, led to a new market. While this is very clear in the Multiasistencia case, it is not so obvious in the IBM Barcelona Lab case. In the latter case, Orchestration was a temporary activity beyond the realm of the firm's traditional business activity, as was the creation of a banking system integration market and its agile implementation capacity (IBM has not disclosed corresponding figures on this matter). In addition, between the game played by the Lab and the market it created, there was another structure called IBM with its own policies, legal rules and boundaries, power structures and global strategies.

Now I turn to the individual level of analysis, that is, managerial capabilities, a theme developed in Chapter 7.

Chapter 7: Network Leadership

7.1.-Introduction

This chapter uses the normative approach, constituting a final reflection based on the empirical findings and adding a more personal view on the individual dimension of Orchestration and possibly even a novel approach to Network Leadership. The chapter discusses the full potential of leadership in terms of its proposed organisational capacities: the ability to organise, to set economic and performance objectives, and to hold the inherent conflict which any innovation implies at bay.

Thus far, we have discussed focal firms' power capacities to steer network behaviour for innovation by exerting a role as reference for the network, controlling boundaries, managing dynamics and transitions and incorporating digital platforms as structural resources. We have also examined network behaviour, the establishment of programmed or digitally enabled commitments, and the use of digital platforms as spaces for co-creation and a means to generate increasing returns to then create dominant designs and new markets.

Individual dimensions, however, are always complicated. Although this thesis is grounded on the notion of autonomy to understand networks as voluntary associations, we have also discussed the importance of commitments to set network boundaries for innovation. In terms of leadership, it is quite a challenge to contribute something, new due to the plethora of literature (academic, managerial and fictional) available on this topic.

This chapter is organised as follows. First, we explore conflict and then examine the way orchestrators manage relationships and power to set direction and performance, and to control this risk. Since the relationship between leaders and followers has already received attention in the literature (Kets de Vries, 1980; 2003; 2009), this chapter focuses more on inter-managerial and inter-leadership relationships, a relatively unexplored area in innovation studies.
To do so, first, we discuss managerial capacities at the individual level. After a reflection on strategic leadership, we briefly examine the essential traits observed as the result of empirical research. Second, we discuss authenticity which is a central notion to fix commitments and trust in relationships, the critical activity of a Network Leader. We then explore the following dimensions: (1) conflict management, (2) organisational capacity and (3) performance. Next follows a discussion of managerial capacities, and the idea of innovation as a conflict, analysing power and politics.

Lastly, we analyse how leaders engage in relationships with other leaders or managers. What the role of commitment between people is. What the relational dynamics between them are and how these influence the critical process of capturing value from innovation.

7.2.- Network Leadership

According to the empirical work carried out, the fundamental characteristic that defines leadership is the capacity to attract others through the ability to break the rules. This is a major force of attraction at the individual level, as confirmed by the literature (Kets de Vries, 2003; 2006; 2009). To that end, leaders have a very clear sense of the rhythm of the music of organising, since they can perceive early signals of change and choose the right moment in which to move. Although for many leaders, having followers is a factor of contingency and risk, this is why leaders may be alone and, on some occasions, isolated (Kets de Vries, 2006; Burns & Stalker, 1961). The major risks of isolation in business ventures are losing track of reality and incorrectly mobilising organisational resources. Defining a path for innovation is all about setting actions in response to real commitments.

This thesis stresses that Network Leadership comprises the ability to *govern* relationships, the complete opposite of isolation. Leading through commitments is emphasizing power and trust. This is achieved by Network Leaders defining boundaries (as sets of constraints), giving co-operative actions a sense of purpose and meaning. Leaders then exert autonomy by fomenting a process of

sense-making and experimentation in order to set a path towards innovation, facilitating peer-to-peer relationships for co-creation.

Leaders (and their teams) detect new waves or new technologies early on. The IBM Barcelona Lab can be seen as a good example. Its leader (and her team) detected that PCs represented a revolution for the banking industry. By the same token, the founder of Multiasistencia and his team created the repairs market (when, previously, the market comprised a highly fragmented set of activities). In both cases, Network Leadership obtained the full potential to understand relationships with other leaders (in banks, for example), with their teams, with providers and with managerial colleagues (new management in the Multiasistencia case). The critical capacity is to build teams, peer-to-peer associations, where everyone takes care of everyone else.

The difference between the two cases is that the leaders in Multiasistencia were able to create a new venture from scratch while the IBM Barcelona Lab is a story of how rules changed within an existing game through new commitments, fomenting innovation and structural change such as new roles, relationships and boundaries of action. The IBM case reveals how part of a commercial branch changed both its role and function by becoming a Lab, thus proposing a change in the rules, conflicting with the "existing game," namely, IBM and its structure.

7.2.1.- Innovation and Isolation

Isolation and Innovation are close phenomena. When new management decided "to place Multiasistencia on the internet", 80% of the Board of Directors was against the idea, seeing this new path as "temporary madness" and certain that the new management team wouldn't last very long. Similar things happened in the IBM case. When the Lab's staff communicated their plans to place PCs in bank branches, as one manager related:

"People, even within the company, really thought we were crazy in the beginning."

As novelty comes about, innovators are considered as "outsiders", according to this researcher's observations. It is important to note that in Multiasistencia, the leaders of change were newcomers, but in the case of IBM, they were perceived to be outsiders or "crazy." Isolation is normally overlooked by the heroic interpretations found in recent approaches to innovation (see Teece, 2009). However, it is a classic argument that very thin boundaries are found between leaders: there may be heroes and scapegoats. Managers of innovation have to deal with tottering organizations affected by political conflicts (creating deep structures) and those associated with managing transitions and organisational loss (examined in more detail in section 7.4).

Depending on the individual, these situations in organisational life produce different degrees of pain. While an innovation does not produce trust, since it often represents a political change that may question future promotions and careers. And potential losses confront people with death (Bowlby, 1985). None of these are allies of trustful environments. Paradoxically, in contrast to what common wisdom may suggest, these are the main reasons why leaders do not tend to be surrounded by trust, making it very difficult for them to obtain commitments and, therefore, increasing the risk of isolation. In general terms, leadership does not presuppose commitments or trust at all, and on many occasions it is just the contrary (Kets de Vries & Engellau, 2004).

Isolation is a subjective concept since, for some, the degree of autonomy may fit into this category. Isolation means not having any trusting relationships within the social system or, in other words, with anyone else. Surprisingly, isolation is praised (possibly unconsciously) for innovative ventures in academic literature. These academic works tend to use the metaphor of a nursery that needs to isolate new-borns from viruses and bacteria. This ignores the basic notions of Psychology, such need for attachment and care, discussed above, as well as the need to be part of a social system (Bowlby, 1962). It also ignores classic managerial literature, such as Barnard's (1938/1968) *The Functions of the Executive*, which claims that "Human organisms *do not function* except in conjunction with other human organisms" (p.11), or Burns & Stalker's classic approach (1961) regarding managerial isolation which can be seen as an extreme of autonomy. In reality, this literature refers to the opposite meaning. Isolation

leads to extreme dependency on others. By contrast, the researcher has observed how network leaders strategically manage this contingent factor by carefully managing their own personal attributes (even a personal "brand") and relationships, as is also examined in the next sections.

Leaders may demonstrate emotional dependency or they may be "reactive" to the opinions of the group (and social trends), thus undermining productive leadership towards innovation. They may do this by compensating for the isolation of "ups" and "downs" by over- or underplaying their role (Burns & Stalker, 1961) and then taking too many risks, instead, developing, dramatic or "heroic" behaviour or bearing other's burdens, costs and opportunisms. We can see an example of this phenomenon in the Multiasistencia case (although we recognise some degree of speculation), when the focal firm was in its period of economic boom when it attracted providers by establishing opportunistic relationships and assuming all co-ordination costs (see Chapter 5). In contrast, underplaying their role may have meant not assuming the hierarchy (and asymmetry) associated with command, and thus promoting friendship, mobs and "circles" instead of professional engagement and commitment. This is more likely to happen during "down" periods. The leader may show empathy, but not friendship. These two attitudes are futile on the individual level. Since they project an image of falseness (Kets de Vries, 2003; Winnicott, 1971), they can worsen isolation (Burns & Stalker, 1961, p. 143) and may separate the leader from his/her area of work: eventually, in reality, worsening the "ups" and "downs".

7.2.2.- Empathy

This thesis has discussed the structural dynamics of "ups" and "downs." In social settings, these cycles can be easily associated with euphoria and loss; as we have argued, people sometimes find innovation painful. In these situations, isolation can be a defence mechanism (Argyris, 1990; 2004) avoiding pain though the use of informational filters, only hearing favourable comments while dismissing others (Burns & Stalker, 1961). *Innovation euphoria*, during the first moments of any innovative venture, can also be painful (ex-post). There are

many examples in economic life of resource misuse. The argument here is that orchestrators drive teams towards understanding these feelings, these emotional moods. Empathy is placing oneself in another's shoes, and it is important to regulate activities and dialogue. It is about people setting networks and linking current activities with future ones by setting a path of efficiency, meaning that they have to manage real people with real emotions that manage resources and real commitments while avoiding pain and isolation which can stem from the "ups" and "downs" mentioned. Therefore, this stream merits more research in innovation studies.

7.2.3.- Exerting Autonomy

Popular wisdom sees leaders as sources of inspiration and admiration (Kets de Vries, 2003; see also Mintzberg, 2002) that are needed in tough times. "We need leadership" people may say. To admire someone means having a favourable opinion of them based on approval, adoration and idealisation, all of which hide poor self-esteem by showing effusive respect to others; therefore, admiration is both a rationalisation and denial of the deadly passion of envy. The ones admiring choose an asymmetrical relationship with others (the objects of their admiration). This inevitably leads to evaluating one's performance by *comparing it to the admired person's.* The performance of the first will always be worse and lead to envy. One may argue that admiration is the surface of a deep structure grounded on envy. By exerting autonomy, orchestrators encourage testing one's own powers and performance, and they actively engage in co-creative and productive commitments with others. In this way, *orchestrators work at deep structural level*.

7.2.4.- Resilience

We argue that leadership is achieved through commitments. This starts at the individual level by committing to oneself: to be true to business or one's own interests, and to hold conflict and open up a new future. It is about binding oneself to the mast.

We also argue that this self-commitment could be called resilience. It consists of exerting self-control as the capacity to manage emotions. This can vary, however, from over-control to moderate control and under control, with the optimum being somewhere in the middle. Over-control means too much constraint and inhibits responses and communication (under-playing); it also reduces expressions and results in the incapacity to process information (Bowlby, 1980), finally, leading to brittleness. Under-control means dramatic behaviour (over-playing) and too few restrictions on information processing and communication (Bowlby, 1985). Resilience thus means modifying the level of self-control according to the circumstances, being resourceful and flexible to adapt to a changing situation, while possessing the ability to process competing, sometimes painful and conflicting, information.

7.2.5.- Authenticity

At the individual level of learning, generating trust and avoiding isolation can be synthesised into the first and foremost way of wheedling power: exerting power over oneself or committing to (or caring for) oneself. This is authenticity, which etymologically means *exert power over* and is directly connected to commitment with oneself and resilience. Authenticity is also related to assertiveness and the search for co-operative and co-creative formulas to solve conflicts.

According to the findings related to leadership in this research, this natural capacity must be complemented by managerial capacities such as: (1) organisational skills, (2) achieving economic or performance results, and (3) conflict management.

7.3.- Conflict Management

Innovation is conflict since it challenges the existing rules of the game. We have argued that power is about holding conflict. Leaders may demonstrate autonomy and exert autonomy over their teams, boosting creativity as the vital force for innovating. Innovation may generate conflicts due to objective mismatches, strategy and resource misalignments, operations and above all, power dynamics. Here, the focus is on orchestrators and examining their capacity to establish commitments to hold conflict (see Chapter 2). This is related to the idea that precommitments are strongly associated with exerting power over visceral human passions, such as fear, envy, pain and hostility. These can worsen during periods of change and innovation due to the changes in political systems and the inherent loss that any innovation represents, which can lead to the misuse of resources.

7.3.1.- Role and Relational Conflicts: Re-thinking Centrality

As has always been the case, new ideas conflict with human interests at empowered and political levels. This is Machiavelli's classic argument: leading change is hostilely received by those who are taking advantage of the existing order, and may even be exploiting current asymmetries which are repressed or hidden by coercion or arbitrary measures. As one Multiasistencia manager claimed,

"Innovations are good for the organisation, but they can be bad for many people. By proposing changes in processes, products or ways of doing things, you are unintentionally questioning their work."

The notion that innovation is "good" for organisations and "bad" for people leads to the natural conclusion that, at best, innovation will obtain mild support from those who may obtain some benefit from the changes the innovation produces in the future. Not all leaders promote change, and there will always be other leaders to represent people that do not want a change to the status quo. In both cases, the leaders are transgressors, since they do not follow existing rules or commitments. Chapter 2 has argued that the orchestrator can be associated with the role of the entrepreneur who breaks the rules of the game and proposes a new rule. In Multiasistencia, new management played this role in 2000. The role of the critic, in opposition to the conservative one, was essentially played by the company President's and COO's³ former "right-hand man." According to one Multiasistencia manager,

"When we joined the firm, there were bets about how long [we] were going to survive in that context. Our predecessors lasted a few months. In our case, the founder took a step back, becoming President, and gave us full support to conduct the change needed."

By stepping back, the President adopted an institutional role to try to settle disputes. Through this research, we have also found mentor roles in major banks and among key trade professionals who, by agreeing to co-create a new venture, supported the process of innovation and also sponsored it by committing to this new business and risking financial resources to advocate and support Multiasistencia's BNOS initiative. In the IBM Barcelona Lab, three roles were played within the focal firm, those of: (1) the champion (or advocate), (2) the business development manager who linked software production to different IBM units and customers, and (3) the architect who ensured software quality production. In Multiasistencia, the roles of architect and business developer were quite similar, as was the role of champion. This therefore, confirms the roles proposed by Hinterhuber (2002). In both cases, the focal firm generated new business (playing the role of nurturer as well) but also served as a reference for the owner of the technology, expanding the pie and acquiring a bigger slice (as suggested by Dhanaraj and Pharke, 2006).

The argument here is that role conflict needs political mastery, i.e. knowing how to manage relationships, since different leaders have different approaches or priorities to determine the path to efficiency, and they may feel endangered due to their current status. If we are asked to form a network, we will normally place ourselves at the centre. Our relationships would be our set of choices (but also the choices others may have made concerning us). As such, we have considered the IBM Barcelona Lab as the "centre" of orchestration, comparing it to the Multiasistencia case. The main difference is that while Multiasistencia started from scratch, building a network and selecting the relationships to structure its

³ Chief Operations Officer (COO)

business, innovation at IBM (as in many other firms) came about with existing dependencies, structures and relationships. Multiasistencia is a metaphor for the evolution of deep structures compared to IBM, which can be considered an example of deep structural change within the global, formalised IBM structure.

Having examined the results, we now take a different perspective, attempting to observe this Orchestration from the perspective of IBM headquarters (HQ), as depicted in Figure 7.1 below. The Lab does not have to be placed at the centre, according to one manager, "in a country - that is, Spain - in a corner of the world - that is, Barcelona.

This perspective enables us to better examine how the Lab (and other similar initiatives) can be considered to have exhibited deviant behaviour or an act of defiance to formal HQ authority. In this situation, the HQ is challenged to regulate the appropriate level of autonomy to ensure cohesion and coherence for its strategies and structures. However, this process can also be seen as a natural way to express new ideas, a desired path, as in formally landscaped gardens where people tend to choose natural paths over the pre-designed ones. The IBM Barcelona Lab case shows that focal firms and innovation can happen anywhere, despite existing rules of the game and structures.

As already argued, Orchestration has an antecedent in innovation dynamics within differentiated networks (Nhoria & Ghoshal, 1997). On the surface level, this organisational phenomena is called "local for global", this can be seen in the Barcelona Lab case, since an initially local innovation came to prominence in the global market of the banking industry and was adopted by Corporate IBM, first as a set of products and then as a practice. In this and similar cases, local leadership may have to confront global leaders for the resources required (Barlett & Ghoshal, 2002; 1990), thus presenting a conflict between the role that local managers would like to play, HQ's pre-defined role and structure, and personal subjectivity with regards to this conflict of roles. The literature emphasises that, in order to be successful, local innovations require a local critical mass, and fluid communication between the local subsidiary and HQ.

Yet how does Orchestration exert power through commitments? The obvious observation is that commitment is a two-way street and hierarchical superiors who are willing (and able) to assume these commitments are essential. Commitments require peer-to-peer relationships, yet HQ controls scarce and unique resources, such as career paths, promotions, salary policy, rewards and punishment.



Figure 7.1: Orchestrating as re-thinking centrality

7.3.2.- Political Mastery

In some firms, the role conflict associated with innovation is not managed through commitments. Arbitrary measures involving coercion are well-known power games found in *divide et impera* and/or *tertius gaudens* tactics (see Burt, 1992; Álvarez & Svejenova, 2005). By *tertius gaudens*, we refer to the fact that HQ can play the "tertius" role as the institutional guarantor of the structure - while instigating conflict (or letting it continue) between innovators and innovation critics. This means taking opportunistic advantage of the normal conflict any innovation implies. Orchestrating activities can be targeted by applying simple *divide et impera* strategies taking advantage of any conflict arising by dealing with different members asymmetrically and applying arbitrary methods to create mobs or "inner circles" to isolate other groups. This may be the reason why some scholars argue that innovation must be initiated at the top of the hierarchy (Hamel, 2007), placing the responsibility for change firmly at the institutional level.

In addition, innovation outcome, as well as the resulting team performance, has its cycles. It is easy to wait for decline in the absence of the level of investment which would have ensured long-term viability. It is worth pointing out here that while some new ventures may be aligned at a given moment in time, others may not be. For example, this could have been the case of the IBM Barcelona Lab if it had clashed with the US Lab responsible for the company's IT infrastructure for the banking industry.

We have seen that IBM allowed autonomy for innovation, giving space to IBM Barcelona Lab to take advantage of the new dynamics associated with the PC and internet developments. Moreover, at the strategic level the Lab wittingly developed: (1) important relationships with banks that counter-balanced internal competition; (2) a strategy of carefully feeding internal relationships to avoid the perception as "outsiders." In this way, the Lab gained the support of the Spanish banking system, as well as the "internal" IBM sales division which saw the Lab as a strategic resource to better control customer behaviour, and then through the

installed base it acquired a better position for new business; and (3) ways of taking advantage of the revolutions represented by PCs and the internet in a new era where technology was no longer going "to be invented only by IBM," as some IBM employees related.

This leads us to emphasize the need for "political" skills, since different leaders consider different approaches or priorities to determine the path to efficiency, but all leaders need to know how to best position themselves to manage these differences. Relationships in global formalized firms are asymmetric and hierarchical. This approach leads to evident and implicit conflict between the various visions, objectives and resources thought to be needed to capitalise on innovation. The question here is how leaders in global firms can make commitments with innovators or, in contrast, why they disregard innovators, leaving them alone facing conflict with critics and product cycles. Administrative Theory pays special attention to opportunism, assuming that managers need to control subordinates. However, it is quite silent on opportunism concerning relationships among leaders. Opportunism - as opposed to commitment (regardless of where it comes from, as argued in Chapter 5) - is incompatible with innovation.

Managerial opportunism normally incorporates arbitrary measures in the selection of managers, or in making them scapegoats out of people to cope with demands that cannot be fulfilled due constraints within organisations, product lifecycles, etc.), which may eventually justify their removal of management (Pfeffer & Salancik, 1978). Since managerial performance and accountability are subject to discretionary assessment, this author's reflection is that the innovation manager should address how innovation is going to be assessed upfront and agree on how accountability is going to be monitored from day one. In the cases studied we have seen that it is difficult to account for results in innovation, a look at how this can be done is an important issue that merits further research.

Nevertheless, in the context of the cases studied, we have observed how management at the institutional level focused on the coherence of the firm to ensure the success of their innovations. This is why IBM Barcelona Lab searched for active engagement and co-operation not only from banks (external firms in terms of legal boundaries), but also from the other Labs in the company, since "they did not want to substitute existing IT investments; they wanted to complement and protect IT assets," as one manager noted. Moreover, the IBM Barcelona Lab pioneered a new business solution, leveraging software development that was highly aligned with IBM's new strategy. According to our research, this new venture was culturally acceptable for the organisation, not only in terms of content, but also in terms of how it was driven. The IBM Barcelona Lab rapidly engaged in co-operative action with other parts of IBM: other IBM Labs, sales departments, consultancy units and other internal divisions. The issue they faced here was how to combine hierarchy and priorities with power distribution, something they achieved by establishing clear references, accountability mechanisms and rules of the game.

7.3.3.- Managing Transitions: linking the old and the new

In any change process there is organisational loss. This concept has been nicknamed as a "chasm" in managerial and academic literature. We prefer to call things by their real name, since imprecision as regards concept names, often leads to imprecision in terms of the concepts themselves. The IBM Barcelona Lab case is a good example of organisational loss: (1) in 1994, it lost its product and was forced to transfer it to the UK; and (2) the Lab's staff was reduced from 70 people in 1994 to approximately 15 in 1997. It was an important loss of community, as shown in the following figure on Lab staffing (see data for 1995-1999).



Figure 7.2: Change in Number of IBM Barcelona Lab Staff

This is how the Lab Director appointed in 1994 remembered this situation:

"In 1994 IBM was restructured as a result of the crisis that had affected the company. So the company decided to consolidate all our small laboratory developments – when we say small, a laboratory with less than a hundred people was considered small – and to merge them into larger laboratories. In our case this meant Hursley in the UK. So in 1994 we were told we were moving. At the beginning, the reaction, the human reaction, was "why?" Everyone was trying to work out the macroeconomic reasons that might have led IBM to take the decision, thinking about their economic situation, their corporate strategic position, etc., people wanted to understand these reasons and to share and communicate them appropriately. IN the end, we had to transfer all our lab's existing knowledge as professionally and appropriately as we could.

You treat the product like one of your children, don't you? There's a series of almost personal and emotional bonds. In the beginning (and I'm not going to try to kid you on this) the reaction of some people in the beginning was to say: "Right, but why? What's the reason for

this? And, who are going to be the product's new parents and now what's going to happen with the laboratory?"

Left alone, with no clear support to cling onto and no hand-holding, a transition can be considered seen as a lack of commitment by the people who stand to lose the product or the activity. As discussed, the processes of disinvestment (or lockout) looks very different if we consider it from the HQ's perspective and commitment is also required to proceed with lock-outs due to its irreversible nature. The Lab had the space it needed to understand the loss of the product, but it also had the capacity to re-invent itself with another type of activity.

As mentioned previously, the same year that Multiasistencia started its transformation (2000), its COO and CIO (who had held the positions since 1989) left the company with a small group of followers to create a new company to compete directly against Multiasistencia. They not only took knowledge with them but also a "copy" of the software developed at Multiasistencia to start up their new venture and run its operations. Multiasistencia sued directors and the new firm, but, after several years, the courts were still unable to determine whether the software could be considered as having been stolen, since defining the intellectual property rights for this kind of intangible asset was very difficult according to the management interviewed. Nevertheless, Multiasistencia's management soon realised that this painful situation had essentially only served to further drain the firm's resources and energy while also constituting a major distraction.

"Multiasistencia's top management decided to focus on the firm's core business and reinforce its relationships with customers and trade professionals. In short, they decided to put Multiasistencia's business back on track and leave the litigation to the lawyers, not considering it a top priority," according to one manager.

As we have seen, other firms entered this newly created market, increasing competition. In of our both cases, management set a path and stuck to a commitment to confront feelings of loss and the associated potential conflict. One IBM Lab manager synthesised this situation as follows, "After losing the product, we asked: What's the next step? Or, where are we headed? What are we going to do? And then they suddenly hand you a blank piece of paper, freedom to think and freedom to design your own future".

From this short description of the situation we can identify the well-known natural phases of loss (see Bowlby, 1985), starting with denial, moving to anger (people leaving, some with products with which to form competition), through to negotiation and acceptance (explaining why a product is going to be transferred or why Multiasistencia should take advantage of the internet) and finally to the creative design of a new future.

In this kind of situation being in the game is more costly than rewarding. From what we have observed, the design of a new future starts by engaging creativity, confirming existing theories (Kets de Vries, 2009) and opening spaces for communication and sharing, as well as letting people elaborate on this process. As one IBM Barcelona Lab employee recognised,

"To be able to think as a team; to be able to tackle the situation and say: "OK, *now we are masters of our own destiny and we can focus on where we think we can contribute most as a team."*

Innovation literature recognises the important psychological investments that people deploy. Here it is vital, as observed in the IBM and Multiasistencia cases, that management is successfully chooses to commit to a new path, re-designing boundaries or actions, and not (as can often be the case) responding hostilely, creating potentially negative relationships. As we have argued, commitment is established to ensure autonomy, not create a slave of visceral passions, which, as we have seen, are a part of business life. Finally, we would like to emphasize the importance of developing an integrative leadership role, that is, the capacity to link the old and the new, and to bring everyone to a new future.

7.4.- Organisational Capability for Co-Creation

We have argued that leadership expresses its power through its capacity to attract and influence others (as a centripetal force) and to spread ideas, thus dynamically expanding boundaries. We have also shown how innovation is a source of conflict and how this conflict can lead to a great risk of isolation for leaders. We turn now to an examination of how to set commitments as the organisational capacity to set boundaries, focussing on: (1) communication skills; (2) managing strategic relationships; (3) managing mutual adaptations; (4) multi-disciplinary teams and (5) recognition.

7.4.1. Communication

Communication is the way in which solutions which can contain conflict are cocreated. Indeed, the good thing about conflict is that it revolves around existing asymmetries and gives the leader a chance to solve them. Power is needed to change the rules and set new boundaries. A classic argument is that authenticity, empathy and assertiveness are essential to deal with this conflict. These traits help us to understand others' points of view, without necessarily sharing them. As one Multiasistencia manager indicated,

"The communication policy of sharing our objectives with them (Stakeholders) was an absolutely key factor, since there was a relationship of sharing the objectives and of *growing together.*"

Or, as indicated by one of the members of Multiasistencia's professional network, "We learned to share information and work in a more transparent way." Transparency engenders trust. This highlights the importance of monitoring, selecting sources of information and commitment. As one IBM Barcelona Lab manager reported,

"I brought in a totally open laboratory layout, so there were *no* "*boxes", to* encourage creativity. There were no offices. And why? Because I was extremely concerned about encouraging dialogue, creating a challenge between people, forcing them to talk to each other."

Dialogue is the major tool for leading team and organisational learning. Dialogue is a free flowing conversation between people, allowing them to discover insights that they could not attain individually (Bohm, 1965). As one IBM Barcelona Lab manager commented,

"Dialogue is a key part of our Mediterranean culture, but it was an intellectual discussion, where my solution did not necessarily have to be the best one. When my solution was presented to collaborators as a whole, the team members, discussed it to death, reviewed it and substantially improved it."

According to Bohm (1965), "Dialogue is a special kind of collaborative conversation, quite distinct from discussion, which is primarily competitive." Regular meetings are good settings to use this technique because: (1) they serve as a source for group decision-making and turn debates into intellectual efforts, not personal confrontations; (2) they represent an exploratory activity, to discover insights which are not within the reach of the workers individually; (3) they are a public instrument by which to recognise contributions (asking somebody to prepare a presentation, for example), while also representing a powerful tool to encourage, influence and reward people; and (4) they allow us to understand people's behaviour and interpersonal relationships.

Above all, these meetings give the message that people are not alone during the turbulent times brought about by innovations. They allow the appropriate level of conflict and different perspectives to be attained, and avoid social processes such as conformity or "groupthink" which can lead to reality avoidance. Trying to establish unilateral control over situations hinders any attempt to, first, find out about conflicts and, second, to try to resolve them through the use of dialogue, since potential conflicts are kept hidden (Argyris 1986; 1990; 1994; Argyris & Schön, 1996). Instead of searching for the roots of discrepancies, wheedling of unilateral control may lead organisations to enter into latent conflict and

encourage the empowerment of leaders who have a tendency to confront and cause conflicts. Gaps, conflicts, and contradictions are living things, and conflict will not simply go away if we try to sweep it under the carpet.

7.4.2. Managing Strategic Relationships

Managing relational dynamics means knowing how to manage the relational lifecycle of co-creation. It implies knowing how to start relationships, how to manage them to ensure equity and peer-to-peer dynamics, and how to terminate them (or leave them in a latent state). It means that priorities are based on strategic relationships (not only arms-length ones), including: (1) realistic issues and realistic commitments; (2) focusing on projects aligned with a defined strategy (for example, as with the IBM Barcelona Lab and IBM as Corporation); (3) searching for complementarities, that is, where an exchange of knowledge and complementary resources might be; and (4) both parties engaging in real commitments on economic and legal levels. The IBM Barcelona Lab Director described what relationship management was about in the following way:

"You really are successful when you apply talent, technology and know-how to *real problems*. And, as our background and our 'baggage' was in banking, we set ourselves a challenge: getting on borad the banks *that were more in line with the strategy* that we wanted to follow as IBM and as a team within IBM.

We were also looking for a leitmotiv, a strategy that could attract talent and hold on to that talent. We had and attract *talent from outside as well as talent from inside* the company. To do that, you need to believe in what you're doing: it's necessary to have a passion for what you're working on.

The first initiative that we started up was to choose a series of key projects where we knew that the clients wanted to innovate; where we knew that there were innovation niches waiting for us to come up with a solution for; and where we could put key people from the laboratory team in those clients, with a marked leadership task: to capture each of the clients' existing innovation and turn it into reality."

This approach defines network behaviour, a style or a "leitmotiv." It is like a piece of music that sets the rhythm, a pace which, paraphrasing White (2008), is a style that defines an identity and which leads to a hierarchy and a way to attract others.

Another important aspect is the emphasis on emergent associations of autonomous actors from inside and outside (the boundaries here being the legal ones), in order to create increasing returns based on the management of relationships. The IBM Barcelona Lab and Multiasistencia developed these projects based on networked associations to engage in co-creative activities with their corporate clients. In the Multiasistencia case, the management forged trust by being transparent and not hiding the problems the company had in 2000. For example, Multiasistencia worked with a major European bank to define a new application to inform end customers in real time every time they called the bank with a claim. The project was developed in collaboration between the two companies during 2006 and was put into production at the end of that year. One executive from this bank declared, "We have to add transparency to our service. Information is the key to increased loyalty."

Finally, relationships end, and this has to be managed as a loss, as previously discussed. Reasons for ending a relationship may be that results do not meet the others' expectations or that one or more of the agents in the relationship decides that the expected objectives will never be achieved. Multiasistencia, for example, lost some customers during its change process from 2000 to 2003. For some, this "break-up" lasted two years, since the IT and process lock-in were extremely significant, and complete disconnection was required. In other cases, the relationship ended because the costs of maintaining it were greater than its benefits, or because the relationship was not successful. In many cases, relationships end simply because the project and its purpose end. As seen in the IBM Barcelona Lab case, if the relationship has been beneficial, it may be kept latent, making it very easy to restart again in the future. In fact, many of the

Lab's competencies were knowledge-based and could be found in latent relationships with satisfied customers.

7.4.3. Mutual Adaptation for Co-creation

We have argued that networks, as loosely-coupled structures, are optimal configurations for innovation since they are formed by autonomous actors that can commit and adapt dynamically to each other to keep the structure alive, evolving and organising for the future. This is a different perspective from the so-called *Conversion Theory* described in Chapter 3. The term "conversion" assumes that a conflict is solved by accepting a new idea. This may be true in some diffusion mechanisms. However, with co-creation, we propose that adoption is an active process of co-construction which, at the same time, is a process of diffusion embracing two types of adaptations: (1) technical adaptations in product features or the production process; and (2) mutual adaptations that bind companies together, often in a direct physical sense, reflecting a mutual commitment which, at the same, is limited in time and empowers the companies.

In the IBM case we stress how management took care of its people and its installed base in order to connect the new technologies (and organisations) with future ones. As one manager justified:

"We weren't trying to do away with anybody's job. Instead, we wanted to use and maximize all the existing technology: In other words, everything that would become the mainframe. And what we did was to take advantage of personal computers and hook them up to the best communication mechanisms that existed at that time, and which still exist in our clients' applications to this day."

During projects, activities were focused to have a more complete vision of the interaction. This "bi-directional adaptation" - as several managers alluded to in both cases - respects cultural and market diversity and differences, and also respects the diverse business requirements that depend on different market conditions and customer needs. This recognises pluralism; in other words, power is distributed in networks. IBM Lab staff not only sub-contracted work with

Independent Software Vendors (ISVs), but also "hired" project directors from the banks to work in the Lab. One manager added that IBM top management was reluctant, "But how we are going to have people - clients - in our innovation and development centres?" And, without a doubt, that has been one of the wagers that have given the best results".

Multiasistencia staff was also rotating permanently with major banks, and developing strong relationships with top management, becoming a critical factor to support the development of the Business Network Operating System (BNOS), as well other co-creative ventures.

Moreover, IBM Lab staff sub-contracted work with ISVs to manage network capacity, depending on the work load. Since orchestrators lead formal and informal relationships, including people "outside" the organisations' legal boundaries, orchestrators normally command "external" and critical resources for their legally defined firms such as key relationships with key customers and key associations, thus leveraging power with the control over these resources.

This emergent type of organising is based on co-creative relationships that are based on exerting autonomy. In the IBM case, management asked the employees to change roles and rotate positions to avoid them feeling too accommodated and to ensure their continued search for new opportunities, a process which empower people.

One of the Lab members stated,

"This results in plans being changed. People have to be dynamic, flexible, and know how to adapt to these changes. I would say that this is a little difficult at first; basically, professionals prefer to know very clearly what is expected of them, exactly when and how. And as soon as they get used to that dynamism, to that flexibility, they take it as a routine part of their work." By the same token, in the Multiasistencia case, Customer Service Representatives (CSRs) were asked to change their role from "information searchers" to "process owners", to control processes proactively and interact with different actors in a learning process to manage exceptions. Key Account managers searched for more commercial opportunities by developing a better understanding of bank processes. Trade professionals were encouraged to become small businessmen, more autonomous and committed to Multiaisstencia. And IT staff members were engaged in co-creative software projects with banks. This is similar to the IBM Barcelona Lab case where staff changed from project management to system analysis and business consultancy positions.

The capacity to rotate from software development to project management and to even business development was considered a part of the Lab's tasks, thus "avoiding repetitive or routine activities," according to one manager. Leadership, according to the Lab Director, was:

"A matter of capturing the existing innovation needs in each of the clients, to turn this into reality and co-create a solution with them. As a result, we had key people in key projects."

Experimenting introduces a flow of ongoing ideas and inventions, a combination of accents and changing harmonic patterns that interweave the structure of relationships and networks. In all cases, problem-solving requires: the tacit accumulation of experience, "actionable knowledge," the use of explicit knowledge, interaction with other people, information systems, and creativity which leads to performance. For the IBM Lab, their hybrid programming technique allowed them to produce software releases and reduce the time-tomarket to only 6 months. As one manager pointed out,

"What we wanted was for the client to be involved in the whole development processes, from the beginning right through to the end. This approach requires trust and transparency since you can't hide any mistakes, but the result is very beneficial since we are all in the same boat." IBM Barcelona Lab engaged in many formal and informal relationships with other IBM Labs as well, reaching commitments with "internal" units. As one Lab member related,

"We had the experience to support multi-platform and operating systems. In 1992, three months before an IBM laboratory in the USA launched the new version of a new operating system, they discovered that it needed 15 or 20 more diskettes than planned to hold the entire operating system. This would have caused project costs to soar and delayed the launch of the new version that was to adapt production to the new conditions. During an informal conversation, one engineer at the Barcelona Lab offered his colleague in the US a compression algorithm he had been working on in his free time. A few days later, initial trials on the algorithm were successfully held. Until 2002, this solution remained IBM's standard software compression algorithm."

7.4.4.- Multi-disciplinary Teams

An interesting finding is group identification beyond legal boundaries in processes of co-creation. One of the IBM managers commented, "As a final result, we have clients who consider themselves *one of us.*" This suggests that trust among team members considers peers to be "equals" in terms of hierarchy, even reaching "members" beyond formal legal boundaries, something which represents an advantage, but also raises many questions in terms of co-creating products and services. For example, who owns the intellectual property rights? If a company can use technology to link or increase lock-in of network actors, could it, for example, also include better ideas for new service development? Is this a process that better fits strategies and resources and their transformation within all the network actors? This discussion underscores the importance of managing secrecy, non-disclosure agreements and incentives and their evolution in cocreation efforts, and how they are different for each player in the network. These questions could be the subject of further lines of research.

7.4.5.- Recognition

Recognition is the cornerstone of any leadership process (Anderson, 1983). And innovation, as seen, questions people's previous achievements. Co-creation recognises actors in the network as creators, thus making recognition a cornerstone of network leadership abilities, allowing the leader to attract more talented people, more projects (business), and more actors, reinforcing the orchestrator's centrality as a reference for innovation.

Managerial recognition of others' contributions is a *sine qua non* condition. We stress here that co-creative networks do not only rely on managerial recognition, but also on empowering people, based on the idea that giving autonomy allows them to develop, as well as to learn to work with a view to achieving one's own as free individuals (Rand, 1967). Exerting autonomy confronts the risks of immature organisations in which its members are prone to identify managers as the only ones who exert power (Kets de Vries, 2003). In contrast, pluralist organisations distribute power in a similar way to democratic political regimes. The issue here is the successful combination hierarchy and priorities with power distribution: with clear references, accountability and rules of the game. As one top manager in the IBM Barcelona Lab added, "Being part of the Lab gave people a *sense of pride* and recognition." This same sense of pride is observed among the trade professionals who work with Multiasistencia, "Being part of the Part of the Multiasistencia network is a way to express our quality," as one trade professional commented.

Formal managerial recognition must also be applied. For example in the IBM Barcelona Lab case, the engineer who developed the compression algorithm became a member of the IBM Academy, enjoying great prestige and recognition. Similarly, the President of Multiasistencia explained the company's key success factors as follows:

"I believe that the success factor was basically the attitude of the management team; if we wanted to criticize something we would say that it was a little adventurous, a little daring, a little bold, a little reflective. But if we wanted to praise it or speak highly of something, we would say that it was brave. We had an extremely determined attitude to confront this permanent challenge of growth, with a great spirit of service, a great vocation to do things well, a very enthusiastic team, with many hours of work."

We have argued that leaders express power through their capacity to attract and influence others (as a centripetal force) and to give autonomy on a peer-to-peer basis, encouraging other network actors to spread the new way of doing business. The idea is that instead of having been *converted* to a new doctrine, actors in the network feel proud to contribute and co-create in many ways: experimenters (system engineers and programmers), cross fertilisers (multi-disciplinary knowledge from other areas), leaders in banks, for example, who promote new ideas, architects and designers, caregivers and storytellers. Co-creation is not, therefore, a solely generative activity; it also works as a diffusion mechanism since co-creation expands the network beyond its legal boundaries, as shown in Figure 7.3. Orchestrated Networks are based on this sense of pride, respect for individuality and autonomy, and team playing, are generative because they produce novelty, and behave like structures of diffusion for this new *doctrine* based on caring for each other, co-creating activities and commitment.





Figure 7.3: Orchestrated Networks: Innovation Generation and Diffusion

7.5. Economic Capacity

According to the research presented, the performance of innovation can be described using the following findings:

- Reduction in time-to-market: for example, software is released every 6 months at the IBM Barcelona Lab, instead of 2 years, or IBM's taking advantage of the PC and internet as early detectors; similarly, Multiasistencia took 1 year to co-create the BNOS which was of greater quality and boasted more features, as opposed to the previous in-house version which took 2 years;
- Increased customer satisfaction and loyalty: for example, according to one IBM executive, "For us, the subject of customer satisfaction was absolutely crucial: customer satisfaction with the solutions that we started up. And I can tell you that we have improved exponentially";
- 3. An increased level of control over innovation through mutual adaptation and integration projects: (1) on a technological level, taking advantage of previous investments and the existing technological base, and (2) on an organisational level regarding inter-organisational processes of software production and service management (both IBM and Multiasistencia), thus reducing transactional costs;
- 4. Reduction in IT transfer costs: a) initiating the innovation process through co-creation; b) identifying industry leaders who are willing to innovate (banks in both cases, in addition to trade professionals in the Multiasistencia case), and thus leverage the innovation diffusion of emergent leaders; and c) reducing co-ordination costs by giving autonomy and co-creative activities based on trust as argued in Chapter 5;
- Increases in productivity: for example, a 50% productivity increase (2000-2003) in Multiasistencia and new software development in the IBM Lab: LAN-DP and WBCC.

We then argued that resilience is the innovation outcome in network changes. This is called resilience at the structural level, examining how much change the network has absorbed to reach a new equilibrium. Our research shows (1) members of the network co-creating with complementary equals; (2) some weak members trying to gain power in order to balance asymmetries, including the focal firm; (3) amicable (or otherwise) separations (or lock-outs) as seen in the cases considered. The degree of resilience can be synthesized by a phase, a motto, which in the case of the IBM Barcelona Lab was re-invention.

7.6.- Proposition: Orchestration as Leading Leaders

Proposition 7: Orchestrators lead leaders in the Network by making commitments and exerting peer-to-peer relationships.

There is never a single leader in networks, as suggested by Merton's notion of ego-centric networks. There are many egos, and it is obvious that, what one person sees as the centre in the network is not the same as that perceived by the other actors. Managing business generation and trust were critical to the management of increased growth in the cases studied.

According to this researcher's observations, perhaps the most salient characteristic of orchestrators is their capability to lead and to be led when circumstances require. Orchestrators lead leaders, propose actions and engage other leaders by using power (of a variety of types ranging from inspiration, influence, and formal authority – where the hierarchy is there to support this they may help others to make sense of situations and/or use coercion when needed). Leaders are implicitly elected by teams and serve as references for innovation, exerting this role of centrality.

Leaders balance dependencies by giving autonomy, empowering people, improving work content and opening new sources of innovation. If we agree on the fact that people tend to do to others what management does to them (Milgram, 2005), then commitment will get commitment in return. The IBM Barcelona Lab was based on autonomy, staffing "key people in key projects" these people (such as technical architects or sales people) were both from IBM and from banks, and dealt with bank and IBM leaders. In the Lab, it was quite the norm for a professional working in development to move on to the project group at the end of the development phase, later going to the client to implement the project based on the product that he or she was personally involved in the development of. Another interesting finding is group identification beyond organisational legal boundaries. In Multiasistencia, autonomy was clearly given to banks (providing them with information to keep track of claims) to trade professionals and to the CSRs that managed the entire process.

As mentioned above, IBM Barcelona Lab orchestrated its staff in different projects where they changed roles and functions, Independent Software Vendors (ISVs) but also hired project directors from the banks. Similarly, Multiasistencia staff was permanently rotating through major banks and developing strong relationships with top management. Since orchestrators lead formal and informal relationships, including those "outside" their organisations' legal boundaries, orchestrators normally command "external" and critical resources to their partner firms, such as key relationships with key customers and key associations, thus leveraging power with control over these resources.

Chapter 8: Conclusions: Perspectives on Innovation and Leadership

This thesis proposes the concept of *Orchestrating* Network Behaviour for Innovation as a managerial function that stands as a proposition and a theoretical hypothesis. Methodologically, it is the fruit of abduction, that is: (1) a creative synthesis and, at the same time, (2) a tool to analyse objective reality. This managerial function generates a path between the current state of affairs and structures and future ones, through the management of network structural dynamics and digital platforms, understood as the infrastructure for innovation. We have also examined network structural dynamics in two longitudinal cases that have allowed this researcher to review 25 years of these companies' evolution.

Orchestrating is a managerial function to set a *new path between the present* and the future, managing a process of ongoing equilibrium between actors' benefits and costs which can vary along the way. It is stressed here that setting a path is "changing the rules" while keeping the game alive - essentially, the principle of efficiency. This function is based on Strategic Leadership, which starts by creating a change in perspective, a new business venture that attracts people and resources. Leaders create networks naturally, but we argue that they require strategic management capacities in order to cope with strategic factors, dependencies and contextual variables. These capacities are required to manage real contextual variables. With this Orchestration function, we explore how management can open up new spaces for creativity to change the existing rules of the game, thus changing structures (not breaking them), as the function is an evolutionary process - a journey or a path - where efficiency is expressed by avoiding disruptions. As a managerial function, it is based on the power to: on the one hand, open new sources of innovation and, on the other, set commitments which shape network boundaries.

Commitment is a principle of demarcation: orchestrators exert power through commitments to control network boundaries, thus affecting the roles and

171

relationship dynamics which determine resolving structural dynamics. A complex structure (based on autonomy and voluntary association) requires boundaries to survive. New opportunities come from "multi-connected" and "multi-disciplinary" networks.

We have stressed this issue particularly in suboptimal regimes, therefore proposing resilience. In this respect, management is about dynamically regulating boundaries through centripetal and centrifugal forces in accordance with particular objectives and circumstances which can also shape roles and relationships. Therefore, Orchestrating Innovation is an ongoing process to change network structures in order to manage the players' autonomy and peer-to-peer relationships for co-creation, sharing investments and engendering trust, equity, holding conflict, and organising forward .

First, we emphasise that market creation is dependent on orchestrating, that is, autonomy, creatvity, commitments and trust. Second, we have already argued how complex structures need boundaries to survive and, by dynamically managing boundaries, they set paths for innovation. These structures lie between chaotic enviroments (with no autonomy), governed by an "invisible hand" or "too much of a vissible hands on approach". Third, we suggest that orchestrating innovation, although perhaps considered a meso-level and transitional phenomenon, highlights the deeply rooted natural behaviour for innovation through the investment of resources to create something new in contact with the uncontrolled flows ("ups" and "downs") of economic cycles. Moreover, we are not defending anything related to a strategic apex or "central planners" to control economic cycles. We stress, instead, how voluntatry associations of free people, orchestrated to build a new path, can serve as an integrative force capable of avoiding disruptions (during economic booms or depressions) by placing economic activity in real patterns, in other words, it is about knowing how to change the rules, not lead disruptions.

As shown in Figure 8.1, Orchestration is a managerial function with three proposed dimensions: Strategic Leadership, Structural Network Dynamics and the Management of Digital Platforms.



Figure 8.1: Orchestrating Network Behaviour for Innovation

In this empirical research, we have found that digital platforms: (1) facilitate the development of new business models and affect markets through dominant designs or references; (2) enhance co-production architectures; and (3) use programmed behaviour naturally linked to boundary management.

Leaders experience tension between authenticity and exploiting their own potential by stressing managerial effectiveness in terms of: (1) organisational capacities; (2) conflict management; and (3) financial results. The above discussion reflects the position held in his thesis. We turn now to a final discussion of the implications and contributions of this research, as well as the limitations and implications at different levels for future research.

Orchestrating and Power

By examining Structural Network Dynamics from the perspective of power, we have found the following characteristics in focal firms (or Orchestrators) regarding power in the following dimensions: centrality, control over resources, and dynamic capabilities. To date, the literature has focused on roles and, in part, on the management of resources.

First, we put forward centrality as the *Reference for Innovation*, emphasising that the locus of innovation is the network itself. In other words, the focal firm's role is that of reference to generate novelty by *opening new sources* of innovation and fomenting communication between actors to facilitate co-creative activities.

Second, power is about controlling boundaries defined as sets of commitments. This control is exerted by managing autonomy and cohesion. This stresses two main variables: autonomy and the predominant relational model among network actors (whether hierarchical or peer-to-peer). Moreover, we have shown that innovation outcomes are maximised when the relational schema is peer-to-peer, thus leading to Generative Networks and processes of Market/Factor creation. We have stressed this issue particularly in suboptimal regimes, therefore proposing resilience as another key characteristic. Finally, orchestrating is about dynamic capacities which we have analysed using the metaphor of centripetal and centrifugal forces. Orchestrating recognises that the focal firm dynamically regulates boundaries or constraints (or the rules of the game) through commitments in accordance with the objectives and circumstances which shape roles and relationships. Commitments can be understood as dynamic agreements on direction setting, network behaviour and a way of holding conflict in structural changes. Orchestrating through commitments emphasises managing autonomy and recognising the actors' capacities to self-organise and control themselves, that is, making Orchestrating for Innovation a focal characteristic of the network.

Generative Networks

Generative networks are expressions of optimality as a result of Orchestrating Network Behaviour for Innovation, as has been shown by empirical research described in chapter 6. Orchestrating is based on commitments, balancing centripetal and centrifugal forces, engendering trust, a symmetric (peer to peer) model of relationships and co-creation.

The orchestrator exerts power, as argued, with the following actions: (1) the focal firm carries out different roles, but that of serving as the reference for innovation is predominant; (2) power is exerted through dynamic boundary control and setting constraints, thus establishing commitments, exerting autonomy and avoiding the extremely costly option of one-to-one supervision; and (3) facilitating co-creation, the emergence of dominant designs and the creation of market factors through positive value returns and network externalities.

The orchestrator exerts autonomy by opening spaces for new sources of innovation, boosting communication among actors, and through controlling digital platforms that provide an infrastructure for innovation, thereby giving *more autonomy to actors* by letting them build new applications, re-using and integrating them with existing applications and components to build more integrative solutions (in the IBM case) and new services (in Multiasistencia case).
Managing autonomy and commitments recognises that innovation can unfold anywhere and that orchestrating implies linking and communicating multiconnected resources or, applying legal boundaries, "internal" and "external" resources (classically called boundary spanning).

Autonomy is expressed through the proposal of new projects by composing software projects considered as the development of re-usable components, in other words, a re-combination of existing resources to expand the resource base, as dynamic capacities suggest. Additionally, this research has shown that project teams synthesise existing knowledge with new information, making it into interconnecting software modules which became the foundation for the modular architecture of digital platforms. Given the common problems and broad applicability of these processes, both focal firms quickly found (1) economies of scale and externalities that helped apply their digital platforms to other sectors of the economy with some additional modifications (for market creation), and (2) economies of scope as they became "innovation factories" for complex, processintensive information services, thus balancing greater scope or breadth with greater focus and specialisation.

Orchestrating is a network capability, not only a question of the focal firm. With this approach, the network is the locus of innovation and Orchestrating is about opening and managing new sources: (1) network interactions among actors; (2) spaces for co-creation and generation of new modules/applications/services, and (3) the replication of existing models in new markets, industries and areas, as seen. The focal firms detected significant inefficiencies within the banking sector's business processes due to their profound knowledge of the business and administrative processes in the industry.

Orchestrating Network Behaviour for Innovation finds its *optimum configuration* when the predominant relational model is peer-to-peer and co-creative. Managing autonomy and commitments recognises that innovation can come about anywhere and that orchestrating implies linking and communicating "internal" and "external" resources and boundary spanning or bridging activities.

We have presented evidence that centripetal forces can be ideological, but that they hide asymmetries in deep structures (such as opportunism and other human passions) which are normally dismissed in organisational studies, but are critical to organizing for innovation. Scientific management is about repression, not containing conflict or setting boundaries for action. Creative destruction, in turn, ignores autonomy, trust and commitments, and assumes a reactive attitude based only on competitive elements, thus dismissing cooperative and co-creative actions. All these are based on denying autonomy, thus ignoring complexity and coming close to representing chaos.

Orchestrating Innovation contributes to the notion of loosely-coupled structures that lie between both configurations of chaos: tight structures and unstructured configurations (sometimes, with all preventions: markets and hierarchies). Both extremes can be chaotic because they restrict freedom and autonomy. Markets are conceptualized as being only market driven and opportunistic; bureaucracies rely on "scientific management" that normally hides profound asymmetries (that sooner or later unfold), cultivates arbitrary ways of organising creativity and autonomy, and that has the capacity to set commitments, that is, force people to "anomic" states.

By doing so, they condemn extremes to deterministic trajectories, almost reducing the levels of freedom to zero (in terms of the multi-connectivity and multi-disciplinary traits found in network behaviour), and drastically reducing the capacity to set paths of innovation. These chaotic extremes have a problem with the future (and, indeed, with innovation), since markets are unpredictable and bureaucracies are prone to *repetitive behaviour*, eventually producing outdated products and services for non-existing markets and jeopardising the firm's performance and its very existence in the future.

This notion is depicted in Figure 8.2. In short, networks are optimum configurations (although through processes using different steps; as discussed, this is not a "final" optimum, but a more modest process-based approach). Thus it is in between these two extremes that we find chaotic behaviour, notions

consistent with theories of complexity and psychodynamics (see Stacey, 2003), as the findings of this study have shown. We have seen how IBM almost faced extinction in the 90's, but we can also look at this from a different perspective. A small Lab in Barcelona that pioneered hybrid programming and solution driven business, today represents 50% of IBM revenues. The same can be said of Multiasistencia, a focal firm who had to confront deep changes from 2000 to 2003. In both cases, they pioneered what, in the end, became a very complex process of metamorphosis, changing the rules of the game and the resource base. Both focal firms brought networks to a new future.

Thus, Orchestrating Network Behaviour is a network's capacity to set a path, employing an inner capacity called *innovativeness* (connected creativities, and commitments) by managing dynamic equilibrium through (1) dynamically managing network boundaries and constraints, reinforcing the notion that no complex system can survive without boundaries; (2) unleashing the power of digital platforms as co-creative spaces, and (3) boosting multi-connectedness and multi-disciplinary teams. This thesis has shown how orchestrating stress (one of the roles of management) and providing illustrations of how to manage networks can be successful strategies for innovation *at the edge of chaos*.



Figure 8.2: Paths for Network Behaviour

Innovation and Digital Platforms: Market Factor Generation

This thesis has proposed that orchestrating can generate new markets, although this is a first step to which more research must be devoted. This concept suggests that orchestrating is one way to command, and that it can generate optimal configurations or loosely-coupled structures based on commitments through establishing strategic relationships and multi-connectivity. This approach suggests that Orchestrating leads to: (1) the generation and diffusion of innovation, and (2) combining divergent and convergent behaviour (although there may be some predominant patterns depending on the phase).

This approach also suggests that Orchestrating Networks for innovation is a meso-level phenomenon that is driven by management. This means that in the first steps when activities do not have enough scale, trust and commitments are essential to create communities. Once the super-additive games create a critical mass, the logic seems to be price driven and solely competitive, thus drastically reducing the degrees of autonomy needed and dismissing the role of management for market/factor creation. Market logic tends to dissolve existing communities and increase competition and opportunism, thus reducing autonomy or, in other words, adding more constraints, which can reduce the ability to innovate.

In the Multiasistencia case, we explore the implementation of a specific Business Network Operating System (BNOS) which played a fundamental role in automatically regulating boundaries and ensuring expected behaviour. It achieved this through deterrent-based control systems to build trust, and increasing efficiency using information tangibility and transparency. In the IBM Barcelona Lab, digital platforms allowed the Lab to rapidly take advantage of the internet to develop multi-channel solutions for banks world-wide, building barriers to competition in this process of co-creation with banks.

In Figure 8.3 below, we present the three levels of commitment (1) digitally enabled; (2) co-creative and (3) irreversible lead by focal firm. Irreversible commitments (from point "B" to "C") lead to dominant designs and dominant business logic, although the path is based on a set of experiments, a set of steps until the new business logic is right. This process may also attract other actors that can incorporate new complements to the network and to the digital platform due to increasing returns and externalities or super-positive sum games. Although initial centripetal forces are grounded on the business opportunity, once the business grows, the attraction of natural forces of leadership and commitments can hide asymmetries due to opportunism.

In fact, digital platforms can be considered as a resource that can generate trust: (1) as a deterrent (for managing Service Level Agreements); (2) as a framing infrastructure for co-creating by standardize some processes; (3) as a knowledge synthesis of the inter-organisational learning process; and (4) as a network resource that can generate trust in the network. In this way, digital platforms may help to create new opportunities to collaborate and facilitate the recombination of this knowledge in novel ways. This can occur in the same or new relationships and co-ordination mechanisms, thereby fomenting business growth, change and further innovation. We have highlighted information systems and the digital platform, in particular, as proxies for commitments and also as a communications system and a creative space. Creativity in social settings may be part inspirational, but it also requires perception and communication in order to become a structural process.

Orchestrating with digital platforms recognises automatic boundary setting for network behaviour (as the Multiasistencia case shows). In this case, the structural pattern or behaviour was driven by an information system setting a specific set of constraints to limit degrees of freedom. The second role information systems play is that of a standard and standardisation platform for the development of new software modules for the specific function of front-office and back-office integration in the banking system. Although business functions are different, the social and structural functions are similar: to set boundaries for human activities and set standards for behaviour. We represent this as a horizontal line that limits the "levels of freedom" in our diagram of cycles, with the hypothesis that "ups" and "downs" can be held at this level; in other words, boundaries of action can be set.

After analysing "up" dynamics, we now discuss "downs" and sub-optimal regimes. Once a new dominant design emerges, the competition (both external and internal) that aspires to this position will focus on prices to take advantage of the new design's weaknesses and offer the same product at better prices, focusing on large-scale production. Orchestrators generally focus on the design and deployment of specialised assets, that is, the generation of new modules through co-creative activities and more specialised assets which are more difficult to copy. Examples of this can be clearly seen during the more mature phases within the IBM Barcelona Lab and Multiasistencia cases.

The following figure highlights, in the ascending lines, centripetal forces predominate, extending limits and generating a positive sum game. Here the orchestrator generates value for the network actors through maintaining control over the technological strategic resource in order to absorb a greater part of the value created. It is here that the commitments are expressed in the form of exerting autonomy and co-creation.

Once an optimum is achieved, the scale factor reveals a fact which requires further research. The way down starts with the product/services' maturity, community dissolution and competition (external in markets and internal/external in global firms). It may be the case that everyone can (or knows how to) establish commitments at the individual level (Bowlby, 1980: 1985), and price and competition become the forces behind this co-ordination. With this set of constraints, factor generation strategies have to make way for the market by positioning strategies and investments in terms of differentiation in order to maintain the equilibrium (Wernerfelt, 1984). Managing from point "C" to "D" in the figure below requires controlling the risks associated with the dissolution of the co-creative community and competition. Obviously, each case will indicate how these variables are quantified.



Figure 8.3: A Dynamic View of Orchestrating

As argued, if the scale becomes too large, it seems that dynamics are superseded by market rules, competition and opportunism, not commitments. Transaction Cost Economics can predict this fact, and, according to this research, this is true in the absence of commitments as boundary criteria. If the criteria to mobilise resources are driven by price, then Transaction Cost Economics can predict that valuable resources will tend to be in-sourced by firms. In the IBM Barcelona Lab case, once a dominant design was released, the major risk faced by the Lab was the dissolution of the technological community due to internal competition (as a hypothesis) and internal policies favouring off-shoring.

However, we argue that networks and digital platforms can reach optimum states for innovation which engender the dynamics for market creation, through which actors can exchange ideas, and then projects and new services can shape the network's structural dynamics. Nevertheless, even when a leading design is achieved and management becomes hierarchical with the rules set by product/market activities, the way down to a new transition is always hidden. This leads to the choice of starting a new orchestrating venture to manage commitments and constraints in order to avoid disruptions, and leading network behaviour for innovation to create new factors and new opportunities, thus emphasising value regimes based on equity or "growing together," containing conflict and organising forward.

Network Leadership

Network Leadership is about changing the rules of the game, attracting people and emphasizing relational capacities and political mastery. Leaders are early detectors of innovation and conflict. Sources of conflict are similar to innovation: hidden asymmetries or new waves that require more than superficial changes, but changes in deep structures, shaping network behaviour.

Leaders contain (instead of repress) the emergence of innovation (and conflict) and thus commit to teams and objectives (irreversible in some cases). This

achieved by Network Leaders leading other leaders, who generate and disseminate innovation. Network leadership is about dynamically managing boundaries, developing innovation sources, enhancing communication and dialogue, and focussing on team performance and business objectives. Network leaders emphasize commitments, that is, they set the direction and monitor it. As a result, they make the network the locus of innovation by promoting strategic relationships for co-creation. In contrast with the idea of leadership as a disruption, we argue that leaders should take care of the organisation by linking the organisational memory, the present state of affairs and possible future one, that is, by linking the old with the new.

The first important factor in power is authenticity as an expression of full power over oneself. This is commitment or a commitment to oneself. Leaders also commit to their *crews*, referring back to the Ulysses myth. They naturally attract people, including people from inside and outside of *formal organisations*. Leaders naturally exert power by giving autonomy to invite people to work by themselves in order to engage in peer-to-peer relational dynamics that lead to co-creation. Orchestrating Innovation thus becomes a focal characteristic of Generative Networks which are based on opening new sources of innovation and combining them through structural communication processes. Orchestrating Generative Networks is about how to crystallize different ideas, knowledge and other resources to turn them into viable innovations.

Not all managers are leaders, and not all leaders occupy managerial positions. In both cases, management and leadership are expressions of power that affect how innovation unfolds in organisations. We have argued how important it is for leaders to understand and exert managerial capacities. Due to their natural gift for attracting people, leaders have a natural attribute for giving autonomy. Commitment is the force that keeps the network united and allows it to renew the resource base of firms and re-organise economic factors and activities to engage in novelty and encourage innovations, that is, ensure its long-term survival. We have also briefly explored the complex relationships between leaders and how this variable shapes innovation. Since organisations are made up of several leaders; the way they establish relationships with others inside and outside of their organisations is important, that is, how they establish networks of mutual influence. The question is how to determine a systematic method with which to understand *how leaders interact*, how they interact with followers, how they manage different types of relationships and different games at the same time, exchanging resources, and how they establish relationships between organisations. Orchestrating Network Behaviour for Innovation is a managerial function that explores this activity by placing "key people in key projects," emphasizing the role of network leadership and its managerial effectiveness in terms of communication, financial outcomes and organisational capacities to manage relationships to produce novelty and innovation.

In Table 8.1 below, the limits between perspectives are not clear-cut. Roles vary, but this table reflects the interpretation of this researcher. Network Leadership is about Governance and resilience, that is, the capacity to modify the level of control according to the circumstances. It also requires having the resourcefulness and flexibility to adapt to a changing situation and the ability to process painful and conflicting information. Moreover, we have argued how important it is to cultivate politics, alliances, networking and understanding of the games of others, looking at all these questions with distance and dealing with them in a positive frame of mind.

In the cases studied, we have proposed some additional perspectives that will require more work and fine-tuning in further research.

	Commanding resources	Market Factor Creation	Building Identity and Purpose	Managing Conflict	Generative Networks for Co-
					creation
Leadership Roles	Commander	Entrepreneur	Integrator/ Institutional	Mentors – Care: Holding and	Reference
or Different Ways to Attract		Recognition (in all phases)	Connecting the old with the new	Expanding	Capacities: organisational,
(Pull)		Agitators		Communication	economic (tangible) results and
				Resilience	recognition.
				Exert autonomy	Resilience and conflict
					management
Structural Change: Roles,	Boundary reminder	Boundary expansion	Boundary reminders	Centrifugal forces may prevail:	Boundaries controlled by peer-to-
Relationships and	Exert control over resources	(centripetal forces prevail)	Dynamic adaptations: equilibrium	competition and community	peer negotiations
Boundaries	(formal and informal):	Positive returns (risk of	between centripetal and centripetal	dissolution	Self-controlled autonomous actors
	recognition and sanctions;	asymmetries)	forces	Focusing on symmetries and	Co-creation
	technology; relationships	Network externalities		complementary factors among	
		Control over: (1) resources;		peers to co-create and make	
		(2) intellectual property; and		investments	
		(3) "complementors"			
Digital Platforms	Deterrent	Digital Platform as ideation	Strategic and structural resource	Deterrent	Component-based digital
	Standards			SLA	platforms
					Complementary digital assets
Human Passions	Maintenance of trust	Personal trust in the leader /	Maintenance of trust	Focus on trust generation	Interactive trust among peers
		venture (expectations)			Trust in the network
					Trust in the digital platforms

Table 8.1: Leadership Perspectives

Limitations

As with any research study, this project has some limitations that warrant attention. First, all research on networks suffers from the difficulty in delineating the unit of analysis, as many levels of analysis are intertwined in networks. If innovation is an expression of leadership and both appear everywhere in the organisation, how, then, does management deal with this phenomenon? How do leader-to-manager or leader-to-leader relationships influence the practical use of the innovation, avoid destructive conflicts and generate value for the organisation? A second limitation is that this research is based on two case studies, thus requiring validation with more empirical research. Achieving additional qualification and validation across larger samples is an obvious goal for a larger research agenda.

Other limitations are related to further questions regarding factor and market creation. Can orchestrating be found in organisations facing market growth? And if so at what level? How are dependencies and market rules combined? How do we link market rules and new ubiquitous digital systems to Porter's model? What are the key variables? Digital platforms reveal lock-in, clear boundaries, defined functions and clear connectivity protocols to ensure functionality. This approach contrasts with the argument of innovation networks, highlighting the capacity to exert autonomy in actors and thus generating lock-out. How do we address these issues?

More research is required in terms of dynamic capacities and the process of cooperative experimenting, exploring, and opening the network to new voices. This researcher sees several different issues here: What are the costs and limits of diversity and differentiation? We know that managing heterogeneous groups is difficult. Moreover, peer-to-peer networks are not exactly diverse. How can we combine both? How do we manage differences in absorptive capacities? How do these differences affect the recombination of resources for innovation? How can incentives dynamically affect the network's evolution? More research is needed on the "innovation manager" theme, accountability, since managerial performance and accountability are subject to discretionary assessment. How can you account for innovation results? In the cases studied, this was an important issue and merits further research.

While we believe that this contribution breaks new ground, we also recognise that future research should consider the co-ordination mechanisms that govern network behaviour for innovation and their differences with other units of analysis.

Final Remarks

The construct *Orchestrating Network Behaviour for Innovation* highlights the governance of different voices and co-creative processes, which work through setting boundaries and changing rules of the game according to circumstances, rather than exerting power to "destroy" people's creativity and natural capacities (or condemn people to anomic contexts). Therefore, networks are strategically orchestrated contexts where dynamic, inter-organisational relationships allow for collaborative action to innovate. Relationships are also complex phenomena that require different levels of analysis to understand their co-dependency mechanisms and how these affect deep structures and network behaviour. Finally, we have described the specificities in peer-to-peer production relationships as activities between peers or equals in a hierarchy, but heterogeneous in the recombination of complementary assets such as knowledge and technology. This combination generates trust for the orchestrator as a reference (or the interactive lighthouse of the network) and is interactive at the relationship level.

Now the ship is reaching harbour and has to be docked. Network Leaders, as in Ulysses' journey, steer ships across the seas of ideas and reality, navigating and bridging the past (and memories which are expressed through behaviour, a style and some routines at the social level), current reality and the future. The path for network behaviour for innovation is found precisely on this journey. It is the

capacity to build non pre-defined futures, thus requiring autonomy, creativity and dynamic setting of constraints to define behaviour. Network behaviour creates structures which shape further network behaviour. This is why complex systems cannot survive without constraints or with too many or too tight ones.

Setting a path requires facing the unknown. Science and rationality are symbols of commitments (or binding rationality). Leaders must know their weaknesses and bind themselves to the mast. Like Ulysses, they are symbols of courage and power through the commitments they establish are symbols to themselves, to their *crews* (asking commitment in return) and to the ship. Leaders see the world through new perspectives, but require Strategic Management to steer ships through the murky waters of time and the changeable weather conditions of organising: what has been, what is, and what may be.

Orchestrating is a managerial function that enables a network capacity, which is not a characteristic of one individual or a single focal firm. It is in this context where Network Leaders steer sets of commitments to build new paths. The critical variables are first: the capacity to build references, second: understanding that circumstances vary along the way, and third: managing dynamic constraints to balance autonomy and multi-connected networks with limits of action to optimize creativity and communication. Network Leaders exert power as a continuous process of dynamically controlling boundaries by reaching a series of optimums, opening new spaces by regulating boundaries, and taking advantage of the potential power of digital platforms. When boundaries expand, it is like opening a window - the air is colder but also fresher. Leaders are in command and give autonomy by putting rules of the game (boundaries) in place, mobilising resources, recognizing waves (unfolding mainsail when needed) and setting direction in co-creative ways, and managing key people in key projects (leading leaders). Leaders steer the ship by showing resilience, that is, by binding themselves to the mast, while at other times being unleashed in order to orchestrate network behaviour towards innovation.

9.- References

Abbott, A. (1995). "Things of Boundaries: Defining the Boundaries of Social Inquiry". *Social Research*, 62(4), 857-882.

Adner, R. (2006). "Match Your Innovation Strategy to Your Innovation Ecosystem". *Harvard Business Review*, *84*(4), 98-107.

Ahuja, G. (2000). "Collaboration Networks, Structural Holes and Innovation: A Longitudinal Study". *Administrative Science Quarterly*, *45*(3): 425-455.

Aldrich, H. (1979). *Organizations and Environments*. Englewood Cliffs, N.J.: Prentice-Hall.

Almeida, P. & Kogut, B. (1999) "Localization of Knowledge and the Mobility of Engineers in Regional Networks". *Management Science*, *45* (7), 905-18.

Alstyne, M. van (1997). "The State of the Network Organization: A Survey of Three Frameworks". *Journal of Organizational Computing & Electronic Commerce*, 7(3), 83-151.

Álvarez, J. L. & Svejenova, S. (2005). *Sharing Executive Power: Roles and Relationships at the Top.* Cambridge: Cambridge University Press

Anderson, J. R. (1983). *The Architecture of Cognition*. Boston: Harvard University Press.

Argyris, C. (1986). "Reinforcing Organizational Defensive Routines: An Unintended Human Resources Activity". *Human Resources Management, 25*(4) (April): 541-555.

Argyris, C. (1990). *Overcoming Organizational Defences: Facilitating Organizational Learning*. Boston: Allyn & Bacon.

Argyris, C. (1994). "Good Communication that Blocks Learning". *Harvard Business Review*, *72*(4), 77-85.

Argyris, C. (2004). *Reasons and Rationalizations*. Oxford; New York: Cambridge: Oxford University Press

Argyris, C. & Schön, D. A. (1974). *Theory in Practice: Increasing Professional Effectiveness*. San Francisco: Jossey-Bass.

Argyris, C. & Schön, D. A. (1996). *Organizational Learning II: Theory, Method and Practice*. Reading, Mass.: Addison-Wesley.

Aron, A. & Westbay, L. (1996). "Dimensions of the prototype of love", *Journal of Personality and Social Psychology*, *70*(3), 535-51.

Augier M. & Teece, D. J. (2008). "Strategy as Evolution with Design: The Foundations of Dynamic Capabilities and the Role of Managers in the Economic System", *Organization Studies*, *29*(8-9), 1187-1208.

Barnard, C. I. (1968). *The Functions of the Executive* (2nd ed.). Cambridge, Mass.: Harvard University Press.

Barney, J. (1994). "Firm Resources and Sustained Competitive Advantage". *Journal of Management, 17*(1), 99-120.

Barney, J. (2007). *Resource-Based Theory: Creating and Sustaining Competitive Advantage*, New York: Oxford University Press.

Bartlett, C. A. & Ghoshal, S. (2002). *Managing Across Borders: The Transnational Solution*. Boston: Harvard Business School Press.

Bartlett, C. A. & Ghoshal, S. (1990). "Managing Innovation in the Transnational Corporation". In C. A. Bartlett, Y. Doz & G. Hedlund (Eds.). *Managing the Global Firm*. London; New York: Routledge.

Beardsley, S. C.; Johnson, B.C. & Manyika, J. M. (2006). "Competitive Advantage from Better Interactions". *McKinsey Quarterly*, *2*, 53-63.

Becker, E. (1977). *The Denial of Death*. New York: The Free Press.

Bleeke, J. & Ernst, D. (1995). "Is Your Strategic Alliance Really a Sale?". *Harvard Business Review*, 73(1), 97-105

Benkler, Y. (2006). *The Wealth of Networks: How Social Production Transforms Markets and Freedom*. New Haven, Conn.: Yale University Press.

Bettelheim, B. (1988). No hay padres perfectos. Barcelona: Crítica.

Bijker, W. E.; Law, J. (1992). *Shaping Technology/Building Society: Studies in Socio-Technical Change*. Cambridge, Mass.: MIT Press.

Bohm, D. (1965). *The Special Theory of Relativity*. New York: W.A. Benjamin.

Bohm, D. & Peat, D. (2002). Science, Order and Creativity. New York: Routledge.

Bowlby, J. (1962). *Attachment*. New York: Tavistock Institute of Human Relations. Basic Books.

Bowlby, J. (1980). *Loss.* New York: Tavistock Institute of Human Relations. Basic Books.

Bowlby, J. (1985) La separación afectiva. Madrid: Paidós.

Breschi, S. & Malerba, F. (2007). *Clusters, Networks and Innovation*. Oxford: Oxford University Press.

Bridges, W. (2003). *Managing Transitions: Making the Most of Change*. Chicago: Da Capo Press.

Brown, J. S.; Durchslang, S. & Hagel III, J. (2002), "Loosening up: how process networks unlock the power of specialization", *Mc Kinsey Quarterly*, 2, 58-69

Brown, S. L. & Eisenhardt, K. M. (1997). "The Art of Continuous Change: Linking Complexity Theory and Time-paced evolution in Relentlessly Shifting Organizations". *Administrative Science Quarterly*, *42*(1): 1-34.

Burns, T. & Stalker, G. M. (1961). *The Management of Innovation*. London: Tavistock.

Burt, R. S. (1992). Structural Holes: The Social Structure of Competition. Cambridge, Mass.: Harvard University Press.

Burt, R.S. (2004). "Structural Holes and Good Ideas". *American Journal of Sociology*, *110*, 349-99.

Chakravarthy, B.; McEvily, S.; Doz, Y. & Rau, D. (2003). "Knowledge Management and Competitive Advantage". In: M. Easterby-Smith, & M. A. Lyles (Eds.) . *Handbook of Organizational Learning and Knowledge Management*. Malden, Mass.; Oxford: Blackwell Publishing

Chandler, A. D. (1962). *Strategy and Structure: Chapters in the History of the Industrial Enterprise*. Cambridge, Mass.: Harvard University Press.

Chandler, A. D. (1977). *The Visible Hand: The Managerial Revolution in American Business*. Cambridge, Mass.: Harvard University Press.

Chandrashekar, A. & Schary, P. B. (1999) "Toward the Virtual Supply Chain: The Convergence of IT and Organization". *International Journal of Logistics Management*, *10*(2), 268-285.

Cheng, Y. & Van de Ven, A. H. (1996) "Learning the Innovation Journey", *Organization Science*, 7(6), 593-614.

Chesbrough, H. & Teece, D. J. (1996). "When Is Virtual, Virtuous? Organizing for Innovation". *Harvard Business Review*, *74*(1), 65-73.

Chesbrough, H. (2003). *Open Innovation: The New Imperative for Creating and Profiling from Technology.* Boston: Harvard Business School Press.

Chesbrough, H. (2006). *Open Business Models.* Boston: Harvard Business School Press.

Chesbrough, H. (2008). "Orchestrating Appropriability: Towards an Endogenous View of Capturing Value from Innovation Investments". In S. Shane (Ed.) *Handbook of Technology and Innovation Management*. West Sussex, UK: Wiley.

Child, J. (1972). "Organizational Structure, Environment and Performance: the role of Strategic Choice". *Sociology*, 6(1), 1-22.

Child, J. (1997). "Strategic Choice in the analysis of Action, Structure, Organization and Environment: Retrospect and Prospect". *Organization Studies* 18(1), 43-76.

Child, J. & Rodrigues, S. (2003). "Social Identity and Organizational Learning". In: M. Easterby-Smith & M. A. Lyles (Eds.). *Handbook of Organizational Learning and Knowledge Management"*. New York: Blackwell Publishing

Child, J.; Faulkner, D. & Tallman, S. (2006). *Cooperative Strategy*. Oxford: Oxford University Press.

Christensen, C. M. (1997). *The Innovator's Dilemma*. Boston: Harper Business Essentials.

Christensen, C. M. & Overdorf, M. (2000). "Meeting the Challenge of Disruptive Change". *Harvard Business Review*, *78*(1), 66-77.

Clegg, S.R. (1989). Frameworks of Power. London: SAGE.

Clemmensen, T.; Khryashcheva, M. & Podshibikhina, O. (2008). "Combining Bases of Trust Development in Virtual Teams". *International Journal of Networking and Virtual Organisations*, *5*,(1), 17-34.

Coase, R. H. (1988). *The Firm, the Market and the Law*. Chicago: The University of Chicago Press.

Cohen, W. M. & Levinthal, D. A. (1990). "Absorptive Capacity: A New Perspective on Learning and Innovation". *Administrative Science Quarterly*, *35*(1), 128-152.

Coleman, J. S. (1990). *Foundations of Social Theory*. Cambridge, Mass.: Harvard University Press.

Contu, A. & Willmott, H. (2003). "Re-Embedding Situatedness: The Importance of Power Relations in Learning Theory". *Organization Science*, *14*(3), 283-296.

Conway, S. & Steward, F (1998). "Mapping Innovation Networks". *International Journal of Innovation Management, 2*(2), 165-196.

Cook, K. S. & Emerson, R. M. (1984). "Exchange Networks and the Analysis of Complex Organizations". *Research in the Sociology of Organizations*, 3, 1-30

Cook, K. & Yamagishi, T. (1992). "Power in Exchange Networks: A power-dependency formulation". *Social Networks*, *14*(3-4), 245-265.

Cowan, R. (2007). "Network Models of Innovation and Knowledge Diffusion". In *Clusters, Networks and Innovation*. Oxford: Oxford University Press.

Crossan, M. M.; Lane, H. W.; White, R. E.; Djurfeldt, L. (1995) "Organizational Learning: Dimensions for a Theory". *The International Journal of Organizational Analysis*, *3*(4), 337-360.

Cyert, R. M. & March, J. G. (1963). *A Behavioural Theory of the Firm*. New Jersey: Prentice Hall.

Damanpour, F. (1991). "Organizational Innovation: A Meta-analysis of Effects of Determinants and Moderators". *Academy of Management Journal*, *34*(3), 555-590.

Damanpour, F. & Gopalakrishnan, S. (1998). "Theories of Organizational structure and innovation adoption: the role of environmental change". *Journal of Engineering and Technology Management*, *15*(1), 1-24.

Day, G. S. & Schoemaker, J.H. (2000). "A different game". In *Wharton on Managing Emerging Technologies*. New Jersey: John Willey & Sons.

DeBresson, C. & Amese, F. (1991). "Networks of Innovators: A Review and Introduction to the Issue". *Research Policy*, *20*(5), 363-379.

Delporte-Verneiren, D. J. E.; Varvest, P. H. M. & Heck, E. van (2004). "In Search of Margin for Business Networks: The European Patent Office". *European Management Journal*, *22*(2): 167-182.

Demers, C. (2007). *Organizational Change Theories: A Synthesis*. San Francisco: Sage Publications.

Dhanaraj, C. & Parkhe, A. (2006). "Orchestrating Innovation Networks". *Academy* of *Management Reviewl*, *31*(3), 659–669.

Di Bella, J. & Nevis, E. (1998). *How Organizations Learn*. San Francisco: Jossey-Bass.

Donaldson, L. (2003). "Organization Theory as a Positive Science". In H. Tsoukas & C. Knudsen (Ed.). *The Oxford Handbook of Organization Theory: Meta-Theoretical Perspectives* (pp. 39-62). Oxford: Oxford University Press.

Dooley, K. & Van de Ven, A.H. (1997). "The Nonlinear Dynamics of Innovation", *Society for Chaos Theory in Psyhology and the Life Sciences, 4*(1), 3-4.

Dosi, G. (1982). "Technological Paradigms and Technological Trajectories". *Research Policy*, *11*(3), 147-162.

Dougherty, D. (2008). "Bridging Social Constraint and Social Action to Design Organization for Innovation". *Organization Studies*, *29*(3), 415-434.

Doz, Y. L. (1996). "The Evolution of Cooperation in Strategic Alliances: Initial Conditions or Learning Processes?". *Strategic Management Journal*, *17*(Special issue), 55-83.

Doz, Y. L.; Olk, P. M. & Ring, P. S. (2000). "Formation Processes of R&D Consortia: Which Path to Take? Where Does it Lead? *Strategic Management Journal*, *21*(3), 239-266.

Doz, Y. L. & Prahalad, C. K. (2005). "Managing DMNCs: A Search for New Paradigm," In S. Ghosha & E. D. Westney (Eds.). *Organization Theory and the Multinational Corporation* (2nd ed.). Houndmills, Basingstoke, Hampshire ; New York : Palgrave Macmillan

Durkheim, E. (1933). *The Division of Labor in Society*. Glencoe, Ill.: The Free Press.

Durkheim, E. (1937/2003). *Las reglas del método sociológico*. Buenos Aires: Gorla.

Dyer, J. & Singh, H. (1998). "The relational view: cooperative strategy and interorganizational sources of competitive advantage". *Academy of Management Review*, *23*(4), 660-79

Dyer, J.; and Kale, P. (2007). "Relational Capabilities: Drivers and implications". In C. E. Helfat, S. Finkelstein, W. Mitchell, M. Peteraf, H. Singh, D. Teece & S. G. Winter (Eds.). *Dynamic Capacities: Understanding Strategic Change in Organizations*. Malden, Mass.: Blackwell Publishing.

Easterby-Smith, M. (1997). "Disciplines of Organizational Learning: Contributions and Critiques". *Human Relations*, *50*(9), 1085-1116.

Easterby-Smith, M.; Araujo, L. (1999). "Organizational Learning: Current Debates and Opportunities". In M. Easterby-Smith, J. Burgoyne & L. Araujo (Eds.). *Organizational Learning and Learning Organization. Developments in Theory and Practice.* London; Thousand Oaks, Calif: Sage Publications.

Easterby-Smith, M.; Crossan, M. & Nicolini, D. (2000). "Organizational Learning: Debates, Past, Present and Future". *Journal of Management Studies*, *37*(6), 783-796.

Easterby-Smith, M. & Lyles M. A. (Eds.) (2003). *Handbook of Organizational Learning and Knowledge Management*. Malden, Mass.; Oxford: Blackwell Publishing.

Eden, C. & Spender, J. C. (1998). *Managerial and Organizational Cognition: Theory, Methods and Research*. London: Sage Publications.

Edquist, C. (2006). "Systems of Innovation: Perspectives and Challenges". In J. Fagerberg, D. C. Mowery & R R. Nelson (Eds.) *The Oxford Handbook of Innovation*. Oxford: Oxford University Press.

Eisenhardt, K. M. (1989). "Building Theories from Case Study Research". *Academy of Management Review*, *14*(4), 532-550.

Eisenhardt, K. M. (1989:2) "Agency Theory: An Assessment and Review". *Academy of Management Review*, *14*(1), 57-74.



Eisenhardt, K. M. & Tabrizi, B. N. (1995). "Accelerating Adaptive Processes: Product Innovation in the Global Computer Industry". *Administrative Science Quarterly*, *40*(1), 84-110.

Eisenhardt, K.M. & Schoonhoven, C.B. (1996). "Resource-based View of Strategic Alliance Formation: Strategic and Social Effects in Entrepreneurial Firms". *Organization Science*, 7(2), 136-150.

Eisenhardt, K. M. & Martin, J. (2000). "Dynamic Capabilities: What are they?". *Strategic Management Journal*, *21*(10-11, special issue), 1105-1121

Elfring, T. & Hulsink, W. (2007). "Networking by Entrepreneurs: Patterns of Tie Formation in Emerging Organizations". *Organization Studies*, 28(12), 1849-1872.

Elkjaer, B. (1999). "In Search of a Social Learning Theory". In: M. Easterby-Smith, J. Burgoyne, & L. Araujo (Eds.). *Organizational Learning and Learning Organization: Developments in Theory and Practice*. London: Sage Publications.

Elkjaer, B. (2003). "Social Learning Theory: Learning as Participation in Social Processes". In M. Easterby-Smith &, M. A. Lyles. *Handbook of Organizational Learning and Knowledge Management*. Malden, Mass.; Oxford: Blackwell Publishing.

Elster, J. (1989). *The Cement of Society*. Cambridge: Cambridge University Press.

Elster, J. (1994) *Lógica y sociedad: contradicciones y mundos posibles*. Madrid: Gedisa.

Elster, J. (2000) Ulises y las sirenas. México: Fondo de Cultura Económica.

Elster, J. (2000) Ulysses Unbound. Cambridge: Cambridge University Press.

Emerson, R. M. (1962). "Power-Dependency Relations". *American Sociological Review*, 27(1), 37-40.

Emerson, R. M. (1987). "Toward a Theory of Value in Social Exchange". In K. S. Cook (Ed.). *Social Exchange Theory*. Newbury Park, Calif.: Sage Publications.

Ferriani, S.; Cattani, G. & Baden-Fuller, C. (2009). "The Relational Antecedents of Project-Entrepreneurship: Network Centrality, Team Composition and Project Performance". *Research Policy*, *38*(10), 1545-1558

Festinger, L. (1957). *A Theory of Cognitive Dissonance*. Stanford, Calif.: Stanford University Press.

Feyerabend, P. (2002). *Against Method*, 3rd ed. Cambridge: Verso.

Fichman, R. (2000). "The Diffusion and Assimilation of Information Technology Innovations". In R. W. Zmud (Ed.). *Framing the Domains of IT Management:* : Projecting the future - through the past . Cincinnati: Pinnaflex. Fiol, C. & Lyles, M. (1985). "Organizational Learning". *Academy of Management Review, 10*(4), 803-813.

Fisher, G. & Giaccardi, E. (2007). "Creativity Support Tools: Accelerating Discovery and Innovation". *Communications of the ACM*, *50*(12): 20-32.

Floricel, S. & Dougherty, D. (2007). "Where Do Games of Innovation Come From? Explaining the Persistence of Dynamic Innovation Patterns". *International Journal of Innovation Management*, 11(1), 65-91.

Freud, S. (1920/2001). "Más allá del principio de placer". In *Obras completas* (18 vol.). Madrid: Amorrortu.

Freud, S. (1924/2001). "Psicología de las masas y análisis del yo". In *Obras completas* (18 vol.). Madrid: Amorrortu.

Freud, S. (1966/2007). *Tótem y tabú*. Madrid: Alianza Editorial.

Freud, S. (1966/2007). *El malestar en la cultura*. Madrid: Alianza Editorial.

Freud, S. (1971). Introducción al Psicoanálisis. Madrid: Alianza Editorial

Fung, V. K.; Fung, W. K. & Wind, Y. (2007). *Competing in a Flat World*. Philadelphia, Pa.: Wharton School Publishing.

Galbraith, J. (1973). *Designing Complex Organizations*. Reading, Mass.: Addison-Wesley.

Galbraith, J. (1977). *Organization Design*. Reading, Mass.: Addison-Wesley.

Galbraith, J. (2005). *Designing the Customer Centric Organization*. San Francisco: Jossey-Bass.

Galbraith, J.; Downey, D. & Kates, A. (2002). *Designing Dynamic Organizations*. New York: Amacom.

Garfinkel, H. (1967). *Studies in Etnomethodology*. Englewood Cliffs, NY: Prentice-Hall.

Garnsey, E. & McGlade, J. (2006). *Complexity and Co-Evolution: Continuity and Change in Socio-Economic Systems*. Northampton, Mass.: Edward Elgar Publishing.

Garvin, D. (2000). *Learning in Action: A Guide to Putting the Learning Organization to Work*. Boston: Harvard Business School Press.

Gawer, A.; Cusumano, M. (2008). "How Companies Become Platform Leaders". *MIT Sloan Management Review*, *49*(2), 27-36

Gergen, K. J. & Gergen, M. M. (1986). *Social Psychology*. New York: Prentice-Hall.

Gersick, C. J. G. (1991). "Revolutionary Change Theories: A Multi-Level Exploration of the Punctuated Equilibrium Paradigm". *Academy of Management Review*, *16*(1), 10-36.

Ghemawat, P. (1991). *Commitment: The Dynamics of Strategy*. New York: Free Press.

Ghemawat, P.(1997). *Games Businesses Play: Cases and Models.* Cambridge, Mass; London: The MIT Press.

Ghoshal, S. & Westney, E. D. (2005). *Organization Theory and the Multinational Corporation*. Basingstoke: Palgrave Macmillan.

Giddens, A. (1987). *Las nuevas reglas del método sociológico*. Madrid: Amorrortu.

Giddens, A. (1991). *Modernidad e identidad del yo*. Barcelona: Península.

Giddens, A. (1995). *La constitución de la sociedad. Bases para la teoría de la estructuración*. Madrid: Amorrortu..

Gigerenzer, G. & Selten, R. (2001). *Bounded Rationality*. Cambridge, Mass.; London: The MIT Press.

Gilsing, V.; Nooteboom, B.; Vanharverbeke, W.; Duysters, G. & Oord, Van Den, A. (2008). "Network Embeddeness and the exploration of novel technologies: technology distance, betweenness, centrality and density". *Research Policy*, . $3\pi(10)$, 1717-1731.

Gnyawali, D. R. & Madhavan, R. (2001). "Cooperative Networks and Competitive Dynamics: a structural embeddedness perspective". *Academy of Management Review*, *26*(3), 431-445.

Granovetter, M. S. (1973). "The Strength of Weak Ties". *American Journal of Sociology*, *78*(6): 1360-1380.

Granovetter, M. S. (1985). "Economic Action and Social Structure: The Problem of Embeddedness". *American Journal of Sociology*, *91*(3), 481-510.

Gresov, C., & Drazin, R. (1997). "Equifinalty, Functional Equivalence in Organizational Design". *Academy of Management Review, 22*(2), 403-428.

Guba, E. G. & Lincon, Y. S. (1994). "Competing Paradigms in Qualitative Research". In N. Denzin & Y. S. Lincon (Eds.). *Handbook of Qualitative Research*. Thousand Oaks, Calif.: Sage Publications.

Gulati R., Nohria N. & Zaheer A. (2000). "Strategic Networks". *Strategic Management Journal*, *21*(3), 203-215.

Gummesson, E. (1999). *Total Relationship Marketing. Rethinking Marketing Management: From 4Ps to 30Rs*. Oxford: Butterworth-Heinemann.

Gurbaxari, V. & Whang, S. (1991). "The Impact of Information Systems on Organizations and Markets". *Communications on ACM*, *34*(1), 567-577.

Habermas, J. (1971/2002). *Knowledge and Human Interests*. Boston: Beacon Press.

Hakansson H. & Snehota I. (1989). "No Business is an Island", Scandinavian Journal of Management Studies. Vol.4, No. 3/89, pp 187-200.

Hakansson H. & Snehota I. (1995). *Developing Relationships in Business Networks*. London, Routledge.

Häcki, R. & Lighton, J. (2001). "The Future of the Networked Company". *McKinsey Quarterly*, *3*, 26-39.

Hagel III, J. & Singer, M. (1999). "Unbundling the Corporation". *Harvard Business Review*, 77(2), 133-141.

Hamel, G. (1998). "Strategy Innovation and the Quest for Value". *Sloan Management Review*, *39*(2), 7-14.

Hamel, G. (2007). *The Future of Management*. Boston: Harvard Business School Press.

Hatch, M. J. (2004). "Dynamics in Organizational Culture". In M. S. Poole & A. H. Van der Ven (Eds.). *Handbook of Organizational Change and Innovation.* Oxford *:*: Oxford University Press.

Hatch, M. J. & Yanow, D. (2003). "Organization Theory as an Interpretive Science". In H. Tsoukas & C. Knudsen (Eds.) *Organization Theory: Meta-Theoretical Perspectives*. Oxford: Oxford University Press.

Hatch, M. J.; Yanow, D. (2008) "Methodology by Metaphor: Ways of Seeing in Painting and Research", *Organization Studies*, *29*(1): 23-44.

Have, P. T. (2004). *Understanding Qualitative Research and Ethnomethodology*. London: Sage Publications.

Hayek, F. A. (1949/2009). *El Individualismo: el verdadero y el falso*. Madrid: Unión Editorial.

Hellgren, B. & Löwstedt, J. (1998). "Agency and Organization: A Social Theory Approach to Cognition". In C. Eden & J. C. Spender (Eds.) *Managerial and Organizational Cognition*. London: Sage Publications.

Hickson, D. J.; Hinings, C. R.; Lee, C. A.; Schneck, R. E. & Pennings, J. M. (1971). "A strategic contingencies' theory of Intra-organizational Power", *Administrative Science Quarterly*, *16*(2), 216-229.



Hinterhuber, A. (2002). "Value Chain Orchestration in Action and the Case of Agrochemical Industry". *Long Range Planning*, *35*(6): 615-635.

Holling, C. S. (1973). "Resilience and Stability in Ecological Systems". Annual Review of Ecology and Systematics, *4*, 2-23.

Huber, G. (1991). "Organizational Learning: The Contributing Process and the Literature". *Organization Science*, 2(1), 88-115.

Huber, G. & Van de Ven, P. (1995). *Longitudinal Field Research Methods*, London: SAGE.

Huisman, K. J. M. & Kort, P. M. (2003). "Strategic Investment in technological innovations". *European Journal of Operational Research*, *144*(1), 209-223.

Huizinga, J. (1927/2007). *Homo Ludens*. Madrid: Alianza Editorial.

Jarrillo, J. C. (1993). *Strategic Networks: Creating the Borderless Organization*. Oxford: Butterworth Heinemann.

Kash, D. E., & Rycoft, R. W. (2000). "Patterns of Innovating Complex Technologies: A Framework for Adaptive Network Strategies". *Research Policy*, *29*(7-8), 819-831.

Kautz, K. & Zumpe, S. (2008). "Just Enough Structure at the Edge of Chaos: Agile Information System Development in Practice Abrahamsson, et al. (Eds.) *Agile Processes in Software Engineering and Extreme Programming: 9th International Conference, XP 2008* (pp. 137-146). Berlin; Heidelberg: Springer-Verlag.

Kelley, T. (2005). *The Ten Faces of Innovation*. San Francisco: Currency Doubleday.

Kets de Vries, M. (2003). *Leaders, Fools and Impostors: Essays on the Psychology of Leadership*. London: iUniverse.

Kets de Vries, M. & Engellau, E. (2004). *Are leaders born or are they made?: The Case of Alexander the Great.* New York: Karnac

Kets de Vries, M. (2006). *The Leader on the Couch: A Clinical Approach to Changing People and Organizations*. San Francisco: Jossey-Bass.

Kets de Vries, M. (2009). *Sex, Money, Happiness and Death: The Quest for Authenticity.* New York: Palgrave

Kets de Vries, M. (1980). Organizational Paradoxes. New York: Routledge.

Kets de Vries, M.; Miller, D. (1984). *The Neurotic Organization*. San Francisco; London: Jossey-Bass.

Kilduff, M. & Tsai, W. (2003). *Social Networks and Organizations*. London: Sage Publications.

Kimberly, J. & Miles, R. (1980). *The Organizational Life-Cycle: Issues in the Creation, Transformation and Decline of Organizations*. San Francisco: Jossey-Bass.

Kogut, B. (2000). "The Network as Knowledge: Generative Rules and the Emergence of Structure". *Strategic Management Journal*, *21*(3), 405-425.

Kogut, B.; Shan, W. & Walker, G. (1992) "The make-or-cooperate Decisions in the context of an Industry Network". In Nohria, N. & Eccles, R. (Eds) *Networks and Organizations: Structure, Forms and Actions*. Boston, MA: Harvard Business School Press: 348-365.

Kogut, B. (1988). "Joint Ventures: Theoretical and Empirical Perspectives" *Strategic Management Journal*, *9*(4), 319-32.

Kotter, J. P. (1978). *Organizational Dynamics: Diagnosis and Intervention*. Reading, Mass.: Addison-Wesley.

Krackhardt, D. (1990). "Assessing the Political Landscape: Structure, Cognition and Power in Organizations". *Administrative Science Quarterly*, *35*(2), 342-369.

Krackhardt, D. (1992). "The Strengh of Strong Ties: The Importance of Philos in Organizations". In N. Nohria & R. G. Eccles (Eds.). *Networks and Organizations: Structures, Forms and Action*. Boston: Harvard School Press, 216-239.

Kuhn, T. (1970). *The Structure of Scientifical Revolutions*. Chicago: University of Chicago Press.

Lam, A. (2007)- "Organizational Innovation" In J. Fagerberg, D. Mowery & R. Nelson (Eds.) *The Oxford Handbook of Innovation.* Oxford: Oxford University Press.

Larsson, R.; Bengtsson, L.; Henriksson, K. & Sparks, J. (1998). "The Interorganizational Learning Dilemma: Collective Knowledge Development in Strategic Alliances". *Organization Science*, *9*(3), 285-305.

Lasserre, P. (1992) "Managing Large Groups in the East and the West". *European Management Journal*, *10*(2) (June): 157-112.

Laukkanen, M. (1998) "Conducting Cause-Mapping Research: Opportunities and Challenges". In C. Eden & J. C. Spender (Eds.). *Managerial and Organizational Cognition* (pp. 168-191). *London:* Sage Publications.

Lawrence, P.R. & Lorsch, J.W. (1967). "Differentiation and Integration in Complex Organizations". *Administrative Science Quarterly*, *12*(1), 1-47.

Lave, C. & March, J. (1975). *An Introduction to Models in the Social Sciences.* New York: Harper & Row.

Lave, J. (1988). *Cognition in Practice: Mind, Mathematics and Culture in Everyday Life*. Cambridge: Cambridge University Press.

Lave, J. & Wenger, E. (1991). *Situated Learning: Legitimate Peripheral Participation*. Cambridge: Cambridge University Press.

Leonard-Barton, D. (1995). *Wellsprings of Knowledge: Building and Sustaining the Sources of innovation*. Boston: Harvard Business School Press.

Levi-Strauss, C. (2002). *El pensamiento salvaje*. Madrid: Fondo de Cultura Económica.

Levitt, T. (2002). "Creativity Is Not Enough". *Harvard Business Review*, *80*(8), 137-145.

Levitt, B. & March, J. (1988). "Organizational Learning". *Annual Review of Sociology*. *14*, 319-338..

Lorenzi, G. & Baden-Fuller, C. (1995). "Creating a Strategic Center to Manage a Web of Partners". *California Management Review*, *37*(3), 146-163.

Lyles, M. & Easterby-Smith, M. (2003). "Organizational Learning and Knowledge Management: Agendas for Future Research". In M. Easterby-Smith & M. Lyles *Handbook of Organizational Learning and Knowledge Management* (pp. 639-652). Oxford: Blackwell Publishing.

Lyytinen, K. & Newman, M. (2008). "Explaining Information Systems Change: a Punctuated Socio-Technical Change Model", *European Journal of Information Systems. 17*, 589-613.

Makhija, M. V. & Ganesh, U. (1997). "The relationship between control and partner learning in learning-related joint ventures". *Organization Science*, *8*(5), 508-527.

Maletz, M.; Nohria, N. (2001) "Managing in the Whitespace". *Harvard Business Review*, *79*(2), 102-111.

Malone, T. W.; Yates, J. & Benjamin, R. I. (1987). "Electronic Markets and Electronics Hierarchies". *Communications of the ACM*, 30(6), 484-497.

March, J. G. (1991). "Exploration and Exploitation in Organizational Learning". *Organization Science*, 2(1), 71-87.

March, J. G.; Olsen, J. P. (1975). "The Uncertainty of the Past: Organizational Learning under Ambiguity". In March, J. G. (1988) *Decisions and Organizations*. Oxford: Blackwell Publishing.

March, J. G. & Simon, H. (1958/1993). *Organizations*. Cambridge, Mass.: Blackwell Publishing.

Markus, M. L. (2004) "Technochange Management: Using IT to Drive Organizational Change". *Journal of Information Technology*, *19*(1), 3-19.

Mason, R. O.; McKenney, J. L. & Copeland, D. G. (1997). "Developing an Historical Tradition in MIS Research", *MIS Quarterly*, *21*(3),, 257-278.

Merton, R. K. (1949). On Theoretical Sociology. Glencoe, Ill. Free Press.

Merton, R. K. (1957). *Social Theory and Social Structure*. Glencoe, Illinois: Free Press.

Milgram, S. (2005). *Obedience to Authority*. London: Pinter and Martin.

Miller, R. & Floricel, S. (2007). "Games of Innovation: A new theoretical perspective", *International Journal of Innovation Management*, *11*(1), 1-35.

Miller, R. & Olleros, X. (2007). "The dynamics of Games of Innovation". *International Journal of Innovation Management*, *11(*1), 37-64

Mintzberg, H. (1973). *The Nature of Managerial Work*. New York; London: Harper and Row.

Mintzberg, H. (1994). *The Rise and Fall of Strategic Planning*. New York; London: The Free Press.

Mintzberg, H. (1998). *The Strategy Safari: A Guided Tour through the Wilds of Strategic Management*. New York; London: The Free Press.

Mintzberg, H. (1999). "Managing Quietly". *Leader to Leader Journal*, Spring: 24-30.

Mintzberg, H. (2002). La estructuración de las organizaciones. Barcelona: Ariel.

Mintzberg, H.; Lampel, J.; Quinn, J. B. & Ghoshal, S. (2003). *The Strategy Process: Concepts, Context and Cases.* Englewood Cliffs, N.J.: Prentice-Hall.

Mintzberg, H. & Van der Heyden, L. (1999). "Organigraphs: Drawing How Companies Really Work". *Harvard Business Review*, 77(5), 87-94.

Mintzberg, H. (1991). *Mintzberg y la dirección*. Madrid: Díaz de Santos.

Molm, L. D., Takahashi, N. & Peterson, G. (2000). "Risk and trust in social exchange: an experimental test of a classic proposition". *American Journal of Sociology*, *105*(5): 1396-1427

Molm, L.D. (2003). "Theoretical Comparisons of forms of exchange". *Sociological Theory*, *21*(1). 1-17

Monge, P. R. & Contractor, N. S. (2003). *Theories of Communication Networks*. Oxford: Oxford University Press.

Montgomery, K. & Oliver, A. (2007). "A Fresh Look at How Professions Take Shape: Dual-Directed Networking Dynamics and Social Boundaries". *Organization Studies*, *28*(5), 661-687.

Morgan, G. (1997). *Images of Organization*. Thousand Oaks, Calif.: Sage Publications.

Moscovici, S. & Mugny, G. (1983). "Minority Influence". In P. B. Paulhus (Ed.). *Basic Group Processes* (pp 41-64). New York: Springer-Velag.

Murray, F. (2002). "Innovation as co-evolution of scientific and technological networks: exploring tissue engineering", *Research Policy*, *31*(8-9), 1389-1403.

Nadler, D. A.; Shaw, R. B. & Walton, A. E. (1995). *Discontinuous Change*. San Francisco: Jossey-Bass.

Nahapiet, J. & Ghoshal, S. (1998). "Social Capital, Intellectual Capital and the Organizational Advantage". *Academy of Management Review*, *23*(2): 242-266.

Nardi, B. (2001). *Context and Consciousness: Activity Theory and Human Computer Interaction*. Cambridge, Mass.: The MIT Press.

Nemeth, C. (1986). "Differential contributions on Minority influence" *Psychological Review*, *93*(1), 23-32.

Niall, H. & Geoff Walsham, G. (2003). "Knowledge and ICTs: A Relational Perspective". In M. Easterby-Smith & M. Lyles (Eds.). *Handbook of Organizational Learning and Knowledge Management*. Malden, Mass.; Oxford: Blackwell Publishing.

Nohria, N. & Ghoshal, S. (1997). *The Differentiated Network: Organizing Multinational Corporations for Value Creation*. San Francisco: Jossey-Bass.

Nonaka, I. & Takeuchi, H. (1995). *The Knowledge-Creating Company*. Oxford: Oxford University Press.

Norton, D. (2009). "Case Study: IBM Uses Agile Methods to Develop WebSphere Service Registry and Repository", *Gartner Research* ID G00163548, 15-January-2009.

Orlikowski, W. J. (1991). "Information Strategy and the Structuring of Organizations". *Information Systems Research*, 2(2), 143-169.

Orlikowski, W. J. (1993). "CASE Tools as Organizational Change: Investigating Incremental and Radical Changes in System Development". *MIS Quarterly*, 17(3), 309-340.

Orlikowski, W. J. (1996). "Improvising Organizational Transformation Over Time: A Situated Change Perspective". *Information Systems Research*, 7(1), 63-92.

Orlikowski, W. J.; Lacono, C. S. (2001). "Research Commentary: Desperately Seeking the "IT" in IT Research-A Call to Theorizing the IT Artifact". *Information Systems Research*, *12*(2), 121-134.

Orlikowski, W. J. & Yates, J. (2002). "It's About Time: Temporal Structuring in Organizations". *Organization Science*, *13*(6), 684-700.

Orsenigo, L.; Pammolli, F. & Riccaboni, M. (2001), "Technological change and network dynamics: Lessons from the pharmaceutical industry", *Research Policy*, 30(3), 485-508.

Orr, Julian E. (1990). "Sharing Knowledge, Celebrating Identity: Community Memory in a Service Culture". In D. S. Middleton, D. Edwards. (Eds.) *Collective Remembering* (pp. 169-189). London: Sage Publications.

Orton, J. D. & Weick, K. E. (1990). "Loosely Coupled Systems: A Reconceptualization". *Academy of Management Review*, *15* (2), 203-223.

Parsons, T. (1937). *The Structure of Social Action.* New York; London: McGraw Hill.

Pavitt, K. (1984). "Sectorial Patterns of Technological Change: Towards a Taxonomy and a Theory". *Research Policy*, *13*(6), 343-373.

Pavitt, K.; and Steinmueller, E. (2002). "Technology in Corporate Strategy: Change Continuity and the Information Revolution", In A. Pettigrew, H. Thomas & R. Whittington (Eds.). *Handbook of Strategy and Management* (pp. 344-372)., London: Sage Publications.

Pedersen, M. K. (1996). *A Theory of Information. The Business Cycle Model. Economics, Organization, Information Systems and Management*. Copenhagen: Samfundslitteratur & Universitetsforlag.

Pedersen, M. K. & Larsen, M. H. (2006). "Innocuous Knowledge: Models of Distributed Knowledge Networks". In K. C. Desouza (Ed.). *New Frontiers of Knowledge Management* (pp. 243-266). Basingstoke: Palgrave Macmillan.

Peirce, C. S. (1992). *The Essential Peirce, Selected Philosophical Writings (1867-1893)* (Volume 1). N. Houser & C. Kloesel (Eds). Bloomington: Indiana University Press.

Penrose, E. H. (1959). *The Theory of the Growth of the Firm*. Oxford: Oxford University Press.

Penrose, E. H. (2008). "The Metamorphosis of the Large Firm". *Organization Studies*, *29*(8-9), 1117-1124. [This article was originally written by E. H. Penrose in 1994.]

Peteraf, M. & Barney, J. (2003). "Unraveling the resource based tangle", *Managerial and Decision Economics, 24(4)*, 309-324.

Peteraf, M.; Pitelis, C. & Zollo, M. (2008). "On The Metamorphosis of (the Theory of) the Firm: An Introduction". *Organization Studies*, *29*(8-9), 1109-1115.

Pettigrew, A. M. (1990). "Longitudinal Field Research on Change: Theory and Practice". *Organizational Science*, 1(3), 267-292.

207

Pfeffer, J. & Salancik, G. R. (1974). "Organizational Decision Making as a Political Process: The Case of a University Budget". *Administrative Science Quarterly*, *19*(2), 135-151.

Pfeffer, J. & Nowak, P. (1976). "Joint Ventures and Interorganizational interdependency", *Administrative Science Quarterly*, *21*(3), 398-418

Pfeffer, J. & Salancik, G. R. (1978). *The External Control of Organizations: A Resource Dependence Perspective.* New York: Harper & Row.

Pffefer, J. (1992). *Managing with Power: Politics and Influence in Organization.* Boston: Harvard Business School Press.

Pisano, G. P. (1990) "The R&D Boundaries of the Firm: An Empirical Analysis". *Administrative Science Quarterly, 35*(1), 153-176.

Pisano, G.P. (1991). "The Governance of Innovation: Vertical Integration and Collaborative Arrangements in the Biotechnology Industry". *Research Policy*, *20*(3), 237-249.

Poole, M. S. & Van de Ven, A. H. (2004). *Handbook of Organizational Change and Innovation*. New York: Oxford University Press.

Poole, M. S.; Van de Ven, A. H.; Dooley, K. & Holmes, M. E. (2000). *Organizational Change and Innovation Processes.* New York: Oxford University Press.

Popper, K. (1957). *The Open Society and its Enemies*. Princeton, N.J.: Princeton University Press.

Popper, K. (1969). *The Logic of Scientific Discovery*. London: Hutchinson.

Porter, M. E. (1980). *Competitive Strategy: Techniques for Analyzing Industries and Competitors*. New York: Free Press.

Porter, M. E. (1983). "The Technological Dimension of Competitive Strategy". *Research on Technology Innovation, Management and Policy*, 1 1-33.

Porter, M. E. (1996). "What is Strategy". *Harvard Business Review, 74*(6), 61-78.

Porter, M. E. (1998) "Clusters and the New Economics of Competition". *Harvard Business Review*, *76*(6,77-90.

Powell, W. (1990). "Neither Markets nor Hierarchy: Network Forms of Organization". *Research in Organizational Behaviour*, *12*, 295-336.

Powell, W.; Doput, K.W. & Smith-Doerr, L. (1996). "Inter-Organizational Collaboration and the Locus of Innovation Networks of Learning in Bio-technology", *Administrative Science Quarterly*, *41*(1), 116-145.

Powell, W.; Koput, K.; Smith-Doerr, L. & Owen-Smith, J. (1999). "Network position and Firm performance: Organizational Returns on Collaboration in the

Biotechnology Industry", *Research on the Sociology of Organizations*, *16*, 129-59.

Powell, W. & Grodal, S. (2006). "Networks of Innovation". In In J. Fagerberg, D. C. Mowery & R R. Nelson (Eds.). *The Oxford Handbook of Innovation*. Oxford: Oxford University Press.

Powell, W.; Koput, K. & Owen-Smith, J. (2005). "Network Dynamics and Field Evolution: The Growth of Interorganizational Collaboration in Life Science". *American Journal of Sociology*, *110*(4): 1132-1205.

Prahalad, C. K. & Hamel, G. (1990) "The Core Competence of the Corporation". *Harvard Business Review*, *68*(3): 79-91.

Prahalad, C. K. (1998). "Managing discontinuities: the emerging challenge". *Research-Technology Management.* . *41* (3), 14-22,

Quinn, J. B. (1980). "Managing Strategic Change". *Sloan Management Review*, 21(4), 3-20.

Quinn, R.E. (1988). *Beyond Rational Management: Mastering the Paradoxes of Competing Demands of High Performance*. San Francisco: Jossey-Bass.

Quinn, J. B.; Baruch, J. J. & Zien, K. A. (1996). "Software-Based Innovation". *Sloan Management Review*, *37*(4), 11-24.

Rand, A. (1967). Introduction to Objectivist Epistemology. New York: Objectivist.

Raymond, A. (2001). *The Cathedral and the Bazaar: Musing on Linux and Open Source by an Accidental Revolutionary.* Cambridge, Mass.: O'Reilly & Associates.

Ring, P. S. & Van de Ven, A. H. (1994). "Development Processes of Cooperative Inter-Organizational Relationships", *Academy of Management Review*, *19*(1), 90-118

Ritala, P.; Armila, L. & Blomqvist, K. (2009). "Innovation Orchestration Capability: Defining the Organizational and Individual Level Determinants". *International Journal of Innovation Management*, *13*(4), 569-591.

Rogers, E. M. (1995). *Diffusion of Innovations* (4th ed.). New York: Free Press.

Ryle, G. (1949). *The Concept of Mind*. London: Hatchinson University Library.

Salk, J. E. & Simonin, B. L. (2003). "Beyond Alliances: Towards a Metatheory of Collaborative Learning". In M. Easterby-Smith & M. Lyles (Eds.). *Handbook of Organizational Learning and Knowledge Management*. Malden, Mass.; Oxford: Blackwell Publishing.

Sambamurthy, V. (2000). "Business Strategy in Hypercompetitive Environments: Rethinking the Logic of IT Differentiation". In Zmud, R. (Ed.). *Framing the Domains of IT Management*. Cincinnati: Pinnaflex.

Scarbrough, H.; Swan, J. (2003). "Discourses in Knowledge Management and the Learning Organization: Their Production and Consumption". In M. Easterby-Smith & M. Lyles (Eds.). *Handbook of Organizational Learning and Knowledge Management*. Malden, Mass.; Oxford: Blackwell Publishing.

Schein, E. (1980). *Organizational Psychology*. Englewood Cliffs, N.J.: Prentice Hall.

Schein, E. (1996). "Three Cultures of Management: The Key to Organizational Learning". *Sloan Management Review*, 38(1), 9-20.

Schein, E. (2004). *Organizational Culture and Leadership*. San Francisco: Jossey-Bass.

Schön, D. (1973). *Beyond the Stable State*. New York: The Norton Library.

Schön, D. (1983). *The Reflective Practitioner: How Professionals Think in Action*. New York: Basic Books.

Schumpeter, J. (1943). *Capitalism, Socialism and Democracy*. London: George Allen & Unwin.

Scott, J. (1991). Social Network Analysis: A Handbook. London: Sage.

Scott, J. (2001). *Power*. Cambridge: Polity Press.

Scott, R. W. (2003). *Organizations: Rational, Natural and Open Systems* (5th ed.) Prentice-Hall.

Seely, J. & Duguid, P. (2001). "Knowledge and Organization: a Social-Practice Perspective". *Organization Science*, *12*(2), 198-213.

Shapiro, S.P. (1987) "The Social Control of Inter-Personal Trust" *American Journal of Sociology*, *93*(3), 623-658.

Shapiro, C. & Varian, H. (1999). *Information Rules. A Strategic Guide to the Network Economy*. Boston: Harvard Business School Press.

Shenkar, O. & Zeira, Y. (1992). "Role Conflict and Role Ambiguity of Chief Executive Officers in International Joint Ventures". *Journal of International Business Studies*, *23*(1), 55-75.

Shneiderman, B. (2007). "Creativity Support Tools: Accelerating Discovery and Innovation". *Communications of the ACM*, *50*(12). 20-32.

Söderlung, J. & and Bredin, K. (2006). "HRM Project Intensive Firms: Changes and Challenges". Human Resource Management, *45*(2), 249-265.

Soh, P.H. & Roberts, E.B. (2003). "Networks of Innovators: A Longitudinal Perspective". *Research Policy*, *32*(9). 1569-1588.

210

Stacey, R. D. (2003) *Strategic Management and Organisational Dynamics: The Challenge of Complexity* (4th ed.). New York: Prentice-Hall.

Straub, D.; Rai, A. & Klein, R. (2004). "Measuring Firm Performance at the Network Level: A Nomology of the Business Impact of Digital Supply Networks". *Journal of Management Information Systems*, *21*(1), 83-114.

Suarez, F. (2004). "Battles for Technological Dominance: An Integrative Framework". *Research Policy*, No. *33*(2): 271-286.

Subramani, M. (2004) "How Do Suppliers Benefit from Information Technology Use in Supply Chain Relationships?". *MIS Quarterly*, *28*(1), 45-73.

Sutton, (1987). "The Process of Organizational Death: Disbanding and reconnecting". *Administrative Science Quarterly*, *32*(4), 542-569.

Teece, D. J. (1987). "Profiting from Technological Innovation: Implications for Integration, Collaboration, Licensing, and Public Policy". *Research Policy*, *15*(6), 285-305.

Teece, D. J. (1998). "Dessign Issues for Innovative firms: Bureaucracy, Incentives and Industry Structure". In A. D. Chandler Jr., P. Hagström, & O. Solvell (Eds.). *The Dynamic Firm: The Role of Technology, Strategy, Organization and Regions* (pp. 134-165). Oxford: Oxford University Press.

Teece, D. J. (1992). "Competition, Cooperation and Innovation: Organizational Arrangements for Regimes of Rapid Technological Progress". *Journal of Economic Behavior and Organization*, *18*(1), 1-25.

Teece, D. J.; Pisano, G. & Shuen, A. (1997). "Dynamic Capabilities and Strategic Management". *Strategic Management Journal*, *18*(7), 509-533.

Teece, D. J. (2007). "Explicating Dynamic Capabilities: The Nature of Microfoundations of (Sustainable) Enterprise Performance". *Strategic Management Journal, 28*(13), 1319-1350.

Teece, D. J. (2009). *Dynamic Capabilities and Strategic Management: Organizing for Innovation and Growth*. New York: Oxford University Press.

Thomas, K. W. (1976). "Conflict and Conflict Management". In M. D. Dunnette (Ed.). *Handbook of Industrial and Organizational Psychology*. Chicago: Rand McNally.

Thompke, S. (2001). "Enlightened Experimentation: The New Imperative for Innovation". *Harvard Business Review.* 79(2), 67-75.

Thompke, S. & Von Hippel, E. (2002). "Customers as Innovators. A New Way to Create Value". *Harvard Business Review*, *80*(4). 74-81.

Thompson, V. A. (1965). "Bureaucracy and Innovation". *Administrative Science Quarterly*, *10*(1), 1-20.

Thompson, J. D. (1967/2003). *Organizations in Action: Social Sciences Bases of Administrative Theory* (2nd Ed.). New York: Transaction Publishers.

Thorelli, H. B. (1986). "Networks: Between Markets and Hierarchies". In *Strategic Management Journal*, χ (1), 37-51.

Tsai, K. H. (2009). "Collaborative Networks and Product Innovation Performance: Towards a Contingency Perspective". *Research Policy*, *38*(5), 765-778.

Tsai, W. (2001). "Knowledge Transfer in Inter-Organizational Networks: Effects of Network Position and Absorptive Capacity on Business Unit Innovation and Performance". *Academy of Management Journal*, *44*(5), 996-1004.

Tsoukas, H. (1996). "The Firm as a Distributed Knowledge System: A Constructionist Approach". *Strategic Management Journal*, *17*(Winter special issue), 11-25.

Tsoukas, H.; Knudsen, C. (2003). *The Oxford Handbook of Organization Theory: Meta-Theoretical Perspectives*. Oxford: Oxford University Press.

Tuomi, I. (2002). *Networks of Innovation.* Oxford: Oxford University Press.

Utterback, J. M. & Suarez, F. F. (1993). "Innovation, competition and industry structure". *Research Policy*, *22*(1), 1-21

Uzzi, B. & Spiro, J. (2005). "Collaboration and Creativity: The Small World Problem". *American Journal of Sociology*, *111*(2), 447-504.

Van Heck, E. & Vervest, P. (2007). "Smart Business Networks: How the Network Wins". *Communications of the ACM*, *50*(6), 28-37.

Van der Heijden, K. & Eden, C. (1998). "The Theory and Praxis of Reflective Learning in Strategy Making". In C. Eden, & J. C. Spender. (Eds.). *Managerial and Organizational Cognition*. San Francisco: Sage Publications.

Van de Ven, A.H. (1986). "Central Problems in the management of innovation". *Management Sciences*, *32* (5), 590-607.

Van de Ven, A. H. & Rogers, E. M. (1988). "Innovations and Organizations: Critical perspectives". *Communications Research*, *15*(5), 632-651.

Van de Ven, A. H. (1992). "Suggestion for Studying Strategy Process Research", *Strategic Management Journal*, *13*(Summer special issue), 169-191.

Van de Ven, A. H. & Garud, R. (1993) "Innovation and Industry Development: The Case of Cochlear Plants". In R. Burgelman & R. Rosenbloom (Eds.). *Research on technological innovation and management policy* (Vol. 5, pp. 1-46). Greenwich, CT: JAI Press..

Van de Ven, A. H. & Poole, M. S. (1995). "Explaining development and Change in Organizations". *Academy of Management Review*, *20*(3), 510-540.
Van de Ven, A. H. & Hargrave, T. (2004). "Social, Technical and Institutional Change: A Literature Review and Synthesis". In M. S. Poole, M. S. & A. H. Van der Ven *Handbook of Organizational Change and Innovation*. Oxford: Oxford University Press.

Van de Ven, A. H.; Ring, P.; Zaheer, A. & Bachman, R. (2006). Relying on Trust in Cooperative Inter-Organizational Relationships". In R. Bachmann & A. Zaheer (Eds.). *Handbook of Trust Research.* London: Edward Elgar.

Van de Ven, A. H.; Polley, D.; Garud, R. & Venkatraman, S. (2008). *The Innovation Journey*. Oxford: Oxford University Press.

Vervest, P.H.M.; Van Heck.; Preiss, K. & Pau, L. (2008) Smart Business Networks. ERIM: The Netherlands

Van Wijk, R.; Van den Bosch, A. J. & Volberda, H. W. (2003). "Knowledge and Networks". In M. Easterby-Smith & M. Lyles (Eds.). *Handbook of Organizational Learning and Knowledge Management*. Malden, Mass.; Oxford: Blackwell Publishing.

Vidich, A. & Lyman, S. M. (1994). "Qualitative Methods". In N. Denzin & Y. S. Lincon. *Handbook of Qualitative Research*. Thousand Oaks: Sage Publications.

Von Hippel, E. (2005). *Democratizing Innovation*. Cambridge, Mass.: The MIT Press.

Von Hippel, E.; Thompke, S. & Sonnack, M. (1999). "Creating Breakthroughs at 3M". *Harvard Business Review*, 77(5), 47-56.

Vries, de, E.J. (2006). "Innovation in Services in Networks of Organizations and in the distribution of services". *Research Policy*, *35*(7), 1037-1051.

Vygotsky, L. S. (1978). *Mind in Society: The Development of Higher Psychological Processes.* Cambridge, Mass.: Harvard University Press.

Vygotsky, L.S. (1987). *The Collected Works of L. S. Vygotsky: Problems of General Psychology*. R. Rieber & A. S. Carton (Eds.). New York; London: Plenum Press.

Wasserman, S. & Galaskiewicz, J. (Eds.) (1994) *Advances in Social Network Analysis: Research in the Social and Behavioral Sciences*. London: SAGE.

Weik, K. E. (1979). *The Social Psychology of Organizing*. Reading Mass.: Addison-Wesleyl.

Weik, K. E. (1989). "Theory Construction as Disciplined Imagination". *Academy* of *Management Review*, *14*(4), 516-531.

Weik, K. E.; Quinn, R. E. (1999). "Organizational Change and Development". *Annual Reviews on Psychology*, *50*(1), 361-86.

Weik, K. E.; Roberts, K. H. (1993). "Collective Mind in Organizations: Heedful Interrelating on Flight Decks". *Administrative Science Quarterly, 38*(3), 357-381.

Wenger, E. (1998). *Communities of Practice: Learning, Meaning and Identity*. New York: Cambridge University Press.

Wenger, E.; McDermott, R. & Snyder, W. M. (2002). *Cultivating Communities of Practice*. Boston: Harvard Business School Press.

Wernerfelt, B. (1984). "A Resource Based view of the Firm". *Strategic Management Journ*al, *5*(2), 171-180.

Wernerfelt, B. (1995). "The Resource Based View of the Firm: Ten Years Later, *Strategic Management Journal*, *16*(3), 171-175.

Wertsch, J. V. (1985). *Vygotsky and the Social Formation of Mind*. Cambridge, Mass.: Harvard University Press.

White, H. C. (2008). *Identity and Control: How Social Formations Emerge* (2nd ed.). Princeton: Princeton University Press.

Williamson, O. E. (1975). *Markets and Hierarchies: Analysis and Antitrust Implications*. New York: Free Press.

Williamson, O. E. (1991). "Comparative Economic Organization: The Analysis of Discrete Structural Alternatives". *Administrative Science Quarterly*, *36*(2), 269-296.

Williamson, O. E. (1994). "Transaction Cost Economics and Organization Theory". In N. J. Smelser & R. Swedberg (Eds.). *The Handbook of Economic Sociology*, (pp. 77-107). Princeton: Princeton University Press.

Wilson, J. M.; O'Leary, M. B.; Metiu, A. & Jett, Q. R. (2008). "Perceived Proximity in Virtual Work: Explaining the Paradox of Far but Close". *Organization Studies*, *29*(7), 979-1001.

Winch, P. (1958). *The Idea of a Social Science and its Relation to Philosophy*. London: Routledge & Kegan Paul.

Winnicott, D. W. (1964). *The Child, the Family and Outside World*. London: Perseus Publishing.

Winnicott, D. W. (1965). *The Maturational Process and the Facilitating Environment*. London: International University Press.

Winnicott, D. W. (1971). *Realidad y juego.* Madrid: Gedisa.

Winnicott, D. W. (1986). *El hogar, nuestro punto de partida*. Madrid: Paidós.

Winnicott, D. W. (1991) Human Nature. New York: Brunner-Mazel.

Wolfe, R.A. (1994). "Organizational Innovation: Review, Critique and Suggested Research directions", *Journal of Management Studies*, 31(3), 405-431.

Yin, R. K. (2003). *Case Study Research: Design and Methods* (ed.). Thousand Oaks, Calif.: Sage Publications.

Zollo, M. & Winter, S. G. (2002). "Deliberate Learning and the Evolution of Dynamic Capabilities". *Organizational Science*, *13*(3), 339-351.

Zuboff, S. (1985). "Automate/Informate: The Two Faces of Intelligent Technology". *Organizational Dynamics*, *14*(2), 5-18.

TITLER I PH.D.SERIEN:

2004

- 1. Martin Grieger Internet-based Electronic Marketplaces and Supply Chain Management
- 2. Thomas Basbøll LIKENESS A Philosophical Investigation
- 3. Morten Knudsen Beslutningens vaklen En systemteoretisk analyse of moderniseringen af et amtskommunalt sundhedsvæsen 1980-2000
- 4. Lars Bo Jeppesen Organizing Consumer Innovation A product development strategy that is based on online communities and allows some firms to benefit from a distributed process of innovation by consumers
- 5. Barbara Dragsted SEGMENTATION IN TRANSLATION AND TRANSLATION MEMORY SYSTEMS An empirical investigation of cognitive segmentation and effects of integrating a TM system into the translation process
- 6. Jeanet Hardis Sociale partnerskaber Et socialkonstruktivistisk casestudie af partnerskabsaktørers virkelighedsopfattelse mellem identitet og legitimitet
- 7. Henriette Hallberg Thygesen System Dynamics in Action
- 8. Carsten Mejer Plath Strategisk Økonomistyring
- 9. Annemette Kjærgaard Knowledge Management as Internal Corporate Venturing

– a Field Study of the Rise and Fall of a Bottom-Up Process

- 10. Knut Arne Hovdal De profesjonelle i endring Norsk ph.d., ej til salg gennem Samfundslitteratur
- 11. Søren Jeppesen Environmental Practices and Greening Strategies in Small Manufacturing Enterprises in South Africa – A Critical Realist Approach
- 12. Lars Frode Frederiksen Industriel forskningsledelse – på sporet af mønstre og samarbejde i danske forskningsintensive virksomheder
- 13. Martin Jes Iversen The Governance of GN Great Nordic – in an age of strategic and structural transitions 1939-1988
- 14. Lars Pynt Andersen The Rhetorical Strategies of Danish TV Advertising A study of the first fifteen years with special emphasis on genre and irony
- 15. Jakob Rasmussen Business Perspectives on E-learning
- 16. Sof Thrane The Social and Economic Dynamics of Networks

 – a Weberian Analysis of Three Formalised Horizontal Networks
- 17. Lene Nielsen Engaging Personas and Narrative Scenarios – a study on how a usercentered approach influenced the perception of the design process in the e-business group at AstraZeneca
- S.J Valstad
 Organisationsidentitet
 Norsk ph.d., ej til salg gennem
 Samfundslitteratur

- 19. Thomas Lyse Hansen Six Essays on Pricing and Weather risk in Energy Markets
- 20. Sabine Madsen Emerging Methods – An Interpretive Study of ISD Methods in Practice
- 21. Evis Sinani The Impact of Foreign Direct Investment on Efficiency, Productivity Growth and Trade: An Empirical Investigation
- 22. Bent Meier Sørensen Making Events Work Or, How to Multiply Your Crisis
- 23. Pernille Schnoor Brand Ethos Om troværdige brand- og virksomhedsidentiteter i et retorisk og diskursteoretisk perspektiv
- 24. Sidsel Fabech Von welchem Österreich ist hier die Rede? Diskursive forhandlinger og magtkampe mellem rivaliserende nationale identitetskonstruktioner i østrigske pressediskurser
- 25. Klavs Odgaard Christensen Sprogpolitik og identitetsdannelse i flersprogede forbundsstater Et komparativt studie af Schweiz og Canada
- 26. Dana B. Minbaeva Human Resource Practices and Knowledge Transfer in Multinational Corporations
- 27. Holger Højlund Markedets politiske fornuft Et studie af velfærdens organisering i perioden 1990-2003
- 28. Christine Mølgaard Frandsen A.s erfaring Om mellemværendets praktik i en

transformation af mennesket og subjektiviteten

29. Sine Nørholm Just
The Constitution of Meaning
– A Meaningful Constitution?
Legitimacy, identity, and public opinion in the debate on the future of Europe

- 1. Claus J. Varnes Managing product innovation through rules – The role of formal and structured methods in product development
- 2. Helle Hedegaard Hein Mellem konflikt og konsensus – Dialogudvikling på hospitalsklinikker
- 3. Axel Rosenø Customer Value Driven Product Innovation – A Study of Market Learning in New Product Development
- 4. Søren Buhl Pedersen Making space An outline of place branding
- 5. Camilla Funck Ellehave Differences that Matter An analysis of practices of gender and organizing in contemporary workplaces
- 6. Rigmor Madeleine Lond *Styring af kommunale forvaltninger*
- 7. Mette Aagaard Andreassen Supply Chain versus Supply Chain Benchmarking as a Means to Managing Supply Chains
- 8. Caroline Aggestam-Pontoppidan From an idea to a standard The UN and the global governance of accountants' competence
- 9. Norsk ph.d.
- 10. Vivienne Heng Ker-ni An Experimental Field Study on the

Effectiveness of Grocer Media Advertising Measuring Ad Recall and Recognition, Purchase Intentions and Short-Term Sales

- 11. Allan Mortensen Essays on the Pricing of Corporate Bonds and Credit Derivatives
- 12. Remo Stefano Chiari Figure che fanno conoscere Itinerario sull'idea del valore cognitivo e espressivo della metafora e di altri tropi da Aristotele e da Vico fino al cognitivismo contemporaneo
- 13. Anders Mcllquham-Schmidt Strategic Planning and Corporate Performance An integrative research review and a meta-analysis of the strategic planning and corporate performance literature from 1956 to 2003
- 14. Jens Geersbro The TDF – PMI Case Making Sense of the Dynamics of Business Relationships and Networks
- 15 Mette Andersen Corporate Social Responsibility in Global Supply Chains Understanding the uniqueness of firm behaviour
- 16. Eva Boxenbaum Institutional Genesis: Micro – Dynamic Foundations of Institutional Change
- 17. Peter Lund-Thomsen Capacity Development, Environmental Justice NGOs, and Governance: The Case of South Africa
- 18. Signe Jarlov Konstruktioner af offentlig ledelse
- 19. Lars Stæhr Jensen Vocabulary Knowledge and Listening Comprehension in English as a Foreign Language

An empirical study employing data elicited from Danish EFL learners

- 20. Christian Nielsen Essays on Business Reporting Production and consumption of strategic information in the market for information
- 21. Marianne Thejls Fischer Egos and Ethics of Management Consultants
- 22. Annie Bekke Kjær Performance management i Procesinnovation – belyst i et social-konstruktivistisk perspektiv
- 23. Suzanne Dee Pedersen GENTAGELSENS METAMORFOSE Om organisering af den kreative gøren i den kunstneriske arbejdspraksis
- 24. Benedikte Dorte Rosenbrink Revenue Management Økonomiske, konkurrencemæssige & organisatoriske konsekvenser
- 25. Thomas Riise Johansen Written Accounts and Verbal Accounts The Danish Case of Accounting and Accountability to Employees
- 26. Ann Fogelgren-Pedersen The Mobile Internet: Pioneering Users' Adoption Decisions
- 27. Birgitte Rasmussen Ledelse i fællesskab – de tillidsvalgtes fornyende rolle
- 28. Gitte Thit Nielsen
 Remerger skabende ledelseskræfter i fusion og opkøb
- 29. Carmine Gioia A MICROECONOMETRIC ANALYSIS OF MERGERS AND ACQUISITIONS

- 30. Ole Hinz Den effektive forandringsleder: pilot, pædagog eller politiker? Et studie i arbejdslederes meningstilskrivninger i forbindelse med vellykket gennemførelse af ledelsesinitierede forandringsprojekter
- Kjell-Åge Gotvassli
 Et praksisbasert perspektiv på dynamiske
 læringsnettverk i toppidretten
 Norsk ph.d., ej til salg gennem
 Samfundslitteratur
- 32. Henriette Langstrup Nielsen Linking Healthcare An inquiry into the changing performances of web-based technology for asthma monitoring
- 33. Karin Tweddell Levinsen Virtuel Uddannelsespraksis Master i IKT og Læring – et casestudie i hvordan proaktiv proceshåndtering kan forbedre praksis i virtuelle læringsmiljøer
- 34. Anika Liversage Finding a Path Labour Market Life Stories of Immigrant Professionals
- 35. Kasper Elmquist Jørgensen Studier i samspillet mellem stat og erhvervsliv i Danmark under 1. verdenskrig
- 36. Finn Janning A DIFFERENT STORY Seduction, Conquest and Discovery
- 37. Patricia Ann Plackett Strategic Management of the Radical Innovation Process Leveraging Social Capital for Market Uncertainty Management

1. Christian Vintergaard Early Phases of Corporate Venturing

- 2. Niels Rom-Poulsen Essays in Computational Finance
- 3. Tina Brandt Husman Organisational Capabilities, Competitive Advantage & Project-Based Organisations The Case of Advertising and Creative Good Production
- Mette Rosenkrands Johansen
 Practice at the top
 how top managers mobilise and use
 non-financial performance measures
- 5. Eva Parum Corporate governance som strategisk kommunikations- og ledelsesværktøj
- 6. Susan Aagaard Petersen Culture's Influence on Performance Management: The Case of a Danish Company in China
- 7. Thomas Nicolai Pedersen The Discursive Constitution of Organizational Governance – Between unity and differentiation The Case of the governance of environmental risks by World Bank environmental staff
- 8. Cynthia Selin Volatile Visions: Transactons in Anticipatory Knowledge
- 9. Jesper Banghøj Financial Accounting Information and Compensation in Danish Companies
- 10. Mikkel Lucas Overby Strategic Alliances in Emerging High-Tech Markets: What's the Difference and does it Matter?
- 11. Tine Aage External Information Acquisition of Industrial Districts and the Impact of Different Knowledge Creation Dimensions

A case study of the Fashion and Design Branch of the Industrial District of Montebelluna, NE Italy

- 12. Mikkel Flyverbom Making the Global Information Society Governable On the Governmentality of Multi-Stakeholder Networks
- 13. Anette Grønning Personen bag Tilstedevær i e-mail som interaktionsform mellem kunde og medarbejder i dansk forsikringskontekst
- 14. Jørn Helder One Company – One Language? The NN-case
- 15. Lars Bjerregaard Mikkelsen Differing perceptions of customer value Development and application of a tool for mapping perceptions of customer value at both ends of customer-supplier dyads in industrial markets
- 16. Lise Granerud Exploring Learning Technological learning within small manufacturers in South Africa
- 17. Esben Rahbek Pedersen Between Hopes and Realities: Reflections on the Promises and Practices of Corporate Social Responsibility (CSR)
- 18. Ramona Samson The Cultural Integration Model and European Transformation. The Case of Romania

2007

1. Jakob Vestergaard Discipline in The Global Economy Panopticism and the Post-Washington Consensus

- 2. Heidi Lund Hansen Spaces for learning and working A qualitative study of change of work, management, vehicles of power and social practices in open offices
- 3. Sudhanshu Rai Exploring the internal dynamics of software development teams during user analysis A tension enabled Institutionalization Model; "Where process becomes the objective"
- 4. Norsk ph.d. Ej til salg gennem Samfundslitteratur
- 5. Serden Ozcan *EXPLORING HETEROGENEITY IN ORGANIZATIONAL ACTIONS AND OUTCOMES A Behavioural Perspective*
- 6. Kim Sundtoft Hald Inter-organizational Performance Measurement and Management in Action

 An Ethnography on the Construction of Management, Identity and Relationships
- 7. Tobias Lindeberg Evaluative Technologies Quality and the Multiplicity of Performance
- 8. Merete Wedell-Wedellsborg Den globale soldat Identitetsdannelse og identitetsledelse i multinationale militære organisationer
- Lars Frederiksen Open Innovation Business Models Innovation in firm-hosted online user communities and inter-firm project ventures in the music industry – A collection of essays
- 10. Jonas Gabrielsen Retorisk toposlære – fra statisk 'sted' til persuasiv aktivitet

- Christian Moldt-Jørgensen Fra meningsløs til meningsfuld evaluering.
 Anvendelsen af studentertilfredshedsmålinger på de korte og mellemlange videregående uddannelser set fra et psykodynamisk systemperspektiv
- 12. Ping Gao Extending the application of actor-network theory Cases of innovation in the telecommunications industry
- Peter Mejlby Frihed og fængsel, en del af den samme drøm? Et phronetisk baseret casestudie af frigørelsens og kontrollens sameksistens i værdibaseret ledelse!
- 14. Kristina Birch Statistical Modelling in Marketing
- 15. Signe Poulsen Sense and sensibility: The language of emotional appeals in insurance marketing
- 16. Anders Bjerre Trolle Essays on derivatives pricing and dynamic asset allocation
- 17. Peter Feldhütter Empirical Studies of Bond and Credit Markets
- 18. Jens Henrik Eggert Christensen Default and Recovery Risk Modeling and Estimation
- Maria Theresa Larsen Academic Enterprise: A New Mission for Universities or a Contradiction in Terms? Four papers on the long-term implications of increasing industry involvement and commercialization in academia

- 20. Morten Wellendorf Postimplementering af teknologi i den offentlige forvaltning Analyser af en organisations kontinuerlige arbejde med informationsteknologi
- 21. Ekaterina Mhaanna Concept Relations for Terminological Process Analysis
- 22. Stefan Ring Thorbjørnsen Forsvaret i forandring Et studie i officerers kapabiliteter under påvirkning af omverdenens forandringspres mod øget styring og læring
- 23. Christa Breum Amhøj Det selvskabte medlemskab om managementstaten, dens styringsteknologier og indbyggere
- 24. Karoline Bromose Between Technological Turbulence and Operational Stability – An empirical case study of corporate venturing in TDC
- 25. Susanne Justesen Navigating the Paradoxes of Diversity in Innovation Practice

 A Longitudinal study of six very different innovation processes – in practice
- 26. Luise Noring Henler Conceptualising successful supply chain partnerships

 Viewing supply chain partnerships from an organisational culture perspective
- 27. Mark Mau Kampen om telefonen Det danske telefonvæsen under den tyske besættelse 1940-45
- 28. Jakob Halskov The semiautomatic expansion of existing terminological ontologies using knowledge patterns discovered

on the WWW – an implementation and evaluation

- 29. Gergana Koleva European Policy Instruments Beyond Networks and Structure: The Innovative Medicines Initiative
- 30. Christian Geisler Asmussen Global Strategy and International Diversity: A Double-Edged Sword?
- 31. Christina Holm-Petersen Stolthed og fordom Kultur- og identitetsarbejde ved skabelsen af en ny sengeafdeling gennem fusion
- 32. Hans Peter Olsen Hybrid Governance of Standardized States Causes and Contours of the Global Regulation of Government Auditing
- 33. Lars Bøge Sørensen Risk Management in the Supply Chain
- 34. Peter Aagaard Det unikkes dynamikker De institutionelle mulighedsbetingelser bag den individuelle udforskning i professionelt og frivilligt arbejde
- 35. Yun Mi Antorini Brand Community Innovation An Intrinsic Case Study of the Adult Fans of LEGO Community
- 36. Joachim Lynggaard Boll Labor Related Corporate Social Performance in Denmark Organizational and Institutional Perspectives

- 1. Frederik Christian Vinten Essays on Private Equity
- 2. Jesper Clement Visual Influence of Packaging Design on In-Store Buying Decisions

- Marius Brostrøm Kousgaard Tid til kvalitetsmåling?

 Studier af indrulleringsprocesser i forbindelse med introduktionen af kliniske kvalitetsdatabaser i speciallægepraksissektoren
- 4. Irene Skovgaard Smith Management Consulting in Action Value creation and ambiguity in client-consultant relations
- 5. Anders Rom Management accounting and integrated information systems How to exploit the potential for management accounting of information technology
- 6. Marina Candi Aesthetic Design as an Element of Service Innovation in New Technologybased Firms
- 7. Morten Schnack Teknologi og tværfaglighed – en analyse af diskussionen omkring indførelse af EPJ på en hospitalsafdeling
- 8. Helene Balslev Clausen Juntos pero no revueltos – un estudio sobre emigrantes norteamericanos en un pueblo mexicano
- 9. Lise Justesen Kunsten at skrive revisionsrapporter. En beretning om forvaltningsrevisionens beretninger
- 10. Michael E. Hansen The politics of corporate responsibility: CSR and the governance of child labor and core labor rights in the 1990s
- 11. Anne Roepstorff Holdning for handling – en etnologisk undersøgelse af Virksomheders Sociale Ansvar/CSR

- 12. Claus Bajlum Essays on Credit Risk and Credit Derivatives
- Anders Bojesen The Performative Power of Competence – an Inquiry into Subjectivity and Social Technologies at Work
- 14. Satu Reijonen Green and Fragile A Study on Markets and the Natural Environment
- 15. Ilduara Busta Corporate Governance in Banking A European Study
- 16. Kristian Anders Hvass A Boolean Analysis Predicting Industry Change: Innovation, Imitation & Business Models The Winning Hybrid: A case study of isomorphism in the airline industry
- 17. Trine Paludan De uvidende og de udviklingsparate Identitet som mulighed og restriktion blandt fabriksarbejdere på det aftayloriserede fabriksgulv
- 18. Kristian Jakobsen Foreign market entry in transition economies: Entry timing and mode choice
- 19. Jakob Elming Syntactic reordering in statistical machine translation
- 20. Lars Brømsøe Termansen Regional Computable General Equilibrium Models for Denmark Three papers laying the foundation for regional CGE models with agglomeration characteristics
- 21. Mia Reinholt The Motivational Foundations of Knowledge Sharing

- 22. Frederikke Krogh-Meibom The Co-Evolution of Institutions and Technology

 A Neo-Institutional Understanding of Change Processes within the Business Press – the Case Study of Financial Times
- 23. Peter D. Ørberg Jensen OFFSHORING OF ADVANCED AND HIGH-VALUE TECHNICAL SERVICES: ANTECEDENTS, PROCESS DYNAMICS AND FIRMLEVEL IMPACTS
- 24. Pham Thi Song Hanh Functional Upgrading, Relational Capability and Export Performance of Vietnamese Wood Furniture Producers
- 25. Mads Vangkilde Why wait? An Exploration of first-mover advantages among Danish e-grocers through a resource perspective
- 26. Hubert Buch-Hansen Rethinking the History of European Level Merger Control A Critical Political Economy Perspective

- 1. Vivian Lindhardsen From Independent Ratings to Communal Ratings: A Study of CWA Raters' Decision-Making Behaviours
- 2. Guðrið Weihe Public-Private Partnerships: Meaning and Practice
- 3. Chris Nøkkentved Enabling Supply Networks with Collaborative Information Infrastructures An Empirical Investigation of Business Model Innovation in Supplier Relationship Management
- 4. Sara Louise Muhr Wound, Interrupted – On the Vulnerability of Diversity Management

- 5. Christine Sestoft Forbrugeradfærd i et Stats- og Livsformsteoretisk perspektiv
- 6. Michael Pedersen *Tune in, Breakdown, and Reboot: On the production of the stress-fit selfmanaging employee*
- 7. Salla Lutz Position and Reposition in Networks – Exemplified by the Transformation of the Danish Pine Furniture Manufacturers
- 8. Jens Forssbæck Essays on market discipline in commercial and central banking
- 9. Tine Murphy Sense from Silence – A Basis for Organised Action How do Sensemaking Processes with Minimal Sharing Relate to the Reproduction of Organised Action?
- 10. Sara Malou Strandvad Inspirations for a new sociology of art: A sociomaterial study of development processes in the Danish film industry
- Nicolaas Mouton
 On the evolution of social scientific metaphors:
 A cognitive-historical enquiry into the divergent trajectories of the idea that collective entities – states and societies, cities and corporations – are biological organisms.
- 12. Lars Andreas Knutsen Mobile Data Services: Shaping of user engagements
- 13. Nikolaos Theodoros Korfiatis Information Exchange and Behavior A Multi-method Inquiry on Online Communities

- 14. Jens Albæk Forestillinger om kvalitet og tværfaglighed på sygehuse

 skabelse af forestillinger i læge- og plejegrupperne angående relevans af nye idéer om kvalitetsudvikling gennem tolkningsprocesser
- 15. Maja Lotz The Business of Co-Creation – and the Co-Creation of Business
- 16. Gitte P. Jakobsen Narrative Construction of Leader Identity in a Leader Development Program Context
- Dorte Hermansen "Living the brand" som en brandorienteret dialogisk praxis: Om udvikling af medarbejdernes brandorienterede dømmekraft
- 18. Aseem Kinra Supply Chain (logistics) Environmental Complexity
- 19. Michael Nørager How to manage SMEs through the transformation from non innovative to innovative?
- 20. Kristin Wallevik Corporate Governance in Family Firms The Norwegian Maritime Sector
- 21. Bo Hansen Hansen Beyond the Process Enriching Software Process Improvement with Knowledge Management
- 22. Annemette Skot-Hansen Franske adjektivisk afledte adverbier, der tager præpositionssyntagmer indledt med præpositionen à som argumenter En valensgrammatisk undersøgelse
- 23. Line Gry Knudsen Collaborative R&D Capabilities In Search of Micro-Foundations

- 24. Christian Scheuer Employers meet employees Essays on sorting and globalization
- 25. Rasmus Johnsen The Great Health of Melancholy A Study of the Pathologies of Performativity
- 26. Ha Thi Van Pham Internationalization, Competitiveness Enhancement and Export Performance of Emerging Market Firms: Evidence from Vietnam
- 27. Henriette Balieu Kontrolbegrebets betydning for kausativalternationen i spansk En kognitiv-typologisk analyse

- 1. Yen Tran Organizing Innovationin Turbulent Fashion Market Four papers on how fashion firms create and appropriate innovation value
- 2. Anders Raastrup Kristensen Metaphysical Labour Flexibility, Performance and Commitment in Work-Life Management
- 3. Margrét Sigrún Sigurdardottir Dependently independent Co-existence of institutional logics in the recorded music industry
- Ásta Dis Óladóttir Internationalization from a small domestic base: An empirical analysis of Economics and Management
- 5. Christine Secher E-deltagelse i praksis – politikernes og forvaltningens medkonstruktion og konsekvenserne heraf
- Marianne Stang Våland What we talk about when we talk about space: End User Participation between Proces-

ses of Organizational and Architectural Design

- 7. Rex Degnegaard Strategic Change Management Change Management Challenges in the Danish Police Reform
- 8. Ulrik Schultz Brix Værdi i rekruttering – den sikre beslutning En pragmatisk analyse af perception og synliggørelse af værdi i rekrutterings- og udvælgelsesarbejdet
- 9. Jan Ole Similä Kontraktsledelse Relasjonen mellom virksomhetsledelse og kontraktshåndtering, belyst via fire norske virksomheter
- 10. Susanne Boch Waldorff Emerging Organizations: In between local translation, institutional logics and discourse
- 11. Brian Kane Performance Talk Next Generation Management of Organizational Performance
- 12. Lars Ohnemus Brand Thrust: Strategic Branding and Shareholder Value An Empirical Reconciliation of two Critical Concepts
- 13. Jesper Schlamovitz Håndtering af usikkerhed i film- og byggeprojekter
- Tommy Moesby-Jensen Det faktiske livs forbindtlighed Førsokratisk informeret, ny-aristotelisk ήθος-tænkning hos Martin Heidegger
- 15. Christian Fich Two Nations Divided by Common Values French National Habitus and the Rejection of American Power

- 16. Peter Beyer Processer, sammenhængskraft og fleksibilitet Et empirisk casestudie af omstillingsforløb i fire virksomheder
- 17. Adam Buchhorn Markets of Good Intentions Constructing and Organizing Biogas Markets Amid Fragility and Controversy
- 18. Cecilie K. Moesby-Jensen Social læring og fælles praksis Et mixed method studie, der belyser læringskonsekvenser af et lederkursus for et praksisfællesskab af offentlige mellemledere
- 19. Heidi Boye Fødevarer og sundhed i senmodernismen

 En indsigt i hyggefænomenet og de relaterede fødevarepraksisser
- 20. Kristine Munkgård Pedersen Flygtige forbindelser og midlertidige mobiliseringer Om kulturel produktion på Roskilde Festival
- 21. Oliver Jacob Weber Causes of Intercompany Harmony in Business Markets – An Empirical Investigation from a Dyad Perspective
- 22. Susanne Ekman Authority and Autonomy Paradoxes of Modern Knowledge Work
- 23. Anette Frey Larsen Kvalitetsledelse på danske hospitaler – Ledelsernes indflydelse på introduktion og vedligeholdelse af kvalitetsstrategier i det danske sundhedsvæsen
- 24. Toyoko Sato Performativity and Discourse: Japanese Advertisements on the Aesthetic Education of Desire

- 25. Kenneth Brinch Jensen Identifying the Last Planner System Lean management in the construction industry
- 26. Javier Busquets Orchestrating Network Behavior for Innovation

TITLER I ATV PH.D.-SERIEN

1992

1. Niels Kornum Servicesamkørsel – organisation, økonomi og planlægningsmetoder

1995

2. Verner Worm Nordiske virksomheder i Kina Kulturspecifikke interaktionsrelationer ved nordiske virksomhedsetableringer i Kina

1999

3. Mogens Bjerre Key Account Management of Complex Strategic Relationships An Empirical Study of the Fast Moving Consumer Goods Industry

2000

4. Lotte Darsø Innovation in the Making Interaction Research with heterogeneous Groups of Knowledge Workers creating new Knowledge and new Leads

2001

5. Peter Hobolt Jensen Managing Strategic Design Identities The case of the Lego Developer Network

- 6. Peter Lohmann The Deleuzian Other of Organizational Change – Moving Perspectives of the Human
- 7. Anne Marie Jess Hansen To lead from a distance: The dynamic interplay between strategy and strategizing – A case study of the strategic management process

- 8. Lotte Henriksen Videndeling

 om organisatoriske og ledelsesmæssige udfordringer ved videndeling i praksis
- 9. Niels Christian Nickelsen Arrangements of Knowing: Coordinating Procedures Tools and Bodies in Industrial Production – a case study of the collective making of new products

2005

10. Carsten Ørts Hansen Konstruktion af ledelsesteknologier og effektivitet

TITLER I DBA PH.D.-SERIEN

2007

1. Peter Kastrup-Misir Endeavoring to Understand Market Orientation – and the concomitant co-mutation of the researched, the researcher, the research itself and the truth

2009

1. Torkild Leo Thellefsen

Fundamental Signs and Significanceeffects

A Semeiotic outline of Fundamental Signs, Significance-effects, Knowledge Profiling and their use in Knowledge Organization and Branding

2. Daniel Ronzani

When Bits Learn to Walk Don't Make Them Trip

Technological Innovation and the Role of Regulation by Law in Information Systems Research: the Case of Radio Frequency Identification (RFID)

2010

1. Alexander Carnera Magten over livet og livet som magt Studier i den biopolitiske ambivalens